

South Sefton Clinical Commissioning Group Southport and Formby Clinical Commissioning Group

# South Sefton Southport & Formby Primary Care Commissioning Committees in Common– Part 1 Agenda

Date: Thursday 18th April 2019 10:00-11:00am

Venue: Family Life Centre, Almond Room, Ash Street Southport PR8 6JH

Members		
Gill Brown	S&F CCG Lay Member (Chair)	GB
Graham Bayliss	SS CCG Lay Member (Vice Chair)	GB
Graham Morris	SS CCG Lay Member	GM
Helen Nichols	S&F CCG Lay Member	HN
Fiona Taylor	S&F SS CCG Chief Officer	FT
Martin McDowell	S&F SS CCG Chief Finance Officer	MMc
Jan Leonard	S&F CCG Director of Place (North)	JL
Debbie Fagan	S&F CCG Chief Nurse and Quality Lead	DF
Angela Price	S&F SS CCG Programme Lead Primary Care	AC

NHSE Senior Commissioning Manager

#### **Non Voting Attendees:**

LMC Representative

Alan Cummings

Healthwatch Representative

Health & Well Being Representative

#### **Minutes**

Debbie Fairclough Interim Programme Lead – Corporate Services DFa

No	Item	Lead	Report	Receive/ Approve	Time
PCCiC 19/01	Apologies for absence	Chair	V		
PCCiC 19/02	Declarations of interest regarding agenda items	All	V		
PCCiC19/03	Minutes of the previous meeting : SS JCC 21.3.19 SF JCC 21.3.19	Chair	R	Α	
PCCiC 19/04	Action points from the previous meeting	Chair	R	R	
PCCiC 19/05	Terms of Reference: South Sefton Primary Care Commissioning Committee Southport and Formby Primary Care Commissioning Committee	JL	R	R	
PCCiC19/06	Delegation Agreement	JL	R	R	
PCCiC19/07	Transition Plan to Delegation	AP	R	R	
PCCiC19/08	Report from Operational Group and Decisions made	JL	V	R	



# South Sefton Clinical Commissioning Group Southport and Formby Clinical Commissioning Group

No	Item	Lead	Report	Receive/ Approve	Time
PCCiC 19/09	Update on GPFV Operational Plan/ Primary Care Programme Report	AP	R	R	
PCCiC 19/10	PCN update	CG/KS	V		
PCCiC 19/11	Overview of GP Contract Changes	AP	Р	R	_
PCCiC 19/12	Health Watch Issues	MK	V		
PCCiC 19/13	Key Issues log	Chair	R	R	
PCCiC 19/14	Any Other Business  Matters previously notified to the Chair no less than 48 hours prior to the meeting.	Chair			
PCCiC 19/15	Date of Next Meeting: 16 <sup>th</sup> May 2019 Room 3A, Merton House, Stanley Road, Bootle L20 3DL	-	-	-	



# **S&F NHSE Joint Commissioning Committee Draft Minutes – Part I**

Date: Thursday 21st March 2019. 10.00am – 11.00am Venue: 3A Merton House, Stanley Road, Bootle, L20 3DL

Members		
Gill Brown	S&F CCG Lay Member (Chair)	GB
Jan Leonard	S&F CCG Chief Redesign and Commissioning Officer (Vice Chair)	JL
Dr Kati Scholtz	S&F CCG Clinical Vice Chair	KS
Alan Cummings	NHSE Senior Commissioning Manager	AC
Susanne Lynch	Head of Medicines Management, S&F CCG	SL
Jan Hughes	NHSE Assistant Contracts Manager	JH
J	J	
Attendees:		SH
Sharon Howard	Programme Manager GPFV	AP
Angela price	Primary Care Programme Lead	MK
Maureen Kelly	Health watch Sefton	CP
Colette Page	SS and S&F CCG Practice Nurse Lead	DJ
Dwayne Johnson	Sefton MBC Director of Social Services and Health	JC
Joe Chattin	Sefton LMC	
Minutes		
Jane Elliott	S&F CCG Senior Administrator	JE

**Attendance Tracker**  $\checkmark$  = Present A = Apologies N = Non-attendance

Name	Membership	Feb 19	Mar 19	Apr 19	May 19	June 19	July 19
Members:							
Gill Brown	S&F CCG Lay Member	✓	✓				
Helen Nichols	S&F CCG Lay Member		Α				
Jan Leonard	S&F CCG Chief Redesign and Commissioning	✓	✓				
Dr Rob Caudwell	S&F CCG Clinical Chair	N	Ν				
Dr Kati Scholtz	S&F CCG Clinical Vice Chair	✓	✓				
Susanne Lynch	S&F CCG Head of Medicines Management	✓	✓				
Brendan Prescott	Deputy Chief Nurse and Quality Officer	N	Ν				
Alan Cummings	NHSE Senior Commissioning Manager	Α	Α				
Attendees:							
Jan Hughes	NHSE Assistant Contract Manager	N	✓				
Sharon Howard	Programme Manager General Practice Forward View	✓	✓				
Angela Price	Primary Care Programme Lead	✓	✓				
Maureen Kelly	Healthwatch Sefton	✓	Α				
Diane Blair	Health watch Sefton		✓				
Dwayne Johnson	Sefton MBC Director of Social Services and Health	N	N				
Joe Chattin	Sefton LMC	N	N				
Anne Downey	NHSE Finance	N	N				
Colette Page	SS and S&F CCG Practice Nurse Lead	Α	Α				

No	Item	Action
JCCiC 19/21.	Introductions and apologies	
	Apologies were received as noted above.	
	It was noted that the meeting was not Quorate. No decisions were made.	
JCCiC 19/22.	Declarations of interest	
10/22	Committee members are reminded of their obligation to declare any interest they may have on any issues arising at committee meetings which might conflict with the business of Southport and Formby Clinical Commissioning Group.	
	Declarations declared by members of the Committee are listed in the CCG's Register of Interests. The Register is available either via the secretary to the governing body or the CCG website.	
	KS declared an interest as a local GP.	
JCCiC 19/23.	Minutes of the previous meeting	
	The minutes of the previous meeting held on 7 <sup>th</sup> February 2019 were agreed.	
JCCiC 19/24.	Action points from the previous meeting	
	The action tracker was discussed and updated.	
JCCiC 19/25.	Report from Operational Group and Decisions made	
	AP reported that the group had discussed:	
	Churchtown Medical Centre list is now open	
	Discussions took place around plans and procedures now in	
	<ul> <li>place take on delegated commissioning</li> <li>LQC phase 5 is progressing and on track for publication in April 2019</li> </ul>	
	<ul> <li>North Locality has concerns around patient movement between practices and is considering a managed list process.</li> <li>The CCG and NHSE will be kept informed of further decisions.</li> </ul>	

JCCiC 19/26.	<ul> <li>GPFV Operational Plan/ Primary Care Programme Report</li> <li>Resilience funding has been received to address the issues relating to Docman documents when patients transfer to another practice</li> <li>Training will be planned over the next 12 months with regards to signposting and document management. First Practice Management and Talking Life will provide training</li> <li>International recruitment team have recruited further staff. Numbers of potential GP recruits are low.</li> <li>ETTF - David Scannell (NHSE) is scheduled to attend the April JOG meeting</li> <li>e-consultations is being promoted and rolled out to all practices. There are some discussions taking place regarding 7 day access services utilising this facility</li> <li>7 day access services. Further funding has been secured to advertise this service further. Digital champion posts will support practice utilisation of the service. There is still some underutilisation of HCA and PN but this is being addressed. NHSE praised the CCG for their promotion of the service so far</li> <li>Clinical pharmacist - the HUB has exceeded all expectations and there have been some good outcomes in patient safety. Practices are keen for the HUB to be rolled out to all practices. This can only be done when all the posts are recruited to</li> <li>PCN - there are 4 Networks in Southport and Formby. Networks are awaiting publication of further guidance around the new DES which requires practice sign up by May 2019. Sefton on track to meet this deadline. NHSE noted how well Sefton had done to have so many bids approved as not all bids were successful as they didn't meet the criteria</li> <li>LQC is business as usual LQC 5 on track</li> </ul>	
	<ul> <li>NHSE praised the CCG for their promotion of the service so far</li> <li>Clinical pharmacist – the HUB has exceeded all expectations and there have been some good outcomes in patient safety. Practices are keen for the HUB to be rolled out to all practices. This can only be done when all the posts are recruited to</li> <li>PCN – there are 4 Networks in Southport and Formby. Networks are awaiting publication of further guidance around the new DES which requires practice sign up by May 2019. Sefton on track to meet this deadline. NHSE noted how well Sefton had done to have so many bids approved as not all bids were successful as they didn't meet the criteria</li> </ul>	
	<ul> <li>delivering the DES themselves or handing this over to the Federation</li> <li>SMI health checks are mandated for practices. This has been included in the LQC phase 5. Practices will offer patients a health check</li> <li>Delegation goes live on 1<sup>st</sup> April 2019. NHSE have been working with the CCG to map across policies and procedures to ensure a smooth transition. A guide has been put together for practices to ensure they know who to contact and for what. Diane Lewis from NHSE has been instrumental in this process and training staff. CCG wish to extend their thanks to Diane for all her hard work</li> </ul>	
JCCiC 19/27.	Delegation As discussed in item JCCiC 19.26	

JCCiC 19/28.	Healthwatch carried out an assessment at Trinity surgery. Only a small number of patients were canvased. Practice did not feel this was a true reflection of the overall picture.  There had been some feedback from a patient who had reported high telephone charges because it took so long to answer the call. It was agreed to ask the practice for information regarding their call waiting time between 8am and 11am. JL will monitor this situation	JL
JCCiC 19/29.	<ul> <li>Key Issues Log</li> <li>Charges for patients who are in telephone queuing systems when contacting GP practices.</li> </ul>	
JCCiC 19/30.	Any Other Business None	
JCCiC 19/31.	Date of Next Meeting:  Thursday 18th April 2019. 10.00am – 11.00am  Family Life Centre, Almond Room, Ash Street Southport PR8 6JH	

#### **Meeting Concluded.**

#### **Motion to Exclude the Public:**

Representatives of the Press and other members of the Public to be excluded from the remainder of this meeting, having regard to the confidential nature of the business to be transacted, publicity on which would be prejudicial to the public interest, (Section 1{2} Public Bodies (Admissions to Meetings), Act 1960)



# **SS NHSE Joint Commissioning Committee DRAFT Minutes – Part I**

Date: Thursday 21st March 2019. 10.00am – 11.00am Venue: 3A Merton House, Stanley Road, Bootle, L20 3DL

Members		
Gill Brown	SF CCG Lay Member (Chair)	GB
Graham Bayliss	SS CCG Lay Member (Chair)	GM
Graham Morris	SS CCG Lay Member	JL
Jan Leonard	SS CCG Chief Redesign and Commissioning Officer	CG
Dr Craig Gillespie	SS CCG Clinical Vice Chair	AC
Alan Cummings	NHSE Senior Commissioning Manager	SL
Suzanne Lynch	SF&SF CCG Head of Medicines Management	BP
Brendan Prescott	Deputy Chief Nurse and Quality Officer	
Attendees:		CP
Colette Page	SSCCG Quality	AP
Angela Price	Primary Care Programme Lead	SH
Sharon Howard	Programme Manager General Practice Forward View	DB
Diane Blair	Healthwatch Sefton	
Minutes		ır
Minutes	22000	JE
Jane Elliott	SSCCG	

Attendance Tracker 
√ = Present 
A = Apologies 
N = Non-attendance 
C= Cancelled

Name	Membe	ership	Feb 19	Mar 19	Apr 19	May 19	June 19
Members:							
Graham Bayliss	SS CCG Lay Member (Chai	r)	N	✓			
Graham Morris	SS CCG Lay Member (Vice	Chair)	Α	✓			
Jan Leonard	SS CCG Chief Redesign an	d Commissioning Officer	✓	✓			
Dr Craig Gillespie	SS CCG Clinical Vice Chair		✓	Α			
Susanne Lynch	S&F CCG Head of Medicine	s Management	✓	✓			
Brendan Prescott	Deputy Chief Nurse and Qua	Deputy Chief Nurse and Quality Officer		N			
Alan Cummings	NHSE Senior Commissionin	g Manager	Α	Α			
Attendees:							
Sharon Howard	Programme Manager Gener	al Practice Forward View	✓	✓			
Angela Price	Primary Care Programme Le	ead	✓	✓			
Diane Blair	Healthwatch Sefton		Α	✓			
Dwayne Johnson	Sefton MBC Director of Soci	ial Services and Health	N	N			
Joe Chattin	Sefton LMC		N	N			
Rebecca McCullough	SSCCG Head of Strategic Financial Planning		N	N			
Jan Hughes	NHSE Assistant Contract Manager		N	✓			
Colette Page	SS&SF CCG Practice Nurse	e Lead	Α	Α			

No	Item	Action
JCCiC 19/21	Apologies for absence Apologies were received as noted above.	
	It was noted that the meeting was not Quorate. No decisions were made.	
JCCiC 19/22	Declarations of interest regarding agenda items	
	GB reminded committee members of their obligation to declare any interest they may have on any issues arising at committee meetings which might conflict with the business of South Sefton Clinical Commissioning Group.	
	Declarations declared by members of the Committee are listed in the CCG's Register of Interests. The Register is available either via the secretary to the governing body or the CCG website.	
JCCiC 19/23	Minutes of the previous meeting	
	The minutes of the previous meeting held on 7 <sup>th</sup> February 2019 were agreed.	
	SS – item JCCiC 19/05 should have read JL reported to the group.	
JCCiC 19/24	Action points from the previous meeting	
	The action tracker was discussed and updated.	
JCCiC 19/25	Report from Operational Group and Decisions made	
	AP reported that the group had discussed:	
	<ul> <li>Bridge Road Medical Centre has submitted an application to close their list. Further information is required from the practice therefore the list remains opens</li> </ul>	
	<ul> <li>Discussions took place around plans and procedures now in place take on delegated commissioning</li> <li>LQC phase 5 is progressing and on track for publication in April 2019</li> </ul>	

JCCiC 19/26	GPFV Operational Plan/ Primary Care Programme Report	
JUGIC 19/26	<ul> <li>Resilience funding has been received to translate common letters in general practice into the top 5 languages spoken locally</li> <li>Training will be planned over the next 12 months with regards to signposting and document management. First Practice Management and Talking Life will provide training</li> <li>International recruitment team have recruited further staff. Numbers of potential GP recruits are low</li> <li>ETTF – David Scannell (NHSE) is scheduled to attend the April JOG meeting</li> <li>e-consultations is being promoted and rolled out to all practices. There are some discussions taking place regarding 7 day access services utilising this facility</li> <li>7 day access services. Further funding has been secured to advertise this service further. Digital champion posts will support practice utilisation of the service. There is still some underutilisation of HCA and PN but this is being addressed. NHSE praised the CCG for their promotion of the service so far.</li> <li>Clinical pharmacist – the HUB has exceeded all expectations and there have been some good outcomes in patient safety. Practices are keen for the HUB to be rolled out to all practices. This can only be done when all the posts are recruited to</li> <li>PCN – there are 4 Networks in South Sefton. Only one practice remains unsubscribed to a network. The CCG are continuing to support this practice. Networks are awaiting publication of further guidance around the new DES which requires practice sign up by May 2019. Sefton had done to have so many bids approved as not all bids were successful as they didn't meet the criteria.</li> <li>LQC is business as usual LQC 5 on track</li> <li>Apex Insight is a tool that enables practices to monitor workload and staffing levels. Software was presented to practices at the Wider Group Meeting. Expressions of interest are now being taken</li> <li>Learning Disability health checks have increased this year with the introduction of the new procedure. 2019/20</li></ul>	
JCCiC 19/27	Delegation As discussed in item JCCiC 19.26	
JCCiC 19/28	Healthwatch Feedback	
- 13. <b>23</b>	Healthwatch had completed and enter and view assessment at the Strand Medical Centre. The practice is aware of access issues and is addressing them. Patients felt that it was hard to gain access to pre-bookable appointments for long term conditions.	
JCCiC 19/29	Key Issues Log None identified	

JCCiC 19/30	Any Other Business	
	None	
JCCiC 19/31	Date of Next Meeting:	
	Thursday 18th April 2019. 10.00am – 11.00am  Family Life Centre, Almond Room, Ash Street Southport PR8 6JH	

#### Meeting Concluded.

#### Motion to Exclude the Public:

Representatives of the Press and other members of the Public to be excluded from the remainder of this meeting, having regard to the confidential nature of the business to be transacted, publicity on which would be prejudicial to the public interest, (Section 1{2} Public Bodies (Admissions to Meetings), Act 1960)





South Sefton Clinical Commissioning Group Southport and Formby Clinical Commissioning Group

# SS SF NHSE Joint Commmissioning Committee in Common – Action Tracker March 2019

Item		CCG	Lead	Time
SSNHSE18/42	ETTF- JL to obtain update from MMcD as to the estates bids for Crosby and Maghull Update 19 <sup>th</sup> April 18 – Bids still live Update 21 <sup>st</sup> June 2018 – Bids still live Update 30 <sup>th</sup> August 2018 – Bids still live Update 18 <sup>th</sup> October 2018 – Bids still live. There has been an expression of interest from Maghull locality. Update 20 <sup>th</sup> December 2018 – David Scannell to be invited to JOG Update 7 <sup>th</sup> Feb 19 – David Scannell to attend Joint operational group meeting in March 21.3.19 Update: David Scannell has confirmed for April meeting via JH	SS	JL	Ongoing
SSNHSE 18/65	SH/AP will forward future minutes from the Workforce Steering Group for circulation to the Committee  Update 18 <sup>th</sup> October 2018 – Meeting has not taken place yet.  Update 20 <sup>th</sup> December 2018 – Meeting has not taken place.  Update 7 <sup>th</sup> Feb 2019 – Sharon not available at meeting ask next time  Update 21.3.19 – no minutes available at this point	ss	SH/AP	ongoing
SFSSNHSE 19/06	A report is to be drawn up showing the good work the 7 day access service has provided. The report will be presented to the Governing Body 21.3.19 Update: SM to be asked to produce report	Both	JL/AP	March 19
SFSSNHSE 19/06	GPFV programme report will be reviewed and updated as some work streams could be amalgamated.  21.3.19 Update: report reviewed and updated. Item closed	Both	AP	March 19
JCCiC 19/29	JL to escalate and monitor telephone call charges when patients are held in queuing systems when contacting their GP	SF	JL	May 19



## **Primary Care Commissioning Committee**

#### **Terms of Reference**

In accordance with its statutory powers under section 13Z of the National Health Service Act 2006 (as amended), NHS England has delegated the exercise of the functions specified in Schedule 2 to NHS Southport & Formby CCG. The delegation is set out in Schedule 1 of the NHS Act.

The CCG has established the NHS Southport & Formby CCG Primary Care Commissioning Committee (the "committee"). The committee will function as a corporate decision-making body for the management of the delegated functions and the exercise of the delegated powers.

The committee is established in accordance with NHS Southport and Formby CCG's constitution, standing orders and schemes of reservation and delegation. These terms of reference set out the membership, remit, responsibilities and reporting arrangements of the committee.

The committee will oversee the effective commissioning of primary medical services and will provide assurances to the governing bodies on the arrangements in place and the outcomes achieved as a result.

It is a committee comprising representatives of the following organisations:

- NHS Southport and Formby CCG
- NHS England

This committee along with the Primary Care Commissioning Committee of NHS South Sefton CCG will meet as committees in common.

#### **Statutory Framework**

NHS England has delegated to the CCG, authority to exercise the primary care commissioning functions set out in Schedule 2 in accordance with section 13Z of the NHS Act.

Arrangements made under section 13Z may be on such terms and conditions (including terms as to payment) as agreed within the Delegation Agreement.

Arrangements made under section 13Z do not affect the liability of NHS England for the exercise of any of its functions. However, the CCGs acknowledge that in exercising its functions (including those delegated to them), it must comply with the statutory duties set out in Chapter A2 of the NHS Act and including:

- a) Management of conflicts of interest (section 140);
- b) Duty to promote the NHS Constitution (section 14P);
- c) Duty to exercise its functions effectively, efficiently and economically (section 14Q);
- d) Duty as to improvement in quality of services (section 14R);

- e) Duty in relation to quality of primary medical services (section 14S);
- f) Duties as to reducing inequalities (section 14T);
- g) Duty to promote the involvement of each patient (section 14U);
- h) Duty as to patient choice (section 14V);
- i) Duty as to promoting integration (section 14Z1);
- j) Public involvement and consultation (section 14Z2).

The committee will also exercise the NHS England duties set out below:

- Duty to have regard to impact on services in certain areas (section 130);
- Duty as respects variation in provision of health services (section 13P).

The committee is established as a committee of the governing body of NHS Southport and Formby CCG in accordance with Schedule 1A of the NHS Act and in accordance with the relevant provisions as set out in the CCG's constitution.

The committee members acknowledge that the committee is subject to any directions made by NHS England or by the Secretary of state.

#### Role of the committee

NHS Southport and Formby CCG Primary Care Commissioning Committee (PCCC) will meet at the same time, in the same place with a shared agenda with NHS South Sefton CCG Primary Care Commissioning Committee (PCCC) and discharge the business under a "committees in common" arrangement. However, each respective PCCC remains accountable for decisions pertaining to their relevant CCG. The terms of reference, roles and responsibilities of each of the CCG's PCCC are aligned to ensure that "committees in common" meeting forum is able to function optimally.

The role of the NHS Southport and Formby CCG PCC is set out below:

The NHS Southport and Formby PCCC has been established to enable the members to make collective decisions on the review, planning and procurement of primary care services in Southport and Formby under delegated authority from NHS England.

In performing its role the committee will exercise its management of the functions in accordance with the delegation agreement entered into between NHS England and NHS Southport and Formby CCG and the delegation agreement entered into between NHS England and NHS South Sefton CCG.

The functions of the committee are undertaken in the context of a desire increase quality, efficiency, productivity and value for money and to remove administrative barriers.

The role of the committee shall be to carry out the functions relating to the commissioning of primary medical services under section 83 of the NHS Act.

This includes the following:

GMS, PMS and APMS contracts (including the design of PMS and APMS contracts,

- monitoring of contracts, taking contractual action such as issuing breach/remedial notices, and removing a contract)
- Newly designed enhanced services ("Local Enhanced Services" and "Directed Enhanced Services")
- Design of the Local Quality Contract
- Design of local incentive schemes as an alternative to the Quality Outcomes Framework (QOF)
- Decision making on whether to establish new GP practices in an area
- · Approving practice mergers
- Making decisions on 'discretionary' payment (e.g. returner/retainer schemes).

#### The CCGs will also carry out the following activities:

- To plan, including needs assessment, primary [medical] care services in Southport and Formby.
- To undertake reviews of primary [medical] care services in Southport and Formby.
- To co-ordinate a common approach to the commissioning of primary care services generally
- To manage the budget for commissioning of primary [medical] care services in Southport and Formby accordance with the schemes of reservation and delegation.

#### Membership

- 1 x lay member Chair
- 1 x lay member Vice chair
- Chief Officer or nominated deputy
- Director of Commissioning and Re-design or nominated deputy
- Programme Lead for Primary Care
- Chief Nurse and Quality Officer or nominated deputy
- Chief Finance Officer or nominated deputy
- NHSE representative

The Chair of the committee shall be agreed by the governing body and will be elected from its lay membership. Where one CCG lay member holds the committee Chair position, the vice chair position will be occupied by an additional lay member.

To support the "committees in common" meeting arrangement the Chair of Southport and Formby CCG PCCC and South Sefton CCG PCCC will act as "facilitator" on a six monthly rotational basis and preside over the running and conduct of the "committees in common" meeting.

In all cases the substantive decision making, as and when required, will remain with the relevant CCG's PCCC and voting undertaken by the respective members only.

The following will be included on the committee as non-voting attendees:

- 1 GP clinical lead Southport and Formby
- LMC representative
- Healthwatch Representative
- Health and Welling Being representative

 Other relevant officers and external advisors will be invited to attend to ensure and enable the delivery of the functions of the committee.

#### **Meetings and Voting**

The committee will operate in accordance with the CCG's standing orders.

The secretary, who shall be the secretary to support the committees in common arrangements, will be responsible for giving notice of meetings. This will be accompanied by an agenda and supporting papers and sent to each member representative no later than seven (7) working days before the date of the meeting.

When the Chair of the committee deems it necessary in light of the urgent circumstances to call a meeting at short notice, the notice period shall be such as they shall specify. If the matter is relevant to both CCGs, then a "committees in common" arrangement shall apply. In the event the matter is relevant to a single CCG then it will not be necessary to convene a "committees in common arrangement".

Each member of the NHS Southport and Formby CCG PCCC shall have one vote. The Committee shall reach decisions by a simple majority of members present, The aim of the committee will be to achieve consensus decision-making wherever possible.

In the event that a vote is required the Chair shall have the casting vote.

As this meeting is held in public there will at times be a need to complete some of the business of the committee in private and this part of the meeting and its agenda items will be closed to the public. NHS Southport and Formby CCG is open and transparent in its decision making and aims to ensure that all appropriate information is in the public domain. Unfortunately some of the business transacted can be commercial and in confidence and this work will form the private business of the committee.

#### Quorum

The Chair or Vice Chair of the committee must be present and three other members, at least one of which must be a member of the CCG Leadership Team of the CCG and one must be a clinician.

#### Frequency of meetings

The Committee shall meet bi-monthly and will meet as part of a "committees in common" arrangement with NHS South Sefton CCG PCCC.

Members of the committee have a collective responsibility for the operation of the committee. They will participate in discussion, review evidence and provide objective expert input to the best of their knowledge and ability, and endeavour to reach a collective view.

Additionally members will be expected to:

- · Attend meetings, having read all papers beforehand
- Act as 'champions', disseminating information and good practice as appropriate
- Identify agenda items to the Secretary of the committees in common at least fifteen working days before the meeting
- Submit papers at least ten (10) working days before the meeting
- Make open and honest declarations of their interests at the commencement of each meeting notifying the committee Chair of any agreed management arrangements, or to notify the committee Chair of any actual, potential or perceived conflict in advance of the meeting

• Uphold the Nolan Principles and all other relevant NHS Code of Conduct requirements

The Committee may delegate tasks to such individuals, sub-groups or individual members as it shall see fit, provided that any such delegations are consistent with the parties' relevant governance arrangements, are recorded in a scheme of delegation, are governed by terms of reference as appropriate and reflect appropriate arrangements for the management of conflicts of interest. In all circumstances the Southport and Formby PCCC will endeavour to seek agreement with South Sefton PCCC of any such delegations to support consistency of approach and continued alignment of ways of working.

Members of the committee shall respect confidentiality requirements as set out in the CCG's Standards of Business Conduct.

The committee will present its minutes to Cheshire and Merseyside sub regional teams of NHS England and the governing bodies of NHS Southport and Formby CCG each month or after each committee for information, including the minutes of any sub-groups to which responsibilities are delegated.

The CCGs will also comply with any reporting requirements set out in its constitution. These terms of reference will be reviewed on an annual basis.

#### Accountability of the committee

Budget and resource accountability arrangements and the decision-making scope of the committee have been agreed and are laid out with the scheme of reservation and delegation that have been approved by the CCG.

For the avoidance of doubt, in the event of any conflict between the terms of the Delegation Agreement, the committee Terms of Reference and the Standing Orders or Standing Financial Instructions of any of the members, the Delegation Agreement shall prevail.

The committee will publish, in line with national requirements all committee procurement decisions onto NHS Southport and Formby CCG Website.

The committee shall ensure that the CCGs engage and consult with the public and its members in the delivery of its functions.

#### **Procurement of Agreed Services**

The detailed arrangements regarding procurement are set out in the Delegation Agreement.

The Southport and Formby CCG Primary Care Commissioning Committee will make procurement decisions relevant to the exercise of the Delegated Functions and in accordance with the detailed arrangements regarding procurement set out in the procurement protocol issued and updated by NHS England from time to time.

In discharging its responsibilities set out in Performance of the Delegated Functions, of the Delegation Agreement and paragraph 1 of Schedule 2 (Delegated Functions), the CCG must comply at all times with Law including its obligations set out in the National Health Service (Procurement, Patient Choice and Competition) (No. 2) Regulations 2013/500 and any other relevant statutory provisions. The CCG shall have regard to any relevant guidance, particularly Monitor's guidance Substantive guidance on the Procurement, Patient Choice and Competition Regulations

#### **Decisions**

The Committee will make decisions within the bounds of its remit. The decisions of the committee

shall be binding on NHS England and NHS Southport and Formby CCG.

The Committee will produce an executive summary report which will be presented to NHS England North (Cheshire and Merseyside) and the governing bodies of NHS Southport and Formby CCG each month/after each committees in common meeting for information.

The Committee shall publish its commissioning decisions on the CCGs websites in line with the requirements of transparency and management of conflicts.

#### **Administrative Arrangements**

The committees in common will be supported by an appropriate Secretary that will be responsible for supporting the Chair that is in occupying the "facilitation role" in the management of the committee's business. The existing roles and responsibilities of the substantive Chair of the respective CCG's PCCC remain extant and are unchanged regardless of which Chair is occupying the committees in common "facilitation role".

The Secretary to the committees in common will ensure:

- Correct minutes are taken and once agreed by the Chair of each CCG's PCCC,
   distributing minutes to the members within seven working days of the meeting taking place
- A Key Issues report is produced following the meeting and submitted to the next meeting
  of the governing body
- An Action Log is produced following each meeting and any outstanding actions are carried forward until complete
- The agenda and accompanying papers are distributed to members at least five working days in advance of the meeting date
- They provide appropriate support to the Chair and Committee members
- The papers of the committee are filed in accordance with NHS Southport and Formby CCG policies and procedures
- The meetings of this Committee will be held in public to enable transparency of decision making. The guidance and arrangement for this is in Appendix 1
- The Work Plan will be agreed at the start of each financial year and will be approved by the Governing Body. The focus of the work programme will be in line with the strategic objectives of the CCG.

Version 2: April 2019 Review Date: April 2020.



### **Primary Care Commissioning Committee**

#### **Terms of Reference**

In accordance with its statutory powers under section 13Z of the National Health Service Act 2006 (as amended), NHS England has delegated the exercise of the functions specified in Schedule 2 to NHS South Sefton CCG. The delegation is set out in Schedule 1 of the NHS Act.

The CCG has established the NHS South Sefton CCG Primary Care Commissioning Committee (the "committee"). The committee will function as a corporate decision-making body for the management of the delegated functions and the exercise of the delegated powers.

The committee is established in accordance with NHS South Sefton CCG's constitution, standing orders and schemes of reservation and delegation. These terms of reference set out the membership, remit, responsibilities and reporting arrangements of the committee.

The committee will oversee the effective commissioning of primary medical services and will provide assurances to the governing bodies on the arrangements in place and the outcomes achieved as a result.

It is a committee comprising representatives of the following organisations:

- NHS South Sefton CCG
- NHS England

This committee along with the Primary Care Commissioning Committee of NHS Southport and Formby CCG will meet as committees in common.

#### **Statutory Framework**

NHS England has delegated to the CCG, authority to exercise the primary care commissioning functions set out in Schedule 2 in accordance with section 13Z of the NHS Act.

Arrangements made under section 13Z may be on such terms and conditions (including terms as to payment) as agreed within the Delegation Agreement.

Arrangements made under section 13Z do not affect the liability of NHS England for the exercise of any of its functions. However, the CCGs acknowledge that in exercising its functions (including those delegated to them), it must comply with the statutory duties set out in Chapter A2 of the NHS Act and including:

- a) Management of conflicts of interest (section 140);
- b) Duty to promote the NHS Constitution (section 14P);

- c) Duty to exercise its functions effectively, efficiently and economically (section 14Q);
- d) Duty as to improvement in quality of services (section 14R);
- e) Duty in relation to quality of primary medical services (section 14S);
- f) Duties as to reducing inequalities (section 14T);
- g) Duty to promote the involvement of each patient (section 14U);
- h) Duty as to patient choice (section 14V);
- i) Duty as to promoting integration (section 14Z1);
- j) Public involvement and consultation (section 14Z2).

The committee will also exercise the NHS England duties set out below:

- Duty to have regard to impact on services in certain areas (section 130);
- Duty as respects variation in provision of health services (section 13P).

The committee is established as a committee of the governing body of NHS South Sefton CCG in accordance with Schedule 1A of the NHS Act and in accordance with the relevant provisions as set out in the CCG's constitution.

The committee members acknowledge that the committee is subject to any directions made by NHS England or by the Secretary of state.

#### Role of the committee

NHS South Sefton CCG Primary Care Commissioning Committee (PCCC) will meet at the same time, in the same place with a shared agenda with NHS Southport and Formby CCG Primary Care Commissioning Committee (PCCC) and discharge the business under a "committees in common" arrangement. However, each respective PCCC remains accountable for decisions pertaining to their relevant CCG. The terms of reference, roles and responsibilities of each of the CCG's PCCC are aligned to ensure that "committees in common" meeting forum is able to function optimally.

The role of the NHS South Sefton CCG PCCC is set out below:

The NHS South Sefton PCCC has been established to enable the members to make collective decisions on the review, planning and procurement of primary care services in South Sefton under delegated authority from NHS England.

In performing its role the committee will exercise its management of the functions in accordance with the delegation agreement entered into between NHS England and NHS South Sefton CCG.

The functions of the committee are undertaken in the context of a desire increase quality, efficiency, productivity and value for money and to remove administrative barriers.

The role of the committee shall be to carry out the functions relating to the commissioning of primary medical services under section 83 of the NHS Act.

#### This includes the following:

- GMS, PMS and APMS contracts (including the design of PMS and APMS contracts, monitoring of contracts, taking contractual action such as issuing breach/remedial notices, and removing a contract)
- Newly designed enhanced services ("Local Enhanced Services" and "Directed Enhanced Services")
- Design of the Local Quality Contract
- Design of local incentive schemes as an alternative to the Quality Outcomes Framework (QOF)
- Decision making on whether to establish new GP practices in an area
- Approving practice mergers
- Making decisions on 'discretionary' payment (e.g. returner/retainer schemes).

#### The CCGs will also carry out the following activities:

- To plan, including needs assessment, primary [medical] care services in South Sefton.
- To undertake reviews of primary [medical] care services in South Sefton.
- To co-ordinate a common approach to the commissioning of primary care services generally
- To manage the budget for commissioning of primary [medical] care services in South Sefton accordance with the schemes of reservation and delegation.

#### Membership

- 1 x lay member Chair
- 1 x lay member Vice chair
- Chief Officer or nominated deputy
- Director of Commissioning and Re-design or nominated deputy
- Programme Lead for Primary Care
- Chief Nurse and Quality Officer or nominated deputy
- Chief Finance Officer or nominated deputy
- NHSE representative

The Chair of the committee shall be agreed by the governing body and will be elected from its lay membership. Where one CCG lay member holds the committee Chair position, the vice chair position will be occupied by an additional lay member.

To support the "committees in common" meeting arrangement the Chair of South Sefton CCG PCCC and NHS Southport and Formby CCG PCCC will act as "facilitator" on a six monthly rotational basis and preside over the running and conduct of the "committees in common" meeting.

In all cases the substantive decision making, as and when required, will remain with the relevant CCG's PCCC and voting undertaken by the respective members only.

The following will be included on the committee as non-voting attendees:

- 1 GP clinical lead South Sefton
- LMC representative
- Healthwatch Representative
- Health and Welling Being representative
- Other relevant officers and external advisors will be invited to attend to ensure and enable the delivery of the functions of the committee.

#### **Meetings and Voting**

The committee will operate in accordance with the CCG's standing orders.

The secretary, who shall be the secretary to support the committees in common arrangements, will be responsible for giving notice of meetings. This will be accompanied by an agenda and supporting papers and sent to each member representative no later than seven (7) working days before the date of the meeting.

When the Chair of the committee deems it necessary in light of the urgent circumstances to call a meeting at short notice, the notice period shall be such as they shall specify. If the matter is relevant to both CCGs, then a "committees in common" arrangement shall apply. In the event the matter is relevant to a single CCG then it will not be necessary to convene a "committees in common arrangement".

Each member of the NHS South Sefton CCG PCCC shall have one vote. The Committee shall reach decisions by a simple majority of members present, The aim of the committee will be to achieve consensus decision-making wherever possible.

In the event that a vote is required the Chair shall have the casting vote.

As this meeting is held in public there will at times be a need to complete some of the business of the committee in private and this part of the meeting and its agenda items will be closed to the public. NHS South Sefton CCG is open and transparent in its decision making and aims to ensure that all appropriate information is in the public domain. Unfortunately some of the business transacted can be commercial and in confidence and this work will form the private business of the committee.

#### Quorum

The Chair or Vice Chair of the committee must be present and three other members, at least one of which must be a member of the CCG Leadership Team of the CCG and one must be a clinician.

#### Frequency of meetings

The Committee shall meet bi-monthly and will meet as part of a "committees in common" arrangement with NHS Southport and Formby CCG PCCC.

Members of the committee have a collective responsibility for the operation of the committee. They will participate in discussion, review evidence and provide objective expert input to the best of their knowledge and ability, and endeavour to reach a collective view.

Additionally members will be expected to:

- Attend meetings, having read all papers beforehand
- Act as 'champions', disseminating information and good practice as appropriate
- Identify agenda items to the Secretary of the committees in common at least fifteen working days before the meeting
- Submit papers at least ten (10) working days before the meeting
- Make open and honest declarations of their interests at the commencement of each meeting notifying the committee Chair of any agreed management arrangements, or to notify the committee Chair of any actual, potential or perceived conflict in advance of the meeting
- Uphold the Nolan Principles and all other relevant NHS Code of Conduct requirements

The Committee may delegate tasks to such individuals, sub-groups or individual members as it shall see fit, provided that any such delegations are consistent with the parties' relevant governance arrangements, are recorded in a scheme of delegation, are governed by terms of reference as appropriate and reflect appropriate arrangements for the management of conflicts of interest. In all circumstances the South Sefton PCCC will endeavour to seek agreement with NHS Southport and Formby PCCC of any such delegations to support consistency of approach and continued alignment of ways of working.

Members of the committee shall respect confidentiality requirements as set out in the CCG's Standards of Business Conduct.

The committee will present its minutes to Cheshire and Merseyside sub regional teams of NHS England and the governing bodies of NHS South Sefton CCG each month or after each committee for information, including the minutes of any sub-groups to which responsibilities are delegated.

The CCGs will also comply with any reporting requirements set out in its constitution. These terms of reference will be reviewed on an annual basis.

#### Accountability of the committee

Budget and resource accountability arrangements and the decision-making scope of the committee have been agreed and are laid out with the scheme of reservation and delegation that have been approved by the CCG.

For the avoidance of doubt, in the event of any conflict between the terms of the Delegation Agreement, the committee Terms of Reference and the Standing Orders or Standing Financial Instructions of any of the members, the Delegation Agreement shall prevail.

The committee will publish, in line with national requirements all committee procurement decisions onto NHS South Sefton CCG Website.

The committee shall ensure that the CCGs engage and consult with the public and its members in the delivery of its functions.

#### **Procurement of Agreed Services**

The detailed arrangements regarding procurement are set out in the Delegation Agreement.

The South Sefton CCG Primary Care Commissioning Committee will make procurement decisions relevant to the exercise of the Delegated Functions and in accordance with the detailed arrangements regarding procurement set out in the procurement protocol issued and updated by NHS England from time to time.

In discharging its responsibilities set out in Performance of the Delegated Functions, of the Delegation Agreement and paragraph 1 of Schedule 2 (Delegated Functions), the CCG must comply at all times with Law including its obligations set out in the National Health Service (Procurement, Patient Choice and Competition) (No. 2) Regulations 2013/500 and any other relevant statutory provisions. The CCG shall have regard to any relevant guidance, particularly Monitor's guidance Substantive guidance on the Procurement, Patient Choice and Competition Regulations

#### **Decisions**

The Committee will make decisions within the bounds of its remit. The decisions of the committee shall be binding on NHS England and NHS South Sefton CCG.

The Committee will produce an executive summary report which will be presented to NHS England North (Cheshire and Merseyside) and the governing bodies of NHS South Sefton CCG each month/after each committees in common meeting for information.

The Committee shall publish its commissioning decisions on the CCGs websites in line with the requirements of transparency and management of conflicts.

#### **Administrative Arrangements**

The committees in common will be supported by an appropriate Secretary that will be responsible for supporting the Chair that is in occupying the "facilitation role" in the management of the committee's business. The existing roles and responsibilities of the substantive Chair of the respective CCG's PCCC remain extant and are unchanged regardless of which Chair is occupying the committees in common "facilitation role".

The Secretary to the committees in common will ensure:

- Correct minutes are taken and once agreed by the Chair of each CCG's PCCC, distributing minutes to the members within seven working days of the meeting taking place
- A Key Issues report is produced following the meeting and submitted to the next meeting of the governing body
- An Action Log is produced following each meeting and any outstanding actions are

carried forward until complete

- The agenda and accompanying papers are distributed to members at least five working days in advance of the meeting date
- They provide appropriate support to the Chair and Committee members
- The papers of the committee are filed in accordance with NHS South Sefton CCG policies and procedures
- The meetings of this Committee will be held in public to enable transparency of decision making. The guidance and arrangement for this is in Appendix 1
- The Work Plan will be agreed at the start of each financial year and will be approved by the Governing Body. The focus of the work programme will be in line with the strategic objectives of the CCG.

Version 2: April 2019 Review Date: April 2020.



Primary Care Commissioning Committee April 2019			
Agenda Item: 19.07	Author of the Paper:		
Report date: 4 <sup>th</sup> April 2019	Angela Price Primary Care Programme Lead angela.price@southseftonccg.nhs.uk Tel: 01513178379	<u>.</u>	
Title: Southport and Formby Delegation Agreem	ent		
Summary/Key Issues:			
Recommendation The Primary Care Commissioning Committee is asl Delegation Agreement	ked to note the attached	Note Approve Ratify	х

Link	s to Corporate Objectives (x those that apply)
х	To focus on the identification of QIPP (Quality, Improvement, Productivity & Prevention) schemes and the implementation and delivery of these to achieve the CCG QIPP target and to support delivery of financial recovery.
х	To progress Shaping Sefton as the strategic plan for the CCG, in line with the NHSE planning requirements set out in the "Five Year Forward View", underpinned by transformation through the agreed strategic blueprints and programmes and as part of the NHS Cheshire and Merseyside Healthcare Partnership.
х	To ensure that the CCG maintains and manages performance & quality across the mandated constitutional measures.
Х	To support Primary Care Development through the development of an enhanced model of care and supporting estates strategy, underpinned by a complementary primary care quality contract.
Х	To advance integration of in-hospital and community services in support of the CCG locality model of care.
	To advance the integration of Health and Social Care through collaborative working with Sefton Metropolitan Borough Council, supported by the Health and Wellbeing Board.

Process	Yes	No	N/A	Comments/Detail (x those that apply)
Patient and Public Engagement			х	
Clinical Engagement			Х	
Equality Impact Assessment			х	
Legal Advice Sought			Х	
Resource Implications Considered			х	
Locality Engagement			Х	
Presented to other Committees			х	

Links	Links to National Outcomes Framework (x those that apply)		
х	Preventing people from dying prematurely		
Х	Enhancing quality of life for people with long-term conditions		
Х	Helping people to recover from episodes of ill health or following injury		
Х	Ensuring that people have a positive experience of care		
Х	Treating and caring for people in a safe environment and protecting them from avoidable harm		

### **Delegation Agreement**

#### 1. Particulars

1.1. This Agreement records the particulars of the agreement made between NHS England and the Clinical Commissioning Group named below.

Area Southport and Formby

Clinical Commissioning Group NHS Southport and Formby Clinical

**Commissioning Group** 

CCG Representative Fiona Taylor, Chief Officer

CCG Address for Notices 5 Curzon Road, Southport, PR8

6PL

Date of Agreement [Leave Blank]

Delegation means the delegation made by NHS

England to the CCG of certain functions relating to primary medical services under section 13Z of the NHS Act and effective from 1 April 2015 (as amended pursuant to the Delegation)

NHS England Representative [Insert details of name of manager of

this Agreement for NHS England]

Local NHS England Team NHS England North (Cheshire and

Merseyside)

NHS England Address for Regatta Place

Notices Brunswick Business Park

Summers Road LIVERPOOL L3 4BL

1.2. This Agreement comprises:

1.2.1. the Particulars (Clause 1);

1.2.2. the Terms and Conditions (Clauses 2 to 24 and Schedule 1 to Schedule 6 and Schedule 8 to this Agreement); and

1.2.3. the Local Terms (Schedule 7).

Signed by

**NHS England** 

(for and on behalf of NHS England)

Thoma Taylor.

Signed by

**NHS Southport and Formby Clinical Commissioning** 

Group

**Fiona Taylor** 

Chief Officer (for and on behalf of NHS Southport

and Formby Wider Group (the Membership)

#### **Terms and Conditions**

#### A. Introduction

#### 2. Interpretation

- 2.1. This Agreement is to be interpreted in accordance with Schedule 1 (Definitions and Interpretation).
- 2.2. If there is any conflict or inconsistency between the provisions of this Agreement and the provisions of the Delegation, the provisions of the Delegation will prevail.
- 2.3. If there is any conflict or inconsistency between the provisions of this Agreement, that conflict or inconsistency must be resolved according to the following order of priority:
  - 2.3.1. the Particulars and Terms and Conditions (Clauses 1 to 24 and, in particular, clause 8.7);
  - 2.3.2. Schedule 1 to Schedule 6 and Schedule 8 to this Agreement; and
  - 2.3.3. Schedule 7 (Local Terms).
- 2.4. This Agreement and any ancillary agreements it refers to constitute the entire agreement and understanding between the Parties relating to the Delegation and supersedes all previous agreements, promises and understandings between them, whether written or oral, relating to its subject matter.

#### 3. Background

- 3.1. NHS England has delegated the Delegated Functions to the CCG under section 13Z of the NHS Act and as set out in the Delegation.
- 3.2. Arrangements made under section 13Z of the NHS Act may be made on such terms and conditions (including terms as to payment) as may be agreed between NHS England and the CCG.

- 3.3. This Agreement sets out the arrangements that apply in relation to the exercise of the Delegated Functions by the CCG.
- 3.4. For the avoidance of doubt, functions relating to the commissioning of primary care pharmacy, dental and optical contracts are not delegated to the CCG under the Delegation. The Delegation relates only to the delegation and reservation of primary medical services commissioning functions as set out in this Agreement.

#### 4. Term

4.1. This Agreement has effect from the date set out in paragraph 5 of the Delegation and will remain in force unless terminated in accordance with clause 17 (*Termination*) below.

#### 5. Principles

- 5.1. In performing their obligations under this Agreement, NHS England and the CCG must:
  - 5.1.1. at all times act in good faith towards each other;
  - 5.1.2. at all times exercise functions effectively, efficiently and economically;
  - 5.1.3. act in a timely manner;
  - 5.1.4. share information and best practice, and work collaboratively to identify solutions, eliminate duplication of effort, mitigate risk and reduce cost;
  - 5.1.5. at all times observe relevant statutory powers, requirements and best practice to ensure compliance with applicable laws and standards including those governing procurement, and Information Law; and
  - 5.1.6. have regard to the needs and views of the other Party and as far as is lawful and reasonably practicable, take such needs and views into account.

#### B. Role of the CCG

#### 6. Performance of the Delegated Functions

- 6.1. The role of the CCG will be to exercise the Delegated Functions in the Area.
- 6.2. The Delegated Functions are the functions set out in Schedule 1of the Delegation and being:
  - 6.2.1. decisions in relation to the commissioning, procurement and management of Primary Medical Services Contracts, including but not limited to the following activities:
    - 6.2.1.1. decisions in relation to Enhanced Services;
    - 6.2.1.2. decisions in relation to Local Incentive Schemes (including the design of such schemes);
    - 6.2.1.3. decisions in relation to the establishment of new GP practices (including branch surgeries) and closure of GP practices;
    - 6.2.1.4. decisions about 'discretionary' payments;
    - 6.2.1.5. decisions about commissioning urgent care (including home visits as required) for out of area registered patients;
  - 6.2.2. the approval of practice mergers;
  - 6.2.3. planning primary medical care services in the Area, including carrying out needs assessments;
  - 6.2.4. undertaking reviews of primary medical care services in the Area;
  - 6.2.5. decisions in relation to the management of poorly performing GP practices and including, without limitation, decisions and liaison with the CQC where the CQC has reported non-compliance with standards (but excluding any decisions in relation to the performers list);
  - 6.2.6. management of the Delegated Funds in the Area;
  - 6.2.7. Premises Costs Directions Functions;
  - 6.2.8. co-ordinating a common approach to the commissioning of primary care services with other commissioners in the Area where appropriate; and
  - 6.2.9. such other ancillary activities that are necessary in order to exercise the Delegated Functions.

- 6.3. Schedule 2 (*Delegated Functions*) sets out further detail in relation to the Delegated Functions and the exercise of such Delegated Functions.
- 6.4. The CCG agrees that it must perform the Delegated Functions in accordance with:
  - 6.4.1. the Delegation;
  - 6.4.2. the terms of this Agreement;
  - 6.4.3. all applicable Law;
  - 6.4.4. the CCG's constitution;
  - 6.4.5. Statutory Guidance; and
  - 6.4.6. Good Practice.
- 6.4A The CCG must have due regard to Guidance and Contractual Notices.
- 6.5. Without prejudice to clause 6.4, the CCG agrees that it must perform the Delegated Functions in such a manner as to ensure NHS England's compliance with NHS England's statutory duties in respect of the Delegated Functions and to enable NHS England to fulfil its Reserved Functions.
- 6.6. When performing the Delegated Functions, the CCG will not do anything, take any step or make any decision outside of its delegated authority as set out in the Delegation.
- 6.7. Without prejudice to any other provision in this Agreement, the CCG must comply with the NHS England central finance team's operational process (as such process is updated from time to time) for the reporting and accounting of the Delegated Funds. In particular, the CCG will be required to permit the NHS England central finance team and/or their agents and contractors authorised by them to have the ability to access the CCG ledger to provide the services required to deliver financial support and assistance to the CCG necessary to enable them to manage the Delegated Funds and exercise the Delegated Functions. NHS England and the CCG will agree any accruals to be made including any adjustments related to the relevant Financial Year expenditure to ensure no net financial impact or gain on the CCG.
- 6.8. The decisions of the CCG in exercising the Delegated Functions will be binding on the CCG and NHS England.

#### 7. Committee

- 7.1. The CCG must establish a committee to exercise its Delegated Functions.
- 7.2. The structure and operation of the committee must be constituted so as to take into account Guidance issued by NHS England including the revised statutory guidance on managing conflicts of interest for CCGs https://www.england.nhs.uk/commissioning/pc-co-comms/coi/

### C. Functions reserved to NHS England

#### 8. Performance of the Reserved Functions

- 8.1. The role of NHS England will be to exercise the Reserved Functions.
- 8.2. Subject to clause 8.3, the Reserved Functions are all of NHS England's functions relating to primary medical services other than the Delegated Functions and including those functions set out in Schedule 2of the Delegation and being:
  - 8.2.1. management of the national performers list;
  - 8.2.2. management of the revalidation and appraisal process;
  - 8.2.3. administration of payments in circumstances where a performer is suspended and related performers list management activities;
  - 8.2.4. Capital Expenditure Functions;
  - 8.2.5. Section 7A Functions;
  - 8.2.6. functions in relation to complaints management;
  - 8.2.7. decisions in relation to the GP Access Fund; and
  - 8.2.8. such other ancillary activities that are necessary in order to exercise the Reserved Functions.
- 8.3. For the avoidance of doubt, the Parties acknowledge that the Delegation may be amended and additional functions may be delegated to the CCG, in which event consequential changes to this Agreement

- shall be agreed with the CCG pursuant to clause 22 (Variations) of this Agreement.
- 8.4. Schedule 3 (*Reserved Functions*) sets out further detail in relation to the Reserved Functions.
- 8.5. To support and assist NHS England in carrying out the Reserved Functions, the CCG will share information with NHS England in accordance with section E (*Information*) below.
- 8.6. NHS England will work collaboratively with the CCG when exercising the Reserved Functions, including discussing with the CCG how it proposes to address GP performance issues.
- 8.7. If there is any conflict or inconsistency between functions that are named as Delegated Functions and functions that are named as Reserved Functions then such functions shall be interpreted as Reserved Functions.
- 8.8. The Parties acknowledge that, as at the date of this Agreement, the CCG shall provide administrative and management services to NHS England in relation to certain Reserved Functions and that such administrative and management services are as follows:
  - 8.8.1. the administrative and management services in relation to the Capital Expenditure Functions and the Capital Expenditure Funds as more particularly set out in clauses 13.13 to 13.16; and
  - 8.8.2. the administrative and management services in relation to the Section 7A Functions and Section 7A Funds as more particularly set out in clauses 13.17 to 13.20.
- 8.9. The Parties further acknowledge that NHS England may ask the CCG to provide certain administrative and management services to NHS England in relation to other Reserved Functions as more particularly set out in clauses 13.21 to 13.23. Such administrative and management services shall only be provided by the CCG following agreement by the CCG.

8.10. Notwithstanding any arrangement for or provision of administrative or management services in respect of certain Reserved Functions, NHS England shall retain and be accountable for the exercise of such Reserved Functions.

### D. Commissioning

#### 9. Monitoring and Reporting - General Requirements

- 9.1. The CCG must comply with any reporting requirements under:
  - 9.1.1. this Agreement (including, without limitation, as required by clause 9 (Monitoring and Reporting General Requirements), clause 12 (Public Information and Access Targets), clause 13 (Financial Provisions and Liability), clause 14 (Claims and Litigation) and Schedule 2 Part 1 paragraph 2 (Primary Medical Services Contract Management) and paragraph 5 (Information Sharing with NHS England));
  - 9.1.2. the CCG Assurance Framework; and
  - 9.1.3. the CCG's constitution.
- 9.2. NHS England shall monitor the exercise and carrying out of the Delegated Functions by the CCG under the terms of this Agreement and as part of the CCG Assurance Framework.
- 9.3. The CCG will notify NHS England of all primary medical services commissioning committee meetings at least seven (7) days in advance of such meetings and NHS England will be entitled to attend such meetings at its discretion.
- 9.4. The CCG must provide to NHS England:
  - 9.4.1. all information in relation to the exercise of the Delegated Functions (including in relation to the Delegation or this Agreement), (and in such form) as requested by NHS England from time to time; and
  - 9.4.2. all such information (and in such form), that may be relevant to NHS England in relation to the exercise by NHS England of its other duties or functions including, without limitation, the Reserved Functions.

9.5. Nothing in this Agreement shall affect NHS England's power to require information from the CCG under sections 14Z17, 14Z18, 14Z19 and 14Z20 of the NHS Act.

#### E. Information

#### 10. Information Sharing and Information Governance

- 10.1. Schedule 4 (Further Information Sharing Provisions) makes further provision about information sharing and information governance.
- 10.2. NHS England and the CCG will enter into a Personal Data Agreement that will describe the processing of Relevant Information that identifies individuals under this Agreement. A template Personal Data Agreement is set out in Schedule 4 (Further Information Sharing Provisions).
- 10.3. The Personal Data Agreement:
  - 10.3.1. sets out the relevant Information Law and best practice, including the requirements of the NHS Digital IG Toolkit;
  - 10.3.2. sets out how that law and best practice will be implemented, including responsibilities of the Parties to co-operate properly and fully with each other;
  - 10.3.3. identifies the Relevant Information that may be processed, including what may be shared, under this Agreement;
  - 10.3.4. identifies the purposes for which the Relevant Information may be so processed and states the legal basis for the processing in each case;
  - 10.3.5. states who is/are the data controller/s and, if appropriate, the data processor/s of Personal Data;
  - 10.3.6. sets out what will happen to the Personal Data on the termination of this Agreement (with due regard to clause 17 (*Termination*) of the Agreement); and
  - 10.3.7. sets out such other provisions as are necessary for the sharing of Relevant Information to be fair, lawful and meet best practice.
- 10.4. NHS England and the CCG will share all Non-Personal Data in accordance with Information Law and their statutory powers as set out

- in section 13Z3 (for NHS England) and section 14Z23 (for the CCG) of the NHS Act.
- 10.5. The Parties agree that, in relation to information sharing and the processing of Relevant Information under the Delegation and this Agreement, they must comply with:
  - 10.5.1. all relevant Information Law requirements including the common law duty of confidence (unless disapplied by statute) and other legal obligations in relation to information sharing including those set out in the NHS Act and the Human Rights Act 1998:
  - 10.5.2. Good Practice; and
  - 10.5.3. relevant guidance (including guidance given by the Information Commissioner, the Caldicott Principles, the requirements of the NHS Information Governance Toolkit to level 2, and guidance issued further to sections 263 and 265 of the HSCA) and consistent with guidance issued under section 13S of the NHS Act to providers.

# 11. IT inter-operability

- 11.1. NHS England and the CCG will work together to ensure that all relevant IT systems operated by NHS England and the CCG in respect of the Delegated Functions and the Reserved Functions are inter-operable and that data may be transferred between systems securely, easily and efficiently.
- 11.2. The Parties will use their respective reasonable endeavours to help develop initiatives to further this aim.

### 12. Public Information and Access Targets

12.1. The CCG must promptly make available to NHS England such information as is required in respect of the Delegated Functions to ensure NHS England's discharge of its statutory duties.

- 12.2. The CCG must ensure that all new Primary Medical Services Contracts contain appropriate provisions such that the CCG is able to discharge its obligations in clause 12.1.
- 12.3. The CCG must ensure that any information provided under this Agreement complies with all relevant national data sets issued by NHS England and NHS Digital.

# F. General

## 13. Financial Provisions and Liability

Notification of the Delegated Funds and Adjustments to the Delegated Funds

- 13.1. NHS England will, in respect of each Financial Year, notify the CCG of the proportion of the funds allocated to NHS England by the Secretary of State pursuant to Chapter 6 of the NHS Act and which are to be paid to the CCG for the purpose of meeting expenditure in respect of the Delegated Functions for that Financial Year (the "Delegated Funds").
- 13.2. Except in relation to pooled funds and subject to the terms of this clause 13 (*Financial Provisions and Liability*) and, in particular, clause 13.4, the CCG must use the Delegated Funds to meet expenditure in respect of the exercise of the Delegated Functions. Without prejudice to the generality of the foregoing, the CCG must make:
  - 13.2.1. all payments in relation to the Primary Medical Services Contracts including payments in relation to QOF and implementing financial adjustments or sanctions (including in relation to breaches of provider obligations); and
  - 13.2.2. all payments under the Premises Costs Directions.
- 13.3. NHS England may, in any Financial Year by sending a notice to the CCG of such increase or decrease, increase or reduce the Delegated Funds:
  - 13.3.1. in order to take into account any monthly adjustments or corrections to the Delegated Funds that NHS England considers appropriate (following discussions with the CCG), including without limitation adjustments following any changes to the Delegation or Delegated Functions (including changes

- pursuant to paragraph 6 or paragraph 16 of the Delegation), changes in allocations, changes in contracts or otherwise;
- 13.3.2. in order to comply with a change in the amount allocated to NHS England by the Secretary of State pursuant to section 223B of the NHS Act:
- 13.3.3. to take into account any Losses arising under clause 13.35;
- 13.3.4. to take into account any Claim Losses;
- 13.3.5. to take into account any adjustments that NHS England considers appropriate (including without limitation in order to make corrections or otherwise to reflect notional budgets) to reflect funds transferred (or that should have been transferred) to the CCG in respect of the Delegated Funds and/or funds transferred (or that should have been transferred) to the CCG and in respect of which the CCG has management or administrative responsibility under clauses 13.13 to 13.23 of this Agreement; or
- 13.3.6. in order to ensure compliance by NHS England of its obligations under the NHS Act (including without limitation, Chapter 6 of the NHS Act) or the HSCA or any action taken or direction made by the Secretary of State under the NHS Act or the HSCA.
- 13.3A NHS England acknowledges that the intention of clause 13.3 is to reflect genuine corrections and adjustments to the Delegated Funds and may not be used to change the allocation of the Delegated Funds unless there are significant or exceptional circumstances that would require such corrections or adjustments (including but not limited to a change in the mandate published by the Department of Health or other external factors).
- 13.4. The CCG acknowledges that it must comply with its statutory financial duties, including those under sections 223H and 223I of the NHS Act to the extent that these sections apply in relation to the receipt of the Delegated Funds.
- 13.5. The CCG acknowledges its duty under section 14S of the NHS Act to assist and support NHS England in discharging its duty under section 13E so far as relating to securing continuous improvement in the quality of primary medical services and agrees that it shall take this duty into

- account in relation to the exercise of the Delegated Functions and the use of the Delegated Funds.
- 13.6. The CCG must ensure that it uses the Delegated Funds in such a way as to ensure that NHS England is able to fulfil its functions, including without limitation the Reserved Functions, effectively and efficiently in accordance with this Agreement.
- 13.7. NHS England may in respect of the Delegated Funds:
  - 13.7.1. notify the CCG of the capital resource limit and revenue resource limit that will apply in any Financial Year;
  - 13.7.2. notify the CCG regarding the payment of sums by the CCG to NHS England in respect of charges referable to the valuation or disposal of assets and such conditions as to records, certificates or otherwise;
  - 13.7.3. by notice, require the CCG to take such action or step in respect of the Delegated Funds, in order to ensure compliance by NHS England of its duties or functions under the NHS Act or the HSCA (including without limitation, Chapter 6 of the NHS Act) or any action taken or direction made by the Secretary of State under the NHS Act or the HSCA (including, without limitation, Chapter 6 of the NHS Act).
- 13.8. Schedule 5 (Financial Provisions and Decision Making Limits) sets out further financial provisions in respect of the exercise of the Delegated Functions and, in particular, Table 1 in Schedule 5 (Financial Provisions and Decision Making Limits) sets out certain financial limits and approvals required in relation to the exercise of the Delegated Functions. NHS England's Standing Financial Instructions shall be updated accordingly.

### Payment and Transfer

13.9. The CCG acknowledges that the Delegated Funds do not form part of and are separate to the funds allocated annually under section 223G of the NHS Act (the "Annual Allocation").

- 13.10. NHS England will pay the Delegated Funds to the CCG using the same revenue transfer process as used for the Annual Allocation or using such other process as notified to the CCG from time to time.
- 13.11. Without prejudice to any other obligation upon the CCG, the CCG agrees that it must deal with the Delegated Funds in accordance with:
  - 13.11.1. the terms and conditions of this Agreement;
  - 13.11.2. the business rules as set out in NHS England's planning guidance or such other documents issued by NHS England from time to time;
  - 13.11.3. any Capital Investment Guidance or Primary Medical Care Infrastructure Guidance;
  - 13.11.4. any Guidance or Contractual Notice issued by NHS England from time to time in relation to the Delegated Funds (including in relation to the form or contents of any accounts in relation to the Delegated Funds); and
  - 13.11.5. the HM Treasury guidance *Managing Public Money* (dated July 2013 and found at <a href="https://www.gov.uk/government/uploads/system/uploads/attachment\_data/file/212123/Managing\_Public\_Money\_AA\_v2\_-chapters\_annex\_web.pdf">https://www.gov.uk/government/uploads/system/uploads/attachment\_data/file/212123/Managing\_Public\_Money\_AA\_v2\_-chapters\_annex\_web.pdf</a>).
- 13.12. Without prejudice to any other obligation upon the CCG, the CCG agrees that it must provide all information, assistance and support to NHS England in relation to the audit and/or investigation (whether internal or external and whether under Law or otherwise) in relation to the use of or payment of the Delegated Funds and the discharge of the Delegated Functions.

Administrative and/or Management Services and Funds in relation to the Capital Expenditure Functions

- 13.13. The Parties acknowledge that the Capital Expenditure Functions are a Reserved Function.
- 13.14. The Parties further acknowledge that:
  - 13.14.1. accordingly, the Delegated Funds do not include any funds in respect of amounts payable in relation to the Capital Expenditure Funds"); and

- 13.14.2. NHS England remains responsible and accountable for the discharge of the Capital Expenditure Functions and nothing in clauses 13.13 to 13.16 shall be construed as a divestment or delegation of NHS England's Capital Expenditure Functions.
- 13.15. Without prejudice to clause 13.14 above, the CCG will comply with any Guidance issued in relation to the Capital Expenditure Functions and shall (on request from NHS England) provide the following administrative services to NHS England in respect of the Capital Expenditure Funds:
  - 13.15.1. the administration and payment of sums that NHS England has approved as payable in relation to the Capital Expenditure Functions;
  - 13.15.2. if requested by NHS England and taking into account (i) any other support or services provided to NHS England by NHS Property Services Limited or otherwise and (ii) any Guidance issued in respect of the Capital Expenditure Functions, the provision of advice and/or recommendations to NHS England in respect of expenditure to be made under the Capital Expenditure Functions; and
  - 13.15.3. such other support or administrative assistance to NHS England that NHS England may reasonably request in order to facilitate the discharge by NHS England of its responsibilities under or in respect of the Capital Expenditure Functions.
- 13.16. NHS England may, at the same time as it transfers the Delegated Funds to the CCG under clause 13.10, transfer to the CCG such amounts as are necessary to enable the discharge of the CCG's obligations under this clause 13 (Financial Provisions and Liability) in respect of the Capital Expenditure Functions.

Administrative and/or Management Services and Funds in relation to Section 7A Functions

- 13.17. The Parties acknowledge that the Section 7A Functions are part of the Reserved Functions.
- 13.18. The Parties further acknowledge that:

- 13.18.1. accordingly, the Delegated Funds do not include any funds in respect of amounts payable in relation to the Section 7A Functions (whether such arrangements are included in or under Primary Medical Services Contracts or not) ("Section 7A Funds"); and
- 13.18.2. NHS England remains responsible and accountable for the discharge of the Section 7A Functions and nothing in this clause 13 (Financial Provisions and Liability) shall be construed as a divestment or delegation of the Section 7A Functions.
- 13.19. The CCG will provide the following services to NHS England in respect of the Section 7A Funds:
  - 13.19.1. the administration and payment of sums that NHS England has approved as payable under or in respect of arrangements for the Section 7A Functions; and
  - 13.19.2. such other support or administrative assistance to NHS England that NHS England may reasonably request in order to facilitate the discharge by NHS England of its responsibilities under or in respect of the Section 7A Funds.
- 13.20. NHS England shall, at the same time as it transfers the Delegated Funds to the CCG under clause 13.10, transfer to the CCG such amounts as are necessary to enable the discharge of the CCG's obligations under this clause 13 (Financial Provisions and Liability) in respect of the Section 7A Funds.

Administrative and/or Management Services and Funds in relation to other Reserved Functions

- 13.21. NHS England may ask the CCG to provide certain management and/or administrative services to NHS England (from a date to be notified by NHS England to the CCG) in relation to:
  - 13.21.1. the carrying out of any of the Reserved Functions; and/or
  - 13.21.2. without prejudice to the generality of clause 13.21.1, the handling and consideration of complaints.

- 13.22. If NHS England makes such a request to the CCG, then the CCG will, but only if the CCG agrees to provide such services, from the date requested by NHS England, comply with:
  - 13.22.1. provisions equivalent to those set out above in relation to the Capital Expenditure Functions (clauses 13.13 to 13.16) and the Section 7A Functions (clauses 13.17 to 13.20) including in relation to the administration of any funds for such functions but only to the extent that such provisions are relevant to the management or administrative services to be provided; and
  - 13.22.2. such other provisions in respect of the carrying out of such management and administrative services as agreed between NHS England and the CCG.
- 13.23. If NHS England asks the CCG to provide certain management and administrative services in relation to the handling and consideration of complaints and if the CCG agrees to provide such management and administrative services (with such agreement to be recorded as a variation pursuant to clause 22 (Variations)) then:
  - 13.23.1. NHS England may, in any Contractual Notice issued by NHS England in respect of such service (and as referred to in clause 13.22.2), specify procedures and responsibilities of the CCG and NHS England in relation to such complaints under the Complaints Regulations and all other Law; and
  - 13.23.2. such Contractual Notice may specify procedures in relation to the provision of an annual report to the Chief Executive of NHS England, procedures in relation to the approval of decisions in relation to complaints and/or the appointment of a responsible person by NHS England pursuant to the Complaints Regulations;
  - 13.23.3. such services shall be arrangements made under the provisions of Regulation 3 of the Complaints Regulations; and
  - 13.23.4. provided that any Contractual Notice issued pursuant to this clause shall be discussed and agreed with the CCG prior to the issue of the Contractual Notice by NHS England.

#### Pooled Funds

- 13.24. The CCG may, for the purposes of exercising the Delegated Functions under this Agreement, establish and maintain a pooled fund in respect of any part of the Delegated Funds with NHS England in accordance with section 13V of the NHS Act except that the CCG may only do so if NHS England (at its absolute discretion) consents in writing to the establishment of the pooled fund (including any terms as to the governance and payments out of such pooled fund).
- 13.25. At the date of this agreement, details of the pooled funds (including any terms as to the governance and payments out of such pooled fund) of NHS England and the CCG are set out in the Local Terms.

Business Plan, Commissioning Plan and Annual Report

- 13.26. Within two (2) months of the date of the Delegation and thereafter three (3) months before the start of each Financial Year, the CCG must prepare a plan setting out how it proposes to exercise the Delegated Functions in that Financial Year and in each of the next two (2) Financial Years (or over such longer period as NHS England may require).
- 13.27. The plan must, in particular, explain how the CCG proposes to ensure NHS England's compliance with its duties in relation to the Delegated Functions under the NHS Act, including without limitation:
  - 13.27.1. sections 223C (expenditure), 223D (controls on total resource use) and 223E (additional controls on resource use) of the NHS Act; and
  - 13.27.2. sections 13E (duty as to improvement in quality of services),13G (duty as to reducing inequalities) and 13Q (public involvement and consultation) of the NHS Act.
- 13.28. The plan must include the following:
  - 13.28.1. details of how the CCG proposes to exercise the Delegated Functions in that Financial Year and in each of the next two (2) Financial Years; and

- 13.28.2. details of how the CCG proposes to ensure NHS England's compliance with its duties to achieve any objectives and requirements relating to the Delegated Functions which are specified in the mandate published by the Department of Health to NHS England for the first Financial Year to which the plan relates; and
- 13.28.3. any other information or detail that NHS England considers necessary to ensure NHS England's compliance with its obligations under section 13T of the NHS Act or any other provision of the NHS Act or other Law.
- 13.29. The CCG must revise the plan at the request of NHS England and submit a revised plan to NHS England before the date specified by NHS England from time to time.
- 13.30. As soon as practicable after the end of each Financial Year (and in any event within two (2) months of the end of each Financial Year or such longer period as NHS England may specify), the CCG must provide to NHS England a report on how the CCG has exercised the Delegated Functions during the previous Financial Year.
- 13.31. The report referred to in clause 13.30 above must include sufficient detail to ensure NHS England's compliance with its statutory obligations under section 13U of the NHS Act.
- 13.32. Following receipt of the report referred to in clause 13.30 above, NHS England may (at its absolute discretion) require such further information from the CCG as NHS England considers necessary to ensure NHS England's compliance with its obligations under section 13U of the NHS Act.
- 13.33. The CCG shall comply with any Contractual Notices issued from time to time by NHS England in relation to the inclusion of information in relation to the Delegated Functions in any plan prepared by the CCG under section 14Z11 of the NHS Act or in any report prepared under section 14Z15 of the NHS Act.

Risk sharing

- 13.34. In accordance with section 13Z(6) of the NHS Act, NHS England retains liability in relation to the exercise of the Delegated Functions and nothing in this Agreement affects the liability of NHS England in relation to the Delegated Functions.
- 13.34A For the avoidance of doubt, NHS England retains liability in respect of any Losses arising in respect of NHS England's negligence, fraud, recklessness or deliberate breach in respect of the Delegated Functions and, if the CCG suffers any Losses in respect of such actions by NHS England, NHS England shall make such adjustments to the Delegated Funds (or other amounts payable to the CCG) in order to reflect any Losses suffered by the CCG (except to the extent that the CCG is liable for such Loss pursuant to clause 13.35).
- 13.35. The CCG is liable (and shall pay) to NHS England for any Losses suffered by NHS England that result from or arise out of the CCG's negligence, fraud, recklessness or deliberate breach of the Delegation (including any actions that are taken that exceed the authority conferred by the Delegation) or this Agreement and, in respect of such Losses, NHS England may, at its discretion and without prejudice to any other rights, either require payment from the CCG or make such adjustments to the Delegated Funds pursuant to clause 13.3. The CCG shall not be liable to the extent that the Losses arose prior to the date of this Agreement.
- 13.36. Nothing in this clause 13 (Financial Provisions and Liability) or this Agreement shall affect or prejudice NHS England's right to exercise its rights (whether arising under administrative law, common law or statute) in relation to actions or steps of the CCG, including any actions or steps that exceed the authority conferred by the Delegation or are a breach of the terms and conditions of this Agreement.

### 14. Claims and Litigation

14.1. Schedule 2 (*Delegated Functions*) sets out further detail in relation to the performance management of the Primary Medical Services Contracts.

- 14.2. Nothing in this clause 14 (Claims and Litigation) shall be interpreted as affecting the reservation to NHS England of the Reserved Functions (including the reservation to NHS England of all functions in relation to the performers list activities).
- 14.3. Except in the circumstances set out in clause 14.7 and subject always to compliance with this clause 14 (*Claims and Litigation*), the CCG shall be responsible for and shall retain the conduct of any Claim.

#### 14.4. The CCG must:

- 14.4.1. comply with any policy issued by NHS England from time to time in relation to the conduct of or avoidance of Claims and/or the pro-active management of Claims;
- 14.4.2. without prejudice to clause 14.4.1, in respect of legal advice or assistance in relation to a Claim, comply with any requirements of NHS England from time to time (whether set out in a policy issued pursuant to clause 14.4.1 or otherwise) in relation to the use of solicitors or barristers and, at the date of this Agreement, NHS England's requirement is that a CCG must obtain prior approval from NHS England in respect of the firm of solicitors instructed to provide legal advice or assistance in relation to a Claim;
- 14.4.3. if it receives any correspondence, issue of proceedings, claim document or other document concerning any Claim or potential Claim, immediately notify NHS England and send to NHS England all copies of such correspondence:
- 14.4.4. co-operate fully with NHS England in relation to such Claim and the conduct of such Claim;
- 14.4.5. provide, at its own cost, to NHS England all documentation and other correspondence that NHS England requires for the purposes of considering and/or resisting such Claim; and/or
- 14.4.6. at the request of NHS England, take such action or step or provide such assistance as may in NHS England's discretion be necessary or desirable having regard to the nature of the Claim and the existence of any time limit in relation to avoiding, disputing, defending, resisting, appealing, seeking a review or compromising such Claim or to comply with the

requirements of the NHSLA or any insurer in relation to such Claim.

- 14.5. NHS England shall use its reasonable endeavours to keep the CCG informed in respect of the conduct and/or outcome of the Claim except that NHS England shall have no obligation to do so due to any administrative or regulatory requirement, the requirement of any insurer or the NHSLA or for any other reason that NHS England may consider necessary or appropriate, at its absolute discretion, in relation to the conduct of that Claim or related matter.
- 14.6. Subject to clause 14.4 and Schedule 5 (Financial Provisions and Decision Making Limits) the CCG is entitled to conduct the Claim in the manner it considers appropriate and is also entitled to pay or settle any Claim on such terms as it thinks fit.

NHS England Stepping into Claims

14.7. NHS England may, at any time following discussion with the CCG, send a notice to the CCG stating that NHS England will take over the conduct of the Claim and the CCG must immediately take all steps necessary to transfer the conduct of such Claim to NHS England. In such cases, NHS England shall be entitled to conduct the Claim in the manner it considers appropriate and is also entitled to pay or settle any Claim on such terms as it thinks fit.

NHS England Stepping out of Claims

14.8. NHS England may, at any time after it has exercised its rights set out in clause 14.7 above and following discussion with the CCG, send a notice to the CCG stating that the CCG will be required to take over the conduct of the Claim from NHS England and NHS England must immediately take all steps necessary to transfer the conduct of such Claim to the CCG. In such cases, the CCG shall be entitled to conduct the Claim in the manner it considers appropriate in accordance with its obligations under this clause 14 (Claims and Litigation) and subject to Schedule 4 (Further Information Sharing Provisions) and Schedule 5 (Financial Provisions and Decision Making Limits).

Claim Losses

- 14.9. The CCG and NHS England shall notify each other within a reasonable time period of becoming aware of any Claim Losses.
- 14.10. If the CCG considers that, as a result of a Claim Loss, the Delegated Funds will be insufficient to meet the Claim Loss as well as discharge the Delegated Functions, then the CCG shall immediately notify NHS England and the Parties shall meet to discuss and agree any adjustment that may be needed pursuant to clause 13.3 (and taking into account any funds, provisions or other resources retained by NHS England in respect of such Claim Losses).
- 14.11. The CCG acknowledges that NHS England will pay to the CCG the funds that are attributable to the Delegated Functions. Accordingly, the CCG acknowledges that the Delegated Funds are required to be used to discharge and/or pay any Claim Losses. NHS England may, in respect of any Claim Losses, at its discretion and without prejudice to any other rights, either require payment from the CCG for such Claim Losses or pursuant to clause 13.3 make such adjustments to the Delegated Funds to take into account the amount of any Claim Losses (other than any Claim Losses in respect of which NHS England has retained any funds, provisions or other resources to discharge such Claim Losses). For the avoidance of doubt, in circumstances where NHS England suffers any Claim Losses, then NHS England shall be entitled to recoup such Claim Losses pursuant to clause 13.3. If and to the extent that NHS England has retained any funds, provisions or other resources to discharge such Claim Losses, then NHS England may either use such funds to discharge the Claim Loss or make an upward adjustment to the amounts paid to the CCG pursuant to clause 13.3.

#### 15. Breach

- 15.1. If the CCG does not comply with the Delegation or the terms of this Agreement, then NHS England may:
  - 15.1.1. exercise its rights under this Agreement; and/or
  - 15.1.2. take such steps as it considers appropriate under the CCG Assurance Framework.

- 15.2. Without prejudice to clause 15.1, if the CCG does not comply with the Delegation or the terms of this Agreement (including if the CCG exceeds its delegated authority under the Delegation), NHS England may (at its sole discretion):
  - 15.2.1. waive such non-compliance in accordance with clause 15.3 and the Delegation;
  - 15.2.2. ratify any decision in accordance with paragraph 15 of the Delegation;
  - 15.2.3. revoke the Delegation and terminate this Agreement in accordance with clause 17 (*Termination*) below;
  - 15.2.4. exercise the Escalation Rights in accordance with clause 16 (Escalation Rights); and/or
  - 15.2.5. exercise its rights under common law.
- 15.3. NHS England may waive any non-compliance by the CCG with the terms of this Agreement provided that the CCG provides a written report to NHS England pursuant to clause 15.4 and, after considering the CCG's written report, NHS England is satisfied that the waiver is justified.

#### 15.4. If:

- 15.4.1. the CCG does not comply (or the CCG considers that it may not be able to comply) with this Agreement and/or the Delegation; or
- 15.4.2. NHS England notifies the CCG that it considers the CCG has not complied, or may not be able to comply with, this Agreement and/or the Delegation,

then the CCG must provide a written report to NHS England within ten (10) days of the non-compliance (or the date on which the CCG considers that it may not be able to comply with this Agreement) or such notification pursuant to clause 15.4.2 setting out:

- 15.4.3. details of and reasons for the non-compliance (or likely non-compliance) with the Agreement and/or the Delegation; and
- 15.4.4. a plan for how the CCG proposes to remedy the non-compliance.

## 16. Escalation Rights

- 16.1. If the CCG does not comply with this Agreement and/or the Delegation, NHS England may exercise the following Escalation Rights:
  - 16.1.1. NHS England may require a suitably senior representative of the CCG to attend a review meeting within ten (10) days of NHS England becoming aware of the non-compliance; and
  - 16.1.2. NHS England may require the CCG to prepare an action plan and report within twenty (20) days of the review meeting (to include details of the non-compliance and a plan for how the CCG proposes to remedy the non-compliance).
- 16.2. Nothing in clause 16 (Escalation Rights) will affect NHS England's right to revoke the Delegation and/or terminate this Agreement in accordance with clause 17 (Termination) below.

#### 17. Termination

- 17.1. The CCG may:
  - 17.1.1. notify NHS England that it requires NHS England to revoke the Delegation; and
  - 17.1.2. terminate this Agreement

with effect from midnight on 31 March in any calendar year, provided that:

- 17.1.3. on or before 30 September of the previous calendar year, the CCG sends written notice to NHS England of its requirement that NHS England revoke the Delegation and intention to terminate this Agreement; and
- 17.1.4. the CCG meets with NHS England within ten (10) Operational Days of NHS England receiving the notice set out at clause 17.1.3 above to discuss arrangements for termination and transition of the Delegated Functions to a successor commissioner.

- in which case NHS England shall revoke the Delegation and this Agreement shall terminate with effect from midnight on 31 March in the next calendar year.
- 17.2. NHS England may revoke the Delegation at midnight on 31 March in any year, provided that it gives notice to the CCG of its intention to terminate the Delegation on or before 30 September in the year prior to the year in which the Delegation will terminate, and in which case clause 17.4 will apply.
- 17.3. The Delegation may be revoked and this Agreement may be terminated by NHS England at any time, including in (but not limited to) the following circumstances:
  - 17.3.1. the CCG acts outside of the scope of its delegated authority;
  - 17.3.2. the CCG fails to perform any material obligation of the CCG owed to NHS England under the Delegation or this Agreement;
  - 17.3.3. the CCG persistently commits non-material breaches of the Delegation or this Agreement;
  - 17.3.4. NHS England is satisfied that its intervention powers under section 14Z21 of the NHS Act apply;
  - 17.3.5. to give effect to legislative changes;
  - 17.3.6. failure to agree to a National Variation in accordance with clause 22 (*Variations*);
  - 17.3.7. NHS England and the CCG agree in writing that the Delegation shall be revoked and this Agreement shall terminate on such date as is agreed; and/or
  - 17.3.8. the CCG merges with another CCG or other body.
- 17.4. This Agreement will terminate immediately upon revocation or termination of the Delegation (including revocation and termination in accordance with this clause 17 (*Termination*)) except that the Survival Clauses will continue in full force and effect. This Agreement shall not terminate immediately if the Delegation is amended by a revocation and re-issue of an amended Delegation.

- 17.5. Upon revocation or termination of the Delegation and this Agreement (including revocation and termination in accordance with this clause 17 (*Termination*)), the Parties must:
  - 17.5.1. agree a plan for the transition of the Delegated Functions from the CCG to the successor commissioner, including details of the transition, the Parties' responsibilities in relation to the transition, the Parties' arrangements in respect of those staff engaged in the Delegated Functions and the date on which the successor commissioner will take responsibility for the Delegated Functions;
  - 17.5.2. implement and comply with their respective obligations under the plan for transition agreed in accordance with clause 17.5.1 above; and
  - 17.5.3. use all reasonable endeavours to minimise any inconvenience or disruption to the commissioning of healthcare in the Area.
- 17.6. Without prejudice to clause 15.3 and for the avoidance of doubt, NHS England may waive any right to terminate this Agreement under this clause 17 (*Termination*).

## 18. Staffing

- 18.1. The Parties acknowledge and agree that the CCG may only engage staff to undertake the Delegated Functions under one of the following three staffing models:
  - 18.1.1. "Model 1 Assignment" under the terms of which the staff of NHS England remain in their current roles and locations and provide services to the CCG under a service level agreement;
  - 18.1.2. "Model 2 Secondment" under the terms of which certain staff of NHS England are seconded to the CCG (and, for the avoidance of doubt, such secondments will terminate on revocation or termination of the Delegation); or
  - 18.1.3. "Model 3 Employment" under the terms of which the CCG may create new posts within the CCG to undertake the Delegated Functions provided that the CCG may only do so if it first offers to existing staff of NHS England an opportunity to

apply for such posts and such staff must be appointed if they are deemed appointable,

together, the "Staffing Models".

- 18.2. The CCG and NHS England, must within six (6) months of the date of this Agreement, agree which of the Staffing Models (set out at clauses 18.1.1 to 18.1.3 above) will be adopted by the CCG and the date on which such Staffing Model shall take effect.
- 18.3. In the absence of any agreement under clause 18.2, and up until such date as the CCG's preferred Staffing Model shall take effect (as referred to in clause 18.2 above), Model 1 described in clause 18.1.1 above will apply. The terms on which Model 1 will apply are set out in Schedule 8 (Assignment of NHS England Staff to the CCG).
- 18.4. The CCG must comply with any Guidance issued by NHS England from time to time in relation to the Staffing Models and such Guidance may make changes to the Staffing Models from time to time.
- 18.5. For the avoidance of doubt, any breach by the CCG of the terms of this clause 18 (Staffing), including any breach of the Guidance issued in accordance with clause 18.4 above, will be a breach of the terms and conditions of this Agreement for the purposes of clauses 13.3 and 13.35.
- 18.6. Without prejudice to clause 18.7, it is the understanding of the Parties that the provisions of the Transfer Regulations will not operate to transfer the employment of any staff of NHS England or any other party to the CCG on the commencement of the Delegation and this Agreement.
- 18.7. The Parties acknowledge that if at any time before or after the revocation or termination of the Delegation and this Agreement the Transfer Regulations do apply, the Parties must co-operate and comply with their obligations under the Transfer Regulations.

### 19. Disputes

19.1. This clause does not affect NHS England's right to take action under the CCG Assurance Framework.

- 19.2. If a dispute arises out of or in connection with this Agreement or the Delegation ("**Dispute**") then the Parties must follow the procedure set out in this clause:
  - 19.2.1. either Party must give to the other written notice of the Dispute, setting out its nature and full particulars ("Dispute Notice"), together with relevant supporting documents. On service of the Dispute Notice, the Agreement Representatives must attempt in good faith to resolve the Dispute;
  - 19.2.2. if the Agreement Representatives are, for any reason, unable to resolve the Dispute within twenty (20) days of service of the Dispute Notice, the Dispute must be referred to the Accountable Officer (or equivalent person) of the CCG and a director of or other person nominated by NHS England (and who has authority from NHS England to settle the Dispute) who must attempt in good faith to resolve it; and
  - 19.2.3. if the people referred to in clause 19.2.2 are for any reason unable to resolve the Dispute within twenty (20) days of it being referred to them, the Parties may attempt to settle it by mediation in accordance with the CEDR model mediation procedure. Unless otherwise agreed between the Parties, the mediator must be nominated by CEDR Solve. To initiate the mediation, a Party must serve notice in writing ("ADR notice") to the other Party to the Dispute, requesting a mediation. A copy of the ADR notice should be sent to CEDR Solve. The mediation will start not later than ten (10) days after the date of the ADR notice.
- 19.3. If the Dispute is not resolved within thirty (30) days after service of the ADR notice, or either Party fails to participate or to continue to participate in the mediation before the expiration of the period of thirty (30) days, or the mediation terminates before the expiration of the period of thirty (30) days, the Dispute must be referred to the Secretary of State, who shall resolve the matter and whose decision shall be binding upon the Parties.

#### 20. Freedom of Information

- 20.1. Each Party acknowledges that the other is a public authority for the purposes of the Freedom of Information Act 2000 ("FOIA") and the Environmental Information Regulations 2004 ("EIR").
- 20.2. Each Party may be statutorily required to disclose further information about the Agreement and the Relevant Information in response to a specific request under FOIA or EIR, in which case:
  - 20.2.1. each Party shall provide the other with all reasonable assistance and co-operation to enable them to comply with their obligations under FOIA or EIR;
  - 20.2.2. each Party shall consult the other regarding the possible application of exemptions in relation to the information requested; and
  - 20.2.3. subject only to clause 14 (Claims and Litigation), each Party acknowledges that the final decision as to the form or content of the response to any request is a matter for the Party to whom the request is addressed.
- 20.3. NHS England may, from time to time, issue a FOIA or EIR protocol or update a protocol previously issued relating to the dealing with and responding to of FOIA or EIR requests in relation to the Delegated Functions. The CCG shall comply with such FOIA or EIR protocols.

#### 21. Conflicts of Interest

- 21.1. The CCG must comply with its statutory duties set out in:
  - 21.1.1. Chapter A2 of the NHS Act (including those statutory duties relating to the management of conflicts of interest as set out at section 14O of the NHS Act);
  - 21.1.2. the National Health Service (Procurement, Patient Choice and Competition) (No. 2) Regulations 2013/500; and
  - 21.1.3. Regulation 24 of the Public Contracts Regulations 2015/102,

- and must perform its obligations under this Agreement in such a way as to ensure NHS England's compliance with its statutory duties in relation to conflicts of interest.
- 21.2. The CCG must have regard to all relevant guidance published by NHS England in relation to conflicts of interest in the co-commissioning context.

#### 22. Variations

- 22.1. The Parties acknowledge that, under paragraph 16 of the Delegation, the Delegation may be reviewed and amended from time to time and that such amendments may be effected by a revocation and re-issue of an amended Delegation.
- 22.2. The Parties acknowledge that, under paragraph 6 of the Delegation, certain additional functions may be delegated from time to time by NHS England to the CCG on a date or dates to be notified to the CCG by NHS England in accordance with clause 8.3. If NHS England amends the Delegation and/or delegates additional functions to the CCG, then NHS England and the CCG shall agree such consequential changes to this Agreement pursuant to this clause 22 (Variations).
- 22.3. Subject to clauses 22.4 to 22.10 below, a variation of this Agreement will only be effective if:
  - 22.3.1. it is materially in the form of the template variation agreement set out at Schedule 6 (*Template Variation Agreement*); and
  - 22.3.2. it is signed by NHS England and the CCG (by their Agreement Representatives or other duly authorised representatives).
- 22.4. The Parties may not vary any provision of this Agreement if the purported variation would contradict or conflict with the Delegation.
- 22.5. NHS England may notify the CCG of any proposed National Variation by issuing a National Variation Proposal by whatever means NHS England may consider appropriate from time to time.

- 22.6. The CCG will be deemed to have received a National Variation Proposal on the date that it is issued by NHS England.
- 22.7. The National Variation Proposal will set out the National Variation proposed and the date on which NHS England requires the National Variation to take effect.
- 22.8. The CCG must respond to a National Variation Proposal within thirty (30) Operational Days following the date that it is issued by serving a written notice on NHS England confirming either:
  - 22.8.1. that it accepts the National Variation Proposal; or
  - 22.8.2. that it refuses to accept the National Variation Proposal, and setting out reasonable grounds for that refusal.
- 22.9. If the CCG accepts the National Variation Proposal in accordance with clause 22.8.1, the CCG agrees (without delay) to take all necessary steps (including executing a variation agreement) in order to give effect to any National Variation by the date on which the proposed National Variation takes effect as set out in the National Variation Proposal.
- 22.10. If the CCG refuses to accept the National Variation Proposal in accordance with clause 22.8.2 or to take such steps as set out in clause 22.9, NHS England may terminate this Agreement and revoke the Delegation in accordance with clause 17.3.6.

### 23. Counterparts

23.1. This Agreement may be executed in counterparts, each of which shall be regarded as an original, but all of which together shall constitute one agreement binding on both of the Parties.

#### 24. Notices

- 24.1. Any notices given under this Agreement must be in writing, must be marked for the appropriate department or person and must be served by hand, post or email to the following address:
  - 24.1.1. in the case of NHS England, to NHS England's address for notices set out in the Particulars; or

24.1.2. in the case of the CCG, to the CCG's address for notices set out in the Particulars.

#### 24.2. Notices sent:

- 24.2.1. by hand will be effective upon delivery;
- 24.2.2. by post will be effective upon the earlier of actual receipt or five (5) working days after mailing; or
- 24.2.3. by email will be effective when sent (subject to no automated response being received).
- 24.3. NHS England may, at its discretion, issue Contractual Notices from time to time relating to the manner in which the Delegated Functions should be exercised by the CCG.
- 24.4. NHS England may, at its discretion, issue Guidance from time to time, including any protocol, policy, guidance or manual relating to the exercise of the Delegated Functions under this Agreement. NHS England acknowledges that in considering the need and/or content of new Guidance it will engage appropriately with CCGs.

# Schedule 1 Definitions and Interpretation

In this Agreement, the following words and phrases will bear the following meanings:

Agreement	means this agreement	between NHS England and the
-----------	----------------------	-----------------------------

CCG comprising the Particulars, the Terms and

Conditions and the Schedules;

Agreement means the CCG Representative and the NHS England

**Representatives** Representative as set out in the Particulars;

APMS Contract means an agreement made in accordance with section

92 of the NHS Act;

**Assigned Staff** means those NHS England staff as agreed between

NHS England and the CCG from time to time;

Caldicott Principles means the patient confidentiality principles set out in the

report of the Caldicott Committee (December 1997 as amended by the 2013 Report, The Information Governance Review – "To Share or Not to Share?") and now included in the NHS Confidentiality Code of

Practice, as may be amended from time to time;

Capital shall have the meaning set out in the Capital Investment

Guidance or such other replacement Guidance as

issued by NHS England from time to time;

Capital Expenditure

**Functions** 

means those functions of NHS England in relation to the use and expenditure of Capital funds (but excluding the

Premises Costs Directions Functions);

Capital Investment

Guidance

means any Guidance issued by NHS England from time to time in relation to the development, assurance and approvals process for proposals in relation to:

- the expenditure of Capital, or investment in property, infrastructure or information and technology; or
- the revenue consequences for commissioners or

## third parties making such investment;

CCG Assurance Framework

means the assurance framework that applies to CCGs

pursuant to the NHS Act;

**Claims** 

means, for or in relation to the Primary Medical Services Contracts (a) any litigation or administrative, mediation, arbitration or other proceedings, or any claims, actions or hearings before any court, tribunal or any governmental, regulatory or similar body, or any department, board or agency or (b) any dispute with, or any investigation, inquiry or enforcement proceedings by, any governmental, regulatory or similar body or agency;

Claim Losses

means all Losses arising in relation to any Claim;

**Complaints Regulations** 

means the Local Authority Social Services and National Health Service Complaints (England) Regulations 2009/309:

**Contractual Notice** 

means a contractual notice issued by NHS England to the CCG or all CCGs (as the case may be) from time to time and relating to the manner in which the Delegated Functions should be exercised by the CCG, in accordance with clause 24.3;

CQC

means the Care Quality Commission;

**Data Controller** 

shall have the same meaning as set out in the DPA;

**Data Subject** 

shall have the same meaning as set out in the DPA;

**Delegated Functions** 

means the functions delegated by NHS England to the CCG under the Delegation and as set out in detail in this Agreement;

**Delegated Funds** 

shall have the meaning in clause 13.1;

DPA

means the Data Protection Act 1998;

#### **Enhanced Services**

means the nationally defined enhanced services, as set out in the Primary Medical Services (Directed Enhanced Services) Directions 2014 or as amended from time to time, and any other enhanced services schemes locally developed by the CCG in the exercise of its Delegated Functions (and excluding, for the avoidance of doubt, any enhanced services arranged or provided pursuant to the Section 7A Functions);

**Escalation Rights** 

means the escalation rights as defined in clause 16 (Escalation Rights);

**Financial Year** 

shall bear the same meaning as in section 275 of the NHS Act:

**GMS Contract** 

means a general medical services contract made under section 84(1) of the NHS Act;

**Good Practice** 

means using standards, practices, methods and procedures conforming to the law, reflecting up-to-date published evidence and exercising that degree of skill and care, diligence, prudence and foresight which would reasonably and ordinarily be expected from a skilled, efficient and experienced commissioner;

Guidance

means any protocol, policy, guidance or manual (issued by NHS England whether under this Agreement or otherwise) and/or any policy or guidance relating to the exercise of the Delegated Functions issued by NHS England from time to time, in accordance with clause 24.4;

**HSCA** 

means the Health and Social Care Act 2012;

**Information Law** 

the DPA, the EU Data Protection Directive 95/46/EC; regulations and guidance made under section 13S and section 251 of the NHS Act; guidance made or given under sections 263 and 265 of the HSCA; the Freedom of Information Act 2000; the common law duty of confidentiality; the Human Rights Act 1998 and all other applicable laws and regulations relating to processing of

## Personal Data and privacy;

Law

means any applicable law, statute, bye-law, regulation, direction, order, regulatory policy, guidance or code, rule of court or directives or requirements of any regulatory body, delegated or subordinate legislation or notice of any regulatory body (including, for the avoidance of doubt, the Premises Costs Directions, the Statement of Financial Entitlements Directions and the Primary Medical Services (Directed Enhanced Services) Directions 2014 as amended from time to time);

Local Incenti Schemes

Incentive means an incentive scheme developed by the CCG in the exercise of its Delegated Functions including (without limitation) as an alternative to QOF;

**Local Terms** means the terms set out in Schedule 7 (Local Terms);

means all damages, loss, liabilities, claims, actions, costs, expenses (including the cost of legal and/or professional services) proceedings, demands and charges;

**National Variation** 

Losses

an addition, deletion or amendment to the provisions of this Agreement mandated by NHS England (whether in respect of the CCG or all or some of other Clinical Commissioning Groups) including any addition, deletion or amendment to reflect changes to the Delegation, changes in Law, changes in policy and notified to the CCG in accordance with clause 22 (Variations);

National Variation Proposal

**NHS Act** 

**Variation** a written proposal for a National Variation, which complies with the requirements of clause 22.7;

**Need to Know** has the meaning set out in paragraph 6.2 of Schedule 4 (Further Information Sharing Provisions);

means the National Health Service Act 2006 (as amended by the Health and Social Care Act 2012 or

other legislation from time to time);

40

**NHS England** means the National Health Service Commissioning

Board established by section 1H of the NHS Act, also

known as NHS England;

**Non-Personal Data** means data which is not Personal Data:

**Operational Days** a day other than a Saturday, Sunday or bank holiday in

England;

**Particulars** means the Particulars of this Agreement as set out in

clause 1 (Particulars);

**Party/Parties** means a party or both parties to this Agreement;

**Personal Data** shall have the same meaning as set out in the DPA and

shall include references to Sensitive Personal Data

where appropriate;

**Personal** 

Agreement

**Data** means the agreement governing Information Law issues

completed further to Schedule 4 (Further Information

Sharing Provisions);

**Personnel** means the Parties' employees, officers,

> members, directors, voluntary staff, consultants, and other contractors and sub-contractors acting on behalf of either Party (whether or not the arrangements with such contractors and sub-contractors are subject to legally binding contracts) and such contractors' and their

sub-contractors' personnel;

**PMS Contract** means an arrangement or contract for the provision of

> primary medical services made under section 83(2) of the NHS Act (including any arrangements which are made in reliance on a combination of that section and other powers to arrange for primary medical services);

**Premises Agreements** means tenancies, leases and other arrangements in

relation to the occupation of land for the delivery of

services under the Primary Medical Services Contracts;

**Premises** Costs means the National Health Service (General Medical

#### **Directions**

Services Premises Costs) Directions 2013, as amended;

# Premises Costs Directions Functions

means NHS England's functions in relation to the Premises Costs Directions:

# Primary Medical Care Infrastructure Guidance

means any Guidance issued by NHS England from time to time in relation to the procurement, development and management of primary medical care infrastructure and which may include principles of best practice;

# Primary Medical Services Contracts

#### means:

- PMS Contracts:
- GMS Contracts; and
- APMS Contracts,

in each case as amended or replaced from time to time and including all ancillary or related agreements directly relating to the subject matter of such agreements, contracts or arrangements but excluding any Premises Agreements;

### **GP Access Fund**

Means the former Prime Minister's challenge fund, announced in October 2013 to help improve access to general practice and stimulate innovative ways of providing primary care services;

# Principles of Best Practice

means the Guidance in relation to property and investment which is to be published either before or after the date of this Agreement;

### QOF

means the quality and outcomes framework;

## **Relevant Information**

means the Personal Data and Non-Personal Data processed under the Delegation and this Agreement, and includes, where appropriate, "confidential patient information" (as defined under section 251 of the NHS Act), and "patient confidential information" as defined in the 2013 Report, The Information Governance Review – "To Share or Not to Share?");

#### **Reserved Functions** means the functions re

means the functions relating to the commissioning of primary medical services which are reserved to NHS England (and are therefore not delegated to the CCG under the Delegation) and as set out in detail in clause 8.2 and Schedule 3 (Reserved Functions) of this Agreement;

**Secretary of State** 

means the Secretary of State for Health from time to time;

**Section 7A Functions** 

means those functions of NHS England exercised pursuant to section 7A of the NHS Act relating to primary medical services;

**Section 7A Funds** 

shall have the meaning in clause 13.18.1;

**Sensitive Personal Data** 

shall have the same meaning as in the DPA;

**Specified Purpose** 

means the purpose for which the Relevant Information is shared and processed, being to facilitate the exercise of the CCG's Delegated Functions and NHS England's Reserved Functions as specified in paragraph 2.1 of Schedule 4 (Further Information Sharing Provisions) to this Agreement;

Statement of Financial Entitlements Directions

means the General Medical Services Statement of Financial Entitlements Directions 2013, as amended or updated from time to time;

**Statutory Guidance** 

means any applicable health and social care guidance, guidelines, direction or determination, framework, standard or requirement to which the CCG and/or NHS England have a duty to have regard, to the extent that the same are published and publicly available or the existence or contents of them have been notified to the CCG by NHS England from time to time;

**Survival Clauses** 

means clauses 10 (Information Sharing and Information Governance), 13 (Financial Provisions and Liability), 14 (Claims and Litigation) 17 (Termination), 18 (Staffing), 19 (Disputes) and 20 (Freedom of Information), together with such other provisions as are required to interpret these clauses (including the Schedules to this Agreement); and

**Transfer Regulations** 

means the Transfer of Undertakings (Protection of Employment) Regulations 2006, as amended.

# Schedule 2 Delegated Functions

### Part 1: Delegated Functions: Specific Obligations

#### 1. Introduction

1.1. This Part 1 of Schedule 2 (*Delegated Functions*) sets out further provision regarding the carrying out of each of the Delegated Functions.

# 2. Primary Medical Services Contract Management

#### 2.1. The CCG must:

- 2.1.1. manage the Primary Medical Services Contracts on behalf of NHS England and perform all of NHS England's obligations under each of the Primary Medical Services Contracts in accordance with the terms of the Primary Medical Services Contracts as if it were named in the contract in place of NHS England;
- 2.1.2. actively manage the performance of the counter-party to the Primary Medical Services Contracts in order to secure the needs of people who use the services, improve the quality of services and improve efficiency in the provision of the services including by taking timely action to enforce contractual breaches and serve notice;
- 2.1.3. ensure that it obtains value for money under the Primary Medical Services Contracts on behalf of NHS England and avoids making any double payments under any Primary Medical Services Contracts;
- 2.1.4. comply with all current and future relevant national Guidance regarding PMS reviews and the management of practices receiving Minimum Practice Income Guarantee (MPIG) (including without limitation the *Framework for Personal Medical Services (PMS) Contracts Review* guidance published by NHS England in September 2014 (<a href="https://www.england.nhs.uk/2014/02/pms/">https://www.england.nhs.uk/2014/02/pms/</a> And <a href="https://www.england.nhs.uk/commissioning/wp-">https://www.england.nhs.uk/commissioning/wp-</a>

content/uploads/sites/12/2016/05/implement-pms-fund-changes.pdf ));

- 2.1.5. notify NHS England immediately (or in any event within two (2) Operational Days) of any breach by the CCG of its obligations to perform any of NHS England's obligations under the Primary Medical Services Contracts;
- 2.1.6. keep a record of all of the Primary Medical Services Contracts that the CCG manages on behalf of NHS England setting out the following details in relation to each Primary Medical Services Contract:
  - 2.1.6.1. name of counter-party;
  - 2.1.6.2. location of provision of services; and
  - 2.1.6.3. amounts payable under the contract (if a contract sum is payable) or amount payable in respect of each patient (if there is no contract sum).
- 2.2. For the avoidance of doubt, all Primary Medical Services Contracts will be in the name of NHS England.
- 2.3. The CCG must comply with any Guidance in relation to the issuing and signing of Primary Medical Services Contracts in the name of NHS England.
- 2.4. Without prejudice to clause 13 (Financial Provisions and Liability) or paragraph 2.1 above, the CCG must actively manage each of the relevant Primary Medical Services Contracts including by:
  - 2.4.1. managing the relevant Primary Medical Services Contract, including in respect of quality standards, incentives and the QOF, observance of service specifications, and monitoring of activity and finance;
  - 2.4.2. assessing quality and outcomes (including clinical effectiveness, patient experience and patient safety);
  - 2.4.3. managing variations to the relevant Primary Medical Services Contract or services in accordance with national policy, service user needs and clinical developments;
  - 2.4.4. agreeing information and reporting requirements and managing information breaches (which will include use of the NHS Digital IG Toolkit SIRI system);

- 2.4.5. agreeing local prices, managing agreements or proposals for local variations and local modifications:
- 2.4.6. conducting review meetings and undertaking contract management including the issuing of contract queries and agreeing any remedial action plan or related contract management processes; and
- 2.4.7. complying with and implementing any relevant Guidance issued from time to time.

#### **Enhanced Services**

- 2.5. The CCG must manage the design and commissioning of Enhanced Services, including re-commissioning these services annually where appropriate.
- 2.6. The CCG must ensure that it complies with any Guidance in relation to the design and commissioning of Enhanced Services.
- 2.7. When commissioning newly designed Enhanced Services, the CCG must:
  - 2.7.1. consider the needs of the local population in the Area;
  - 2.7.2. support Data Controllers in providing 'fair processing' information as required by the DPA;
  - 2.7.3. develop the necessary specifications and templates for the Enhanced Services, as required to meet the needs of the local population in the Area;
  - 2.7.4. when developing the necessary specifications and templates for the Enhanced Services, ensure that value for money will be obtained;
  - 2.7.5. consult with Local Medical Committees, each relevant Health and Wellbeing Board and other stakeholders in accordance with the duty of public involvement and consultation under section 14Z2 of the NHS Act;
  - 2.7.6. obtain the appropriate read codes, to be maintained by NHS Digital;
  - 2.7.7. liaise with system providers and representative bodies to ensure that the system in relation to the Enhanced Services will be functional and secure; and

2.7.8. support GPs in entering into data processing agreements with data processors in the terms required by the DPA.

## **Design of Local Incentive Schemes**

- 2.8. The CCG may design and offer Local Incentive Schemes for GP practices, sensitive to the needs of their particular communities, in addition to or as an alternative to the national framework (including as an alternative to QOF or directed Enhanced Services), provided that such schemes are voluntary and the CCG continues to offer the national schemes.
- 2.9. There is no formal approvals process that the CCG must follow to develop a Local Incentive Scheme, although any proposed new Local Incentive Scheme:
  - 2.9.1. is subject to consultation with the Local Medical Committee;
  - 2.9.2. must be able to demonstrate improved outcomes, reduced inequalities and value for money; and
  - 2.9.3. must reflect the changes agreed as part of the national PMS reviews.
- 2.10. The ongoing assurance of any new Local Incentive Schemes will form part of the CCG's assurance process under the CCG Assurance Framework.
- 2.11. Any new Local Incentive Scheme must be implemented without prejudice to the right of GP practices operating under a GMS Contract to obtain their entitlements which are negotiated and set nationally.
- 2.12. NHS England will continue to set national standing rules, to be reviewed annually, and the CCG must comply with these rules which shall for the purposes of this Agreement be Guidance.

## **Making Decisions on Discretionary Payments**

2.13. The CCG must manage and make decisions in relation to the discretionary payments to be made to GP practices in a consistent, open and transparent way.

2.14. The CCG must exercise its discretion to determine the level of payment to GP practices of discretionary payments, in accordance with the Statement of Financial Entitlements Directions.

# Making Decisions about Commissioning Urgent Care for Out of Area Registered Patients

- 2.15. The CCG must manage the design and commissioning of urgent care services (including home visits as required) for its patients registered out of area (including re-commissioning these services annually where appropriate).
- 2.16. The CCG must ensure that it complies with any Guidance in relation to the design and commissioning of these services.

## 3. Planning the Provider Landscape

- 3.1. The CCG must plan the primary medical services provider landscape in the Area, including considering and taking decisions in relation to:
  - 3.1.1. establishing new GP practices in the Area;
  - 3.1.2. managing GP practices providing inadequate standards of patient care;
  - 3.1.3. the procurement of new Primary Medical Services Contracts (in accordance with any procurement protocol issued by NHS England from time to time);
  - 3.1.4. closure of practices and branch surgeries;
  - 3.1.5. dispersing the lists of GP practices;
  - 3.1.6. agreeing variations to the boundaries of GP practices; and
  - 3.1.7. coordinating and carrying out the process of list cleansing in relation to GP practices, according to any policy or Guidance issued by NHS England from time to time.
- 3.2. In relation to any new Primary Medical Services Contract to be entered into, the CCG must, without prejudice to any obligation in Schedule 2, Part 2, paragraph 3 (*Procurement and New Contracts*) and Schedule 2, Part 1, paragraph 2.3:
  - 3.2.1. consider and use the form of Primary Medical Services Contract that will ensure compliance with NHS England's

- obligations under Law including the Public Contracts Regulations 2015/102 and the National Health Service (Procurement, Patient Choice and Competition) (No. 2) Regulations 2013/500 taking into account the persons to whom such Primary Medical Services Contracts may be awarded;
- 3.2.2. provide to NHS England confirmation as required from time to time that it has considered and complied with its obligations under this Agreement and the Law; and
- 3.2.3. for the avoidance of doubt, Schedule 5 (Financial Provisions and Decision Making Limits) deals with the sign off requirements for Primary Medical Services Contracts.

## 4. Approving GP Practice Mergers and Closures

- 4.1. The CCG is responsible for approving GP practice mergers and GP practice closures in the Area.
- 4.2. The CCG must undertake all necessary consultation when taking any decision in relation to GP practice mergers or GP practice closures in the Area, including those set out under section 14Z2 of the NHS Act (duty for public involvement and consultation). The consultation undertaken must be appropriate and proportionate in the circumstances and should include consulting with the Local Medical Committee.
- 4.3. Prior to making any decision in accordance with this paragraph 4 (Approving GP Practice Mergers and Closures), the CCG must be able to clearly demonstrate the grounds for such a decision and must have fully considered any impact on the GP practice's registered population and that of surrounding practices. The CCG must be able to clearly demonstrate that it has considered other options and has entered into dialogue with the GP contractor as to how any closure or merger will be managed.
- 4.4. In making any decisions pursuant to paragraph 4 (Approving GP Practice Mergers and Closures), the CCG shall also take account of its obligations as set out in Schedule 2, part 2, paragraph 3 (Procurement and New Contracts), where applicable.

# 5. Information Sharing with NHS England in relation to the Delegated Functions

- 5.1. This paragraph 5 (Information Sharing with NHS England) is without prejudice to clause 9.4 or any other provision in this Agreement. The CCG must provide NHS England with:
  - 5.1.1. such information relating to individual GP practices in the Area as NHS England may reasonably request, to ensure that NHS England is able to continue to gather national data regarding the performances of GP practices;
  - 5.1.2. such data/data sets as required by NHS England to ensure population of the primary medical services dashboard;
  - 5.1.3. any other data/data sets as required by NHS England; and
  - 5.1.4. the CCG shall procure that providers accurately record and report information so as to allow NHS England and other agencies to discharge their functions.
- 5.2. The CCG must use the NHS England approved primary medical services dashboard, as updated from time to time, for the collection and dissemination of information relating to GP practices.
- 5.3. The CCG must (where appropriate) use the NHS England approved GP exception reporting service (as notified to the CCGs by NHS England from time to time).
- 5.4. The CCG must provide any other information, and in any such form, as NHS England considers necessary and relevant.
- 5.5. NHS England reserves the right to set national standing rules (which may be considered Guidance for the purpose of this Agreement), as needed, to be reviewed annually. NHS England will work with CCGs to agree rules for, without limitation, areas such as the collection of data for national data sets and IT intra-operability. Such national standing rules set from time to time shall be deemed to be part of this Agreement.

# 6. Making Decisions in relation to Management of Poorly Performing GP Practices

- 6.1. The CCG must make decisions in relation to the management of poorly performing GP practices and including, without limitation, decisions and liaison with the CQC where the CQC has reported non-compliance with standards (but excluding any decisions in relation to the performers list).
- 6.2. In accordance with paragraph 6.1 above, the CCG must:
  - 6.2.1. ensure regular and effective collaboration with the CQC to ensure that information on general practice is shared and discussed in an appropriate and timely manner;
  - 6.2.2. ensure that any risks identified are managed and escalated where necessary;
  - 6.2.3. respond to CQC assessments of GP practices where improvement is required;
  - 6.2.4. where a GP practice is placed into special measures, lead a quality summit to ensure the development and monitoring of an appropriate improvement plan (including a communications plan and actions to manage primary care resilience in the locality); and
  - 6.2.5. take appropriate contractual action in response to CQC findings.

#### 7. Premises Costs Directions Functions

- 7.1. The CCG must comply with the Premises Costs Directions and will be responsible for making decisions in relation to the Premises Costs Directions Functions.
- 7.2. In particular, but without limiting the generality of paragraph 7.1, the CCG shall make decisions concerning:
  - 7.2.1. applications for new payments under the Premises Costs
    Directions (whether such payments are to be made by way of
    grants or in respect of recurring premises costs); and
  - 7.2.2. revisions to existing payments being made under the Premises Costs Directions.

- 7.3. The CCG must comply with any decision-making limits set out in Schedule 5 (*Financial Provisions and Decision Making Limits*) when taking decisions in relation to the Premises Costs Directions Functions.
- 7.4. The CCG will comply with any guidance issued by the Secretary of State or NHS England in relation to the Premises Costs Directions, including the Principles of Best Practice, and any other Guidance in relation to the Premises Costs Directions.
- 7.5. The CCG must work cooperatively with other CCGs to manage premises and strategic estates planning.
- 7.6. The CCG must liaise where appropriate with NHS Property Services Limited and Community Health Partnerships Limited in relation to the Premises Costs Directions Functions.

#### Schedule 2

#### Part 2 – Delegated Functions: General Obligations

#### 1. Introduction

1.1. This Part 2 of Schedule 2 (*Delegated Functions*) sets out general provisions regarding the carrying out of the Delegated Functions.

### 2. Planning and reviews

2.1. The CCG is responsible for planning the commissioning of primary medical services.

#### 2.2. The role of the CCG includes:

- 2.2.1. carrying out primary medical health needs assessments (to be developed by the CCG) to help determine the needs of the local population in the Area;
- 2.2.2. recommending and implementing changes to meet any unmet primary medical services needs; and
- 2.2.3. undertaking regular reviews of the primary medical health needs of the local population in the Area.

#### 3. Procurement and New Contracts

- 3.1. The CCG will make procurement decisions relevant to the exercise of the Delegated Functions and in accordance with the detailed arrangements regarding procurement set out in the procurement protocol issued and updated by NHS England from time to time.
- 3.2. In discharging its responsibilities set out in clause 6 (Performance of the Delegated Functions) of this Agreement and paragraph 1 of this Schedule 2 (Delegated Functions), the CCG must comply at all times with Law including its obligations set out in the National Health Service (Procurement, Patient Choice and Competition) (No. 2) Regulations 2013/500 and any other relevant statutory provisions. The CCG must have regard to any relevant guidance, particularly Monitor's guidance Substantive guidance on the Procurement, Patient Choice and Competition Regulations

(<a href="https://www.gov.uk/government/uploads/system/uploads/attachment\_dat">https://www.gov.uk/government/uploads/system/uploads/attachment\_dat</a> a/file/283505/SubstantiveGuidanceDec2013\_0.pdf).

- 3.3. Where the CCG wishes to develop and offer a locally designed contract, it must ensure that it has consulted with its Local Medical Committee in relation to the proposal and that it can demonstrate that the scheme will:
  - 3.3.1. improve outcomes;
  - 3.3.2. reduce inequalities; and
  - 3.3.3. provide value for money.

### 4. Integrated working

- 4.1. The CCG must take an integrated approach to working and co-ordinating with stakeholders including NHS England, Local Professional Networks, local authorities, Healthwatch, acute and community providers, the Local Medical Committee, Public Health England and other stakeholders.
- 4.2. The CCG must work with NHS England and other CCGs to co-ordinate a common approach to the commissioning of primary medical services generally.
- 4.3. The CCG and NHS England will work together to coordinate the exercise of their respective performance management functions.

## 5. Resourcing

5.1. NHS England may, at its discretion provide support or staff to the CCG. NHS England may, when exercising such discretion, take into account, any relevant factors (including without limitation the size of the CCG, the number of Primary Medical Services Contracts held and the need for the Local NHS England Team to continue to deliver the Reserved Functions).

# Schedule 3 Reserved Functions

#### 1. Introduction

- 1.1. This Schedule 3 (Reserved Functions) sets out further provision regarding the carrying out of the Reserved Functions.
- 1.2. The CCG will work collaboratively with NHS England and will support and assist NHS England to carry out the Reserved Functions.

## 2. Management of the national performers list

- NHS England will continue to perform its primary medical care functions under the National Health Service (Performers Lists) (England) Regulations 2013.
- 2.2. NHS England's functions in relation to the management of the national performers list include:
  - 2.2.1. considering applications and decision-making in relation to inclusion on the national performers list, inclusion with conditions and refusals;
  - 2.2.2. identifying, managing and supporting primary care performers where concerns arise; and
  - 2.2.3. managing suspension, imposition of conditions and removal from the national performers list.
- 2.3. NHS England may hold local Performance Advisory Group ("PAG") meetings to consider all complaints or concerns that are reported to NHS England in relation to a named performer and NHS England will determine whether an initial investigation is to be carried out.
- 2.4. NHS England may notify the CCG of all relevant PAG meetings at least seven (7) days in advance of such meetings. NHS England may require a representative of the CCG to attend such meetings to discuss any performer concerns and/or quality issues that may impact on individual performer cases.

2.5. The CCG must develop a mechanism to ensure that all complaints regarding any named performer are escalated to the Local NHS England Team for review. The CCG will comply with any Guidance issued by NHS England in relation to the escalation of complaints about a named performer.

### 3. Management of the revalidation and appraisal process

- 3.1. NHS England will continue to perform its functions under the Medical Profession (Responsible Officers) Regulations 2010 (as amended by the Medical Profession (Responsible Officers) (Amendment) Regulations 2013).
- 3.2. All functions in relation to GP appraisal and revalidation will remain the responsibility of NHS England, including:
  - 3.2.1. the funding of GP appraisers;
  - 3.2.2. quality assurance of the GP appraisal process; and
  - 3.2.3. the responsible officer network.
- 3.3. Funding to support the GP appraisal is incorporated within the global sum payment to GP practices.
- 3.4. The CCG must not remove or restrict the payments made to GP practices in respect of GP appraisal.

# 4. Administration of payments and related performers list management activities

- 4.1. NHS England reserves its functions in relation to the administration of payments to individual performers and related performers list management activities under the National Health Service (Performers Lists) (England) Regulations 2013 and other relevant legislation.
- 4.2. NHS England may continue to pay GPs who are suspended from the national performers list under the Secretary of State's Determination: Payments to Medical Practitioners Suspended from the Medical Performers List (1 April 2013).

4.3. For the avoidance of doubt, the CCG is responsible for any ad hoc or discretionary payments to GP practices (including those under section 96 of the NHS Act) in accordance with clause 6.2.1.4 and Schedule 2 (Delegated Functions) Part 1 paragraphs 2.13 and 2.14 of this Agreement, including where such payments may be considered a consequence of actions taken under the National Health Service (Performers Lists) (England) Regulations 2013.

#### 5. Section 7A Functions

- 5.1. In accordance with clauses 13.17 to 13.20, NHS England retains the Section 7A Functions and will be responsible for taking decisions in relation to the Section 7A Functions.
- 5.2. In accordance with clauses 13.17 to 13.20, the CCG will provide certain management and/or administrative services to NHS England in relation to the Section 7A Functions.

### 6. Capital Expenditure Functions

6.1. In accordance with clauses 13.13 to 13.16, NHS England retains the Capital Expenditure Functions and will be responsible for taking decisions in relation to the Capital Expenditure Functions.

### 7. Functions in relation to complaints management

- 7.1. NHS England retains its functions in relation to complaints management and will be responsible for taking decisions in relation to the management of complaints. Such complaints include (but are not limited to):
  - 7.1.1. complaints about GP practices and individual named performers;
  - 7.1.2. controlled drugs; and
  - 7.1.3. whistleblowing in relation to a GP practice or individual performer.
- 7.2. The CCG must immediately notify the Local NHS England Team of all complaints received by or notified to the CCG and must send to the Local NHS England Team copies of any relevant correspondence.

- 7.3. The CCG must co-operate fully with NHS England in relation to any complaint and any response to such complaint.
- 7.4. In accordance with clauses 13.21 to 13.23, NHS England may ask the CCG to provide certain management and/or administrative services to NHS England (from a date to be notified by NHS England to the CCG) in relation to the handling and consideration of complaints.

# 8. Such other ancillary activities that are necessary in order to exercise the Reserved Functions

- 8.1. NHS England will carry out such other ancillary activities that are necessary in order for NHS England to exercise the Reserved Functions.
- 8.2. NHS England will continue to comply with its obligations under the Controlled Drugs (Supervision of Management and Use) Regulations 2013.
- 8.3. The CCG must assist NHS England's controlled drug accountable officer ("CDAO") to carry out its functions under the Controlled Drugs (Supervision of Management and Use) Regulations 2013.
- 8.4. The CCG must nominate a relevant senior individual within the CCG (the "CCG CD Lead") to liaise with and assist NHS England to carry out its functions under the Controlled Drugs (Supervision of Management and Use) Regulations 2013.
- 8.5. The CCG CD Lead must, in relation to the Delegated Functions:
  - 8.5.1. on request provide NHS England's CDAO with all reasonable assistance in any investigation involving primary medical care services;
  - 8.5.2. report all complaints involving controlled drugs to NHS England's CDAO;
  - 8.5.3. report all incidents or other concerns involving the safe use and management of controlled drugs to NHS England's CDAO;
  - 8.5.4. analyse the controlled drug prescribing data available; and

8.5.5. on request supply (or ensure organisations from whom the CCG commissions services involving the regular use of controlled drugs supply) periodic self-declaration and/or self-assessments to NHS England's CDAO.

# Schedule 4 Further Information Sharing Provisions

#### 1. Introduction

- 1.1. The purpose of this Schedule 4 (Further Information Sharing Provisions) and the associated Personal Data Agreement is to set out the scope for the secure and confidential sharing of information between the Parties on a **Need To Know** basis between individual Personnel in order to enable the Parties to exercise their primary medical care commissioning functions in accordance with the law. This Schedule and the associated Personal Data Agreement is designed to:
  - 1.1.1. inform about the reasons why Relevant Information may need to be shared and how this will be managed and controlled by the organisations involved;
  - 1.1.2. describe the purposes for which the Parties have agreed to share Relevant Information;
  - 1.1.3. set out the lawful basis for the sharing of information between the Parties, and the principles that underpin the exchange of Relevant Information;
  - 1.1.4. describe roles and structures to support the exchange of Relevant Information between the Parties;
  - apply to the sharing of Relevant Information relating to GPs where necessary;
  - 1.1.6. apply to the sharing of Relevant Information whatever the medium in which it is held and however it is transmitted;
  - 1.1.7. ensure that Data Subjects are, where appropriate, informed of the reasons why Personal Data about them may need to be shared and how this sharing will be managed;
  - 1.1.8. apply to the activities of the Parties' Personnel; and
  - 1.1.9. describe how complaints relating to Personal Data sharing between the Parties will be investigated and resolved, and how the information sharing will be monitored and reviewed.

## 2. Purpose

- 2.1. The Specified Purpose(s) of the data sharing initiative is to facilitate the exercise of the CCG's Delegated Functions and NHS England's Reserved Functions:
  - 2.1.1. the management of the primary medical service performers' list in accordance with section 91 of the NHS Act;
  - 2.1.2. management of GP revalidation and appraisal;
  - 2.1.3. administration of payments and related performers list management activities;
  - 2.1.4. planning and delivering the provision of appropriate care services:
  - 2.1.5. improving the health of the local population;
  - 2.1.6. performance management of GP providers;
  - 2.1.7. investigating and responding to incidents and complaints; and
  - 2.1.8. reducing risk to individuals, service providers and the public as a whole.
- 2.2. Specific and detailed purposes are set out in the Personal Data Agreement appended to this Schedule.

### 3. Benefits of information sharing

3.1. The benefits of sharing information are the achievement of the Specified Purposes set out above, with benefits for service users and other stakeholders in terms of the improved local delivery of primary healthcare services.

### 4. Legal basis for Sharing

- 4.1. Each Party shall comply with all relevant Information Law requirements and good practice in relation to the processing of Relevant Information shared further to this Agreement.
- 4.2. The Parties shall identify the lawful basis for sharing Relevant Information for each purpose and data flow, and document these in the attached Personal Data Agreement.

#### 5. Relevant Information to be shared

5.1. The Relevant Information to be shared is set out in the attached Personal Data Agreement.

#### 6. Restrictions on use of the Shared Information

- 6.1. Each Party shall only process the Relevant Information as is necessary to achieve the Specified Purpose, and, in particular, shall not use or process Relevant Information for any other purpose unless agreed in writing by the Data Controller that released the information to the other. There shall be no other use or onward transmission of the Relevant Information to any third party without a lawful basis first being determined, and the originating Data Controller being notified.
- 6.2. Access to, and processing of, the Relevant Information provided by a Party must be the minimum necessary to achieve the Specified Purpose. Information and Sensitive Personal Data will be handled at all times on a restricted basis, in compliance with Information Law requirements, and Personnel should only have access to Personal Data on a justifiable **Need to Know** basis for the purpose of performing their duties in connection with the services they are there to deliver. The **Need to Know** requirement means that the Data Controllers' Personnel will only have access to Personal Data or Sensitive Personal Data if it is lawful for such Personnel to have access to such data for the Specified Purpose and the function they are required to fulfil at that particular time, in relation to the Specified Purpose, cannot be achieved without access to the Personal Data or Sensitive Personal Data specified.
- 6.3. Having this Agreement in place does not give licence for unrestricted access to data that the other Data Controller may hold. It lays the parameters for the safe and secure sharing and processing of information for a justifiable **Need to Know** purpose.
- 6.4. Neither Party shall subcontract any processing of the Relevant Information without the prior written consent of the other Party. Where a Party subcontracts its obligations, it shall do so only by way of a written agreement with the sub-contractor which imposes the same

- obligations as are imposed on the Data Controllers under this Agreement.
- 6.5. Neither Party shall cause or allow Data to be transferred to any territory outside the European Economic Area without the prior written permission of the responsible Data Controller.
- 6.6. Any particular restrictions on use of certain Relevant Information are included in the attached Personal Data Agreement.

## 7. Ensuring fairness to the Data Subject

- 7.1. In addition to having a lawful basis for sharing information, the DPA generally requires that the sharing must be fair. In order to achieve fairness to the Data Subjects, the Parties will put in place the following arrangements:
  - 7.1.1. amendment of internal guidance to improve awareness and understanding among Personnel;
  - 7.1.2. amendment of privacy notices and policies; and
  - 7.1.3. consideration given to further activities to promote public understanding where appropriate.
- 7.2. Each Party shall procure that its notification to the Information Commissioner's Office reflects the flows of information under this Agreement.
- 7.3. Further provision in relation to specific data flows is included in the attached Personal Data Agreement.

#### 8. Governance: Personnel

8.1. Each Party must take reasonable steps to ensure the suitability, reliability, training and competence, of any Personnel who have access to the Personal Data (and Sensitive Personal Data) including reasonable background checks and evidence of completeness should be available on request by each Party.

- 8.2. The Parties agree to treat all Relevant Information as confidential and imparted in confidence and must safeguard it accordingly. Where the Personnel are not healthcare professionals (for the purposes of the DPA) the employing Parties must procure that its Personnel operate under a duty of confidentiality which is equivalent to that which would arise if that person were a health professional.
- 8.3. Each Party shall ensure that all Personnel required to access the Personal Data (including Sensitive Personal Data) are informed of the confidential nature of the Personal Data and each Party shall include appropriate confidentiality clauses in employment/service contracts of all Personnel that have any access whatsoever to the Relevant Information, including details of sanctions against any employee acting in a deliberate or reckless manner that may breach the confidentiality or the non-disclosure provisions of Information Law requirements, or causes damage to or loss of the Relevant Information.
- 8.4. Each Party shall provide evidence (further to any reasonable request) that all Personnel that have any access to the Relevant Information whatsoever are adequately and appropriately trained to comply with their responsibilities under Information Law and this Agreement.
- 8.5. Each Party shall ensure that:
  - 8.5.1. only those employees involved in delivery of the Agreement use or have access to the Relevant Information; and
  - 8.5.2. that such access is granted on a strict **Need to Know** basis and shall implement appropriate access controls to ensure this requirement is satisfied and audited. Evidence of audit should be made freely available on request by the originating Data Controller. These access controls are set out in the attached Personal Data Agreement; and
  - 8.5.3. specific limitations on the Personnel who may have access to the Information are set out in the attached Personal Data Agreement.
- 9. Governance: Protection of Personal Data

- 9.1. At all times, the Parties shall have regard to the requirements of Information Law and the rights of Data Subjects.
- 9.2. Wherever possible (in descending order of preference), only anonymised information, or strongly or weakly pseudonymised information will be shared and processed by Parties, without the need to share easily identifiable Personal Data. The Parties shall cooperate in exploring alternative strategies to avoid the use of Personal Data in order to achieve the Specified Purpose. However, it is accepted that some Relevant Information shared further to this Agreement may be Personal Data/Sensitive Personal Data.
- 9.3. Processing of any Personal Data or Sensitive Personal Data shall be to the minimum extent necessary to achieve the Specified Purpose, and on a **Need to Know** basis. If either Party:
  - 9.3.1. becomes aware of any unauthorised or unlawful processing of any Relevant Information or that any Relevant Information is lost or destroyed or has become damaged, corrupted or unusable; or
  - 9.3.2. becomes aware of any security breach,

in respect of the Relevant Information it shall promptly notify the other Party. The Parties shall fully cooperate with one another to remedy the issue as soon as reasonably practicable.

- 9.4. In processing any Relevant Information further to this Agreement, each Party shall:
  - 9.4.1. process the Personal Data (including Sensitive Personal Data) only in accordance with the terms of this Agreement and otherwise only in accordance with written instructions from the originating Data Controller in respect of its Relevant Information:
  - 9.4.2. process the Personal Data (including Sensitive Personal Data) only to the extent as is necessary for the provision of the Specified Purpose or as is required by law or any regulatory body;
  - 9.4.3. process the Personal Data (including Sensitive Personal Data) only in accordance with Information Law requirements and shall not perform its obligations under this Agreement in

- such a way as to cause any other Data Controller to breach any of their applicable obligations under Information Law; and
- 9.4.4. process the Personal Data in accordance with the eight data protection principles (the "Data Protection Principles") in Schedule 1 to the DPA.
- 9.5. Each Party shall act generally in accordance with the Seventh Data Protection Principle, and in particular shall implement and maintain appropriate technical and organisational measures to protect the Personal Data (and Sensitive Personal Data) against unauthorised or unlawful processing and against accidental loss, destruction, damage, alteration or disclosure. These measures shall be appropriate to the harm which might result from any unauthorised or unlawful processing, accidental loss, destruction or damage to the Personal Data (and Sensitive Personal Data) and having regard to the nature of the Personal Data (and Sensitive Personal Data) which is to be protected. In particular, each Data Controller shall:
  - 9.5.1. ensure that only Personnel authorised under this Agreement have access to the Personal Data (and Sensitive Personal Data);
  - 9.5.2. ensure that the Relevant Information is kept secure and in an encrypted form, and shall use all reasonable security practices and systems applicable to the use of the Relevant Information to prevent and to take prompt and proper remedial action against, unauthorised access, copying, modification, storage, reproduction, display or distribution, of the Relevant Information;
  - 9.5.3. obtain prior written consent from the originating Data Controller in order to transfer the Relevant Information to any third party;
  - 9.5.4. permit the other Data Controllers or the Data Controllers' representatives (subject to reasonable and appropriate confidentiality undertakings), to inspect and audit the data processing activities carried out further to this Agreement (and/or those of its agents, successors or assigns) and comply with all reasonable requests or directions to enable the Data Controllers to verify and/or procure that the other

- Data Controller is in full compliance with its obligations under this Agreement; and
- 9.5.5. if requested, provide a written description of the technical and organisational methods and security measures employed in processing Personal Data.
- 9.5.6. Specific requirements as to information security are set out in the Schedule.
- 9.5.7. Each Party shall use best endeavours to achieve and adhere to the requirements of the NHS Information Governance Toolkit, particularly in relation to Confidentiality and Data Protection Assurance, Information Security Assurance and Clinical Information Assurance.
- 9.5.8. The Parties' Single Points of Contact ("SPoC") set out in paragraph 14 (Governance: Single Points of Contact) below will be the persons who, in the first instance, will have oversight of third party security measures.

#### 10. Governance: Transmission of Information between the Parties

- 10.1. This paragraph supplements paragraph 9 (Governance: Protection of Personal Data) of this Schedule.
- 10.2. Transfer of Personal Data between the Parties shall be done through secure mechanisms including use of the N3 network, encryption, and approved secure (NHS.net / gcsx) email.
- 10.3. Faxes shall only be used to transmit Personal Data in an emergency.
- 10.4. Wherever possible, Personal Data should be transmitted in pseudonymised form, with only reference to the NHS number in 'clear' transmissions. Where there are significant consequences for the care of the patient, then additional data items, such as the postcode, date of birth and/or other identifiers should also be transmitted, in accordance with good information governance and clinical safety practice, so as to ensure that the correct patient record / data is identified.
- 10.5. Any other special measures relating to security of transfer are specified in the attached Personal Data Agreement.

- 10.6. Each Party shall keep an audit log of Relevant Information transmitted and received in the course of this Agreement.
- 10.7. The Parties' Single Point of Contact notified pursuant to paragraph 14 (Governance: Single Points of Contact) will be the persons who, in the first instance, will have oversight of the transmission of information between the Parties.

## 11. Governance: Quality of Information

- 11.1. The Parties will take steps to ensure the quality of the Relevant Information and to comply with the fourth Data Protection Principle.
- 11.2. Special measures relating to ensuring quality are set out in the attached Personal Data Agreement.

### 12. Governance: Retention and Disposal of Shared Information

- 12.1. The non-originating Party shall securely destroy or return the Relevant Information once the need to use it has passed or, if later, upon the termination of this Agreement, howsoever determined. Where Relevant Information is held electronically the Relevant Information will be deleted and formal notice of the deletion sent to the Party that shared the Relevant Information. Once paper information is no longer required, paper records will be securely destroyed or securely returned to the Party they came from.
- 12.2. Each Party shall provide an explanation of the processes used to securely destroy or return the information, or verify such destruction or return, if requested by the other Party and shall comply with any request of the Data Controllers to dispose of data in accordance with specified standards or criteria.
- 12.3. If either Party is required by any law, regulation, or government or regulatory body to retain any documents or materials that it would otherwise be required to return or destroy under this paragraph 12 (Governance: Retention and Disposal of Shared Information), it shall

- notify the other Party in writing of that retention, giving details of the documents or materials that it must retain.
- 12.4. Retention of any data shall comply with the Fifth Data Protection Principle and with all good practice including the Records Management NHS Code of Practice, as updated or amended from time to time.
- 12.5. Any special retention periods are set out in attached Personal Data Agreement.
- 12.6. Each Party shall ensure that Relevant Information held in paper form is held in secure files, and, when it is no-longer needed, destroyed using a cross cut shredder or subcontracted to a confidential waste company that complies with European Standard EN15713.
- 12.7. Each Party shall ensure that, when no longer required, electronic storage media used to hold or process Personal Data are destroyed or overwritten to current policy requirements.
- 12.8. Electronic records will be considered for deletion once the relevant retention period has ended.
- 12.9. In the event of any bad or unusable sectors of electronic storage media that cannot be overwritten, the Party shall ensure complete and irretrievable destruction of the media itself in accordance with policy requirements.

#### 13. Governance: Complaints and Access to Personal Data

- 13.1. Each Party shall assist the other in responding to any request made under Information Law made by persons who wish to access copies of information held about them ("Subject Access Requests").
- 13.2. Complaints about information sharing shall be routed through each Party's own complaints procedure but reported to the Single Points of Contact set out in paragraph 14 (Governance: Single Points of Contact) below.

- 13.3. The Parties shall use all reasonable endeavours to work together to resolve any dispute or complaint arising under this Agreement or any data processing carried out further to it.
- 13.4. Basic details of the Agreement shall be included in the appropriate log under each Party's Publication Scheme.

## 14. Governance: Single Points of Contact

14.1. The Parties each shall appoint a single point of contact to whom all queries relating to the particular information sharing should be directed in the first instance. Details of the single points of contact shall be set out in the attached Personal Data Agreement.

### 15. Monitoring and review

15.1. The Parties shall monitor and review on an ongoing basis the sharing of Relevant Information to ensure compliance with Information Law and best practice. Specific monitoring requirements are set out in the attached Personal Data Agreement.

## **Template Personal Data Agreement**

Data flow: [Description]

# Description of information flow and single points of contact for parties involved

Originating Data Controller	[Insert:]			
Contact details for single point of contact for Originating Data Controller	Name of point of contact	Title	Contact (email)	Contact (phone)
Recipient Data Controller	[Insert:]			
Contact details for single point of contact of Recipient Data Controller	Name of point of contact	Title	Contact (email)	Contact (phone)

## Description of information to be shared

Comprehensive	[Insert:]
description of	
Relevant	
Information to be	
shared	
Anonymised / not	Yes / No
information about	
individual	
persons	
Strongly	Yes / No
pseudonyimsed	
Weakly	Yes / No
pseudonymised	
Person -	Yes / No
identifiable data	
Justification for	[Insert or N/A:]

## Legal basis for disclosure and use

DPA Schedule 2	[Insert or N/A:]		
condition/s			
DPA Schedule 3	[Insert or N/A:]		
condition/s			
Confidentiality	Explicit consent	Yes / No	
		[If yes, how documented?:]	
	Implied Consent	Yes / No	
		[If yes, how have you implied	
		consent?:]	
	Statutory required/permitted	[Incort statutory basis:]	
	Statutory required/permitted disclosure	[Insert statutory basis:]	
	uisciosure		
	Public interest disclosure	[Insert how the public interest	
		favours use/disclosure of the	
		information:]	
		-	
	Other legal basis	[Insert:]	
s. 13Z3 / 14Z23	S. 13Z3 condition(s) to	[Insert:]	
NHS Act 2006	permit disclosure		
justification	S. 14Z23 condition(s) to	[Insert:]	
	permit disclosure		
Other specific			
legal			

considerations	
Restrictions on use	e of information
[Insert:]	

## Governance arrangements

Specific measures to	[Insert:]
ensure fairness to the	
Data Subject, including	
privacy impact	
assessments undertaken	
Access controls on use	[Insert:]
of information	
Specific limitations on	[Insert:]
Personnel who may	
access information	
Other specific security	[Insert:]
requirements	
(transmission)	
Other specific security	[Insert:]
requirements (general)	
Specific requirements as	[Insert:]
to ensuring quality of	
information	
Specific requirements for	[Insert:]
retention and destruction	
of information	
Specific monitoring and	[Insert:]
review arrangements	

# Schedule 5 Financial Provisions and Decision Making Limits

## Financial Limits and Approvals

- 1. The CCG shall ensure that any decisions in respect of the Delegated Functions and which exceed the financial limits set out below are only taken:
  - 1.1. by the following persons and/or individuals set out in column 2 of Table1 below; and
  - 1.2. following the approval of NHS England (if any) as set out in column 3 of the Table 1 below.
- 2. NHS England may, from time to time, update Table 1 by sending a notice to the CCG of amendments to Table 1.

Table 1 – Financial Limits			
Decision	Person/Individual	NHS England Approval	
General			
Taking any step or action in relation to the settlement of a Claim, where the value of the settlement exceeds £100,000	CCG Accountable Officer or Chief Finance Officer or Chair	NHS England Head of Legal Services and Local NHS England Team Director or Director of Finance	
Any matter in relation to the Delegated Functions which is novel, contentious or repercussive	CCG Accountable Officer or Chief Finance Officer or Chair	Local NHS England Team Director or Director of Finance or NHS England Region Director or Director of Finance or NHS England Chief Executive or Chief Financial Officer	
Revenue Contracts			
The entering into of any Primary Medical Services Contract which has or is capable of having a term which exceeds five (5) years	CCG Accountable Officer or Chief Finance Officer or Chair	Local NHS England Team Director or Director of Finance	

## Capital

Note: As at the date of this Agreement, the CCG will not have delegated or directed responsibility for decisions in relation to Capital expenditure (and these decisions are retained by NHS England) but the CCG may be required to carry out certain administrative services in relation to Capital expenditure under clause 13 (Financial Provisions and Liability).

## Schedule 6

## **Template Variation Agreement**

Variation Reference:		[insert reference]
Prop	osed by:	[insert party] [Note – only NHS England may propose National Variations]
Date of Proposal:		[insert date]
Date	of Variation Agreeme	nt: [insert date]
•	alised words and phrasem in the Agreement ref	es in this Variation Agreement have the meanings given terred to above.
1.	The Parties have agree	ed the [National] Variation summarised below:
2.	<u> </u>	on is reflected in the attached Schedule and the Parties ent is varied accordingly.
3.	The Variation takes effect on [insert date].	
	ITNESS OF WHICH to ate(s) shown below	ne Parties have signed this Variation Agreement on
Signo	ed by	NHS England [Insert name of Authorised Signatory] [for and on behalf of] [ ]
Signe	ed by	[Insert name] Clinical Commissioning Group [Insert name of Authorised Signatory][for and on behalf of] [ ]

## Schedule to Variation Agreement

[Insert details of variation]

## Schedule 7 Local Terms

"There are no Local Terms" in this Schedule 7

# Schedule 8 Assignment of NHS England Staff to the CCG

#### 1. Introduction

- 1.1. The purpose of this Schedule 8 (Assignment of NHS England Staff to the CCG) is to give clarity to the CCG and NHS England, in circumstances where NHS England staff are assigned to the CCG under Model 1 of the Staffing Models.
- 1.2. In accordance with clause 18 of this Agreement, the Parties have agreed that the CCG may only engage staff to undertake the Delegated Functions under one of the three Staffing Models referred to in that clause.
- 1.3. The Parties agree and acknowledge that until such time as the CCG's preferred Staffing Model takes effect, the engagement of staff to undertake the Delegated Functions shall be in accordance with the terms of this Schedule 8 (Assignment of NHS England Staff to the CCG) (the "Arrangements").

#### 2. Duration

2.1. The Arrangements shall commence on the date of this Agreement and shall continue until the date on which the Parties agree which of the Staffing Models (set out at clauses 18.1.1 to 18.1.3) will be adopted by the CCG and the date on which such Staffing Model shall take effect.

#### 3. Services

- 3.1. NHS England agrees to make available the Assigned Staff to the CCG to perform administrative and management support services together with such other services specified in Schedule 7 (Local Terms) (the "Services") so as to facilitate the CCG in undertaking the Delegated Functions pursuant to the terms of this Agreement.
- 3.2. NHS England shall take all reasonable steps to ensure that the Assigned Staff shall:

- 3.2.1. faithfully and diligently perform duties and exercise such powers as may from time to time be reasonably assigned to or vested in them; and
- 3.2.2. perform all duties assigned to them pursuant to this Schedule 8 (Assignment of NHS England Staff to the CCG).
- 3.3. The CCG shall notify NHS England if the CCG becomes aware of any act or omission by any Assigned Staff which may have a material adverse impact on the provision of the Services or constitute a material breach of the terms and conditions of employment of the Assigned Staff.
- 3.4. NHS England shall be released from its obligations to make the Assigned Staff available for the purposes of this Schedule 8 (Assignment of NHS England Staff to the CCG) whilst the Assigned Staff are absent:
  - 3.4.1. by reason of industrial action taken in contemplation of a trade dispute;
  - 3.4.2. as a result of the suspension or exclusion of employment or secondment of any Assigned Staff by NHS England;
  - 3.4.3. in accordance with the Assigned Staff's respective terms and conditions of employment and policies, including, but not limited to, by reason of training, holidays, sickness, injury, trade union duties, paternity leave or maternity or where absence is permitted by Law;
  - 3.4.4. if making the Assigned Staff available would breach or contravene any Law;
  - 3.4.5. as a result of the cessation of employment of any individual Assigned Staff; and/or
  - 3.4.6. at such other times as may be agreed between NHS England and the CCG.

## 4. Employment of the Assigned Staff

4.1. NHS England shall employ the Assigned Staff and shall be responsible for the employment of the Assigned Staff at all times on whatever terms and conditions as NHS England and the Assigned Staff may agree from time to time.

- 4.2. NHS England shall pay the Assigned Staff their salaries and benefits and make any deductions for income tax liability and national insurance or similar contributions it is required to make from the Assigned Staff's salaries and other payments.
- 4.3. The Assigned Staff shall carry out the Services from NHS England's places of work and may be required to attend the offices of the CCG from time to time in the course of carrying out the Services. Nothing in this Schedule 8 (Assignment of NHS England Staff to the CCG) shall be construed or have effect as constituting any relationship of employer and employee between the CCG and the Assigned Staff.
- 4.4. NHS England shall not, and shall procure that the Assigned Staff shall not, hold themselves out as employees of the CCG.

## 5. Management

- 5.1. NHS England shall have day-to-day control of the activities of the Assigned Staff and deal with any management issues concerning the Assigned Staff including, without limitation, performance appraisal, discipline and leave requests.
- 5.2. The CCG agrees to provide all such assistance and co-operation that NHS England may reasonably request from time to time to resolve grievances raised by Assigned Staff and to deal with any disciplinary allegations made against Assigned Staff arising out of or in connection with the provision of the Services which shall include, without limitation, supplying NHS England with all information and the provision of access to all documentation and personnel as NHS England requires for the purposes of considering and dealing with such issues and participating promptly in any action which may be necessary.

#### 6. Conduct of Claims

6.1. If the CCG becomes aware of any matter that may give rise to a claim by or against a member of Assigned Staff, notice of that fact shall be given as soon as possible to NHS England. NHS England and the CCG shall co-operate in relation to the investigation and resolution of any such claims or potential claims. 6.2. No admission of liability shall be made by or on behalf of the CCG and any such claim shall not be compromised, disposed of or settled without the consent of NHS England.

### 7. Confidential Information and Property

- 7.1. For the avoidance of doubt, this paragraph 8 (Confidential Information and Property) is without prejudice to any other provision of this Agreement in relation to confidential information.
- 7.2. It is acknowledged that to enable the Assigned Staff to provide the Services, the Parties may share information of a highly confidential nature being information or material which is the property of NHS England or the CCG or which NHS England or the CCG are obliged to hold confidential including, without limitation, all official secrets, information relating to the working of any project carried on or used by the relevant Party, research projects, strategy documents, tenders, financial information, reports, ideas and know-how, employee confidential information and patient confidential information and any proprietary party information (any and all of the foregoing being "Confidential Information").
- 7.3. The Parties agree to adopt all such procedures as the other party may reasonably require and to keep confidential all Confidential Information and that the Parties shall not (save as required by law) disclose the Confidential Information in whole or in part to anyone and agree not to disclose the Confidential Information other than in connection with the provision of the Services.
- 7.4. The obligations under this Agreement apply to all and any Confidential Information whether the Confidential Information was in or comes into the possession of the relevant person prior to or following this Agreement and such obligations shall continue at all times following the termination of the Arrangements but shall cease to apply to information which may come into the public domain otherwise than through unauthorised disclosure by NHS England or the CCG, as the case may be.

## 8. Intellectual Property

8.1. All Intellectual Property (meaning any invention, idea, improvement, discovery, development, innovation, patent, writing, concept design made, process information discovered, copyright work, trademark, trade name and/or domain name) made, written, designed, discovered or originated by the Assigned Staff shall be the property of NHS England to the fullest extent permitted by law and NHS England shall be the absolute beneficial owner of the copyright in any such Intellectual Property.



April 2019				
Agenda Item: 19.07	Author of the Paper:			
Report date: 4 <sup>th</sup> April 2019	Angela Price Primary Care Programme Lead angela.price@southseftonccg.nhs.uk Tel: 01513178379			
Title: South Sefton Delegation Agreement				
Summary/Key Issues:				
Recommendation The Primary Care Commissioning Committee is ask Delegation Agreement	Note x  ked to note the attached Approve Ratify			

Links to Corporate Objectives (x those that apply)				
х	To focus on the identification of QIPP (Quality, Improvement, Productivity & Prevention) schemes and the implementation and delivery of these to achieve the CCG QIPP target and to support delivery of financial recovery.			
х	To progress Shaping Sefton as the strategic plan for the CCG, in line with the NHSE planning requirements set out in the "Five Year Forward View", underpinned by transformation through the agreed strategic blueprints and programmes and as part of the NHS Cheshire and Merseyside Healthcare Partnership.			
х	To ensure that the CCG maintains and manages performance & quality across the mandated constitutional measures.			
х	To support Primary Care Development through the development of an enhanced model of care and supporting estates strategy, underpinned by a complementary primary care quality contract.			
х	To advance integration of in-hospital and community services in support of the CCG locality model of care.			
	To advance the integration of Health and Social Care through collaborative working with Sefton Metropolitan Borough Council, supported by the Health and Wellbeing Board.			

Process	Yes	No	N/A	Comments/Detail (x those that apply)
Patient and Public			Х	
Engagement				
Clinical Engagement			Х	
Equality Impact			Х	
Assessment				
Legal Advice Sought			Х	
Resource Implications			Х	
Considered				
Locality Engagement			Х	
Presented to other			Х	
Committees				

Links to National Outcomes Framework (x those that apply)				
Х	Preventing people from dying prematurely			
Х	Enhancing quality of life for people with long-term conditions			
Х	Helping people to recover from episodes of ill health or following injury			
Х	Ensuring that people have a positive experience of care			
Х	Treating and caring for people in a safe environment and protecting them from avoidable harm			

# **Delegation Agreement**

#### 1. Particulars

1.1. This Agreement records the particulars of the agreement made between NHS England and the Clinical Commissioning Group named below.

Area South Sefton

Clinical Commissioning Group NHS South Sefton Clinical

**Commissioning Group** 

CCG Representative Fiona Taylor, Chief Officer

CCG Address for Notices 3<sup>rd</sup> Floor

**Merton House** 

Bootle L20 3DL

Date of Agreement [Leave Blank]

Delegation means the delegation made by NHS

England to the CCG of certain functions relating to primary medical services under section 13Z of the NHS Act and effective from 1 April 2015 (as amended pursuant to the Delegation)

NHS England Representative [Insert details of name of manager of

this Agreement for NHS England]

Local NHS England Team NHS England North (Cheshire and

Merseyside)

NHS England Address for Regatta Place

Notices

Brunswick Business Park

Summers Boad

Summers Road LIVERPOOL L3 4BL

- 1.2. This Agreement comprises:
  - 1.2.1. the Particulars (Clause 1);
  - 1.2.2. the Terms and Conditions (Clauses 2 to 24 and Schedule 1 to Schedule 6 and Schedule 8 to this Agreement); and

Mona Taylor.

1.2.3. the Local Terms (Schedule 7).

Signed by

**NHS England** 

(for and on behalf of NHS England)

Signed by

**NHS South Sefton Clinical Commissioning Group** 

**Fiona Taylor** 

Chief Officer (for and on behalf of NHS South Sefton

**Wider Group (the Membership)** 

## **Terms and Conditions**

## A. Introduction

## 2. Interpretation

- 2.1. This Agreement is to be interpreted in accordance with Schedule 1 (Definitions and Interpretation).
- 2.2. If there is any conflict or inconsistency between the provisions of this Agreement and the provisions of the Delegation, the provisions of the Delegation will prevail.
- 2.3. If there is any conflict or inconsistency between the provisions of this Agreement, that conflict or inconsistency must be resolved according to the following order of priority:
  - 2.3.1. the Particulars and Terms and Conditions (Clauses 1 to 24 and, in particular, clause 8.7);
  - 2.3.2. Schedule 1 to Schedule 6 and Schedule 8 to this Agreement; and
  - 2.3.3. Schedule 7 (Local Terms).
- 2.4. This Agreement and any ancillary agreements it refers to constitute the entire agreement and understanding between the Parties relating to the Delegation and supersedes all previous agreements, promises and understandings between them, whether written or oral, relating to its subject matter.

## 3. Background

- 3.1. NHS England has delegated the Delegated Functions to the CCG under section 13Z of the NHS Act and as set out in the Delegation.
- 3.2. Arrangements made under section 13Z of the NHS Act may be made on such terms and conditions (including terms as to payment) as may be agreed between NHS England and the CCG.

- 3.3. This Agreement sets out the arrangements that apply in relation to the exercise of the Delegated Functions by the CCG.
- 3.4. For the avoidance of doubt, functions relating to the commissioning of primary care pharmacy, dental and optical contracts are not delegated to the CCG under the Delegation. The Delegation relates only to the delegation and reservation of primary medical services commissioning functions as set out in this Agreement.

#### 4. Term

4.1. This Agreement has effect from the date set out in paragraph 5 of the Delegation and will remain in force unless terminated in accordance with clause 17 (*Termination*) below.

## 5. Principles

- 5.1. In performing their obligations under this Agreement, NHS England and the CCG must:
  - 5.1.1. at all times act in good faith towards each other;
  - 5.1.2. at all times exercise functions effectively, efficiently and economically;
  - 5.1.3. act in a timely manner;
  - 5.1.4. share information and best practice, and work collaboratively to identify solutions, eliminate duplication of effort, mitigate risk and reduce cost;
  - 5.1.5. at all times observe relevant statutory powers, requirements and best practice to ensure compliance with applicable laws and standards including those governing procurement, and Information Law; and
  - 5.1.6. have regard to the needs and views of the other Party and as far as is lawful and reasonably practicable, take such needs and views into account.

## B. Role of the CCG

6. Performance of the Delegated Functions

- 6.1. The role of the CCG will be to exercise the Delegated Functions in the Area.
- 6.2. The Delegated Functions are the functions set out in Schedule 1of the Delegation and being:
  - 6.2.1. decisions in relation to the commissioning, procurement and management of Primary Medical Services Contracts, including but not limited to the following activities:
    - 6.2.1.1. decisions in relation to Enhanced Services;
    - 6.2.1.2. decisions in relation to Local Incentive Schemes (including the design of such schemes);
    - 6.2.1.3. decisions in relation to the establishment of new GP practices (including branch surgeries) and closure of GP practices;
    - 6.2.1.4. decisions about 'discretionary' payments;
    - 6.2.1.5. decisions about commissioning urgent care (including home visits as required) for out of area registered patients;
  - 6.2.2. the approval of practice mergers;
  - 6.2.3. planning primary medical care services in the Area, including carrying out needs assessments;
  - 6.2.4. undertaking reviews of primary medical care services in the Area;
  - 6.2.5. decisions in relation to the management of poorly performing GP practices and including, without limitation, decisions and liaison with the CQC where the CQC has reported non-compliance with standards (but excluding any decisions in relation to the performers list);
  - 6.2.6. management of the Delegated Funds in the Area;
  - 6.2.7. Premises Costs Directions Functions;
  - 6.2.8. co-ordinating a common approach to the commissioning of primary care services with other commissioners in the Area where appropriate; and
  - 6.2.9. such other ancillary activities that are necessary in order to exercise the Delegated Functions.

- 6.3. Schedule 2 (*Delegated Functions*) sets out further detail in relation to the Delegated Functions and the exercise of such Delegated Functions.
- 6.4. The CCG agrees that it must perform the Delegated Functions in accordance with:
  - 6.4.1. the Delegation;
  - 6.4.2. the terms of this Agreement;
  - 6.4.3. all applicable Law;
  - 6.4.4. the CCG's constitution;
  - 6.4.5. Statutory Guidance; and
  - 6.4.6. Good Practice.
- 6.4A The CCG must have due regard to Guidance and Contractual Notices.
- 6.5. Without prejudice to clause 6.4, the CCG agrees that it must perform the Delegated Functions in such a manner as to ensure NHS England's compliance with NHS England's statutory duties in respect of the Delegated Functions and to enable NHS England to fulfil its Reserved Functions.
- 6.6. When performing the Delegated Functions, the CCG will not do anything, take any step or make any decision outside of its delegated authority as set out in the Delegation.
- 6.7. Without prejudice to any other provision in this Agreement, the CCG must comply with the NHS England central finance team's operational process (as such process is updated from time to time) for the reporting and accounting of the Delegated Funds. In particular, the CCG will be required to permit the NHS England central finance team and/or their agents and contractors authorised by them to have the ability to access the CCG ledger to provide the services required to deliver financial support and assistance to the CCG necessary to enable them to manage the Delegated Funds and exercise the Delegated Functions. NHS England and the CCG will agree any accruals to be made including any adjustments related to the relevant Financial Year expenditure to ensure no net financial impact or gain on the CCG.
- 6.8. The decisions of the CCG in exercising the Delegated Functions will be binding on the CCG and NHS England.

### 7. Committee

- 7.1. The CCG must establish a committee to exercise its Delegated Functions.
- 7.2. The structure and operation of the committee must be constituted so as to take into account Guidance issued by NHS England including the revised statutory guidance on managing conflicts of interest for CCGs https://www.england.nhs.uk/commissioning/pc-co-comms/coi/

# C. Functions reserved to NHS England

### 8. Performance of the Reserved Functions

- 8.1. The role of NHS England will be to exercise the Reserved Functions.
- 8.2. Subject to clause 8.3, the Reserved Functions are all of NHS England's functions relating to primary medical services other than the Delegated Functions and including those functions set out in Schedule 2of the Delegation and being:
  - 8.2.1. management of the national performers list;
  - 8.2.2. management of the revalidation and appraisal process;
  - 8.2.3. administration of payments in circumstances where a performer is suspended and related performers list management activities;
  - 8.2.4. Capital Expenditure Functions;
  - 8.2.5. Section 7A Functions;
  - 8.2.6. functions in relation to complaints management;
  - 8.2.7. decisions in relation to the GP Access Fund; and
  - 8.2.8. such other ancillary activities that are necessary in order to exercise the Reserved Functions.
- 8.3. For the avoidance of doubt, the Parties acknowledge that the Delegation may be amended and additional functions may be delegated to the CCG, in which event consequential changes to this Agreement

- shall be agreed with the CCG pursuant to clause 22 (Variations) of this Agreement.
- 8.4. Schedule 3 (*Reserved Functions*) sets out further detail in relation to the Reserved Functions.
- 8.5. To support and assist NHS England in carrying out the Reserved Functions, the CCG will share information with NHS England in accordance with section E (*Information*) below.
- 8.6. NHS England will work collaboratively with the CCG when exercising the Reserved Functions, including discussing with the CCG how it proposes to address GP performance issues.
- 8.7. If there is any conflict or inconsistency between functions that are named as Delegated Functions and functions that are named as Reserved Functions then such functions shall be interpreted as Reserved Functions.
- 8.8. The Parties acknowledge that, as at the date of this Agreement, the CCG shall provide administrative and management services to NHS England in relation to certain Reserved Functions and that such administrative and management services are as follows:
  - 8.8.1. the administrative and management services in relation to the Capital Expenditure Functions and the Capital Expenditure Funds as more particularly set out in clauses 13.13 to 13.16; and
  - 8.8.2. the administrative and management services in relation to the Section 7A Functions and Section 7A Funds as more particularly set out in clauses 13.17 to 13.20.
- 8.9. The Parties further acknowledge that NHS England may ask the CCG to provide certain administrative and management services to NHS England in relation to other Reserved Functions as more particularly set out in clauses 13.21 to 13.23. Such administrative and management services shall only be provided by the CCG following agreement by the CCG.

8.10. Notwithstanding any arrangement for or provision of administrative or management services in respect of certain Reserved Functions, NHS England shall retain and be accountable for the exercise of such Reserved Functions.

# D. Commissioning

## 9. Monitoring and Reporting - General Requirements

- 9.1. The CCG must comply with any reporting requirements under:
  - 9.1.1. this Agreement (including, without limitation, as required by clause 9 (Monitoring and Reporting General Requirements), clause 12 (Public Information and Access Targets), clause 13 (Financial Provisions and Liability), clause 14 (Claims and Litigation) and Schedule 2 Part 1 paragraph 2 (Primary Medical Services Contract Management) and paragraph 5 (Information Sharing with NHS England));
  - 9.1.2. the CCG Assurance Framework; and
  - 9.1.3. the CCG's constitution.
- 9.2. NHS England shall monitor the exercise and carrying out of the Delegated Functions by the CCG under the terms of this Agreement and as part of the CCG Assurance Framework.
- 9.3. The CCG will notify NHS England of all primary medical services commissioning committee meetings at least seven (7) days in advance of such meetings and NHS England will be entitled to attend such meetings at its discretion.
- 9.4. The CCG must provide to NHS England:
  - 9.4.1. all information in relation to the exercise of the Delegated Functions (including in relation to the Delegation or this Agreement), (and in such form) as requested by NHS England from time to time; and
  - 9.4.2. all such information (and in such form), that may be relevant to NHS England in relation to the exercise by NHS England of its other duties or functions including, without limitation, the Reserved Functions.

9.5. Nothing in this Agreement shall affect NHS England's power to require information from the CCG under sections 14Z17, 14Z18, 14Z19 and 14Z20 of the NHS Act.

## E. Information

## 10. Information Sharing and Information Governance

- 10.1. Schedule 4 (Further Information Sharing Provisions) makes further provision about information sharing and information governance.
- 10.2. NHS England and the CCG will enter into a Personal Data Agreement that will describe the processing of Relevant Information that identifies individuals under this Agreement. A template Personal Data Agreement is set out in Schedule 4 (Further Information Sharing Provisions).
- 10.3. The Personal Data Agreement:
  - 10.3.1. sets out the relevant Information Law and best practice, including the requirements of the NHS Digital IG Toolkit;
  - 10.3.2. sets out how that law and best practice will be implemented, including responsibilities of the Parties to co-operate properly and fully with each other;
  - 10.3.3. identifies the Relevant Information that may be processed, including what may be shared, under this Agreement;
  - 10.3.4. identifies the purposes for which the Relevant Information may be so processed and states the legal basis for the processing in each case;
  - 10.3.5. states who is/are the data controller/s and, if appropriate, the data processor/s of Personal Data;
  - 10.3.6. sets out what will happen to the Personal Data on the termination of this Agreement (with due regard to clause 17 (*Termination*) of the Agreement); and
  - 10.3.7. sets out such other provisions as are necessary for the sharing of Relevant Information to be fair, lawful and meet best practice.
- 10.4. NHS England and the CCG will share all Non-Personal Data in accordance with Information Law and their statutory powers as set out

- in section 13Z3 (for NHS England) and section 14Z23 (for the CCG) of the NHS Act.
- 10.5. The Parties agree that, in relation to information sharing and the processing of Relevant Information under the Delegation and this Agreement, they must comply with:
  - 10.5.1. all relevant Information Law requirements including the common law duty of confidence (unless disapplied by statute) and other legal obligations in relation to information sharing including those set out in the NHS Act and the Human Rights Act 1998:
  - 10.5.2. Good Practice; and
  - 10.5.3. relevant guidance (including guidance given by the Information Commissioner, the Caldicott Principles, the requirements of the NHS Information Governance Toolkit to level 2, and guidance issued further to sections 263 and 265 of the HSCA) and consistent with guidance issued under section 13S of the NHS Act to providers.

## 11. IT inter-operability

- 11.1. NHS England and the CCG will work together to ensure that all relevant IT systems operated by NHS England and the CCG in respect of the Delegated Functions and the Reserved Functions are inter-operable and that data may be transferred between systems securely, easily and efficiently.
- 11.2. The Parties will use their respective reasonable endeavours to help develop initiatives to further this aim.

## 12. Public Information and Access Targets

12.1. The CCG must promptly make available to NHS England such information as is required in respect of the Delegated Functions to ensure NHS England's discharge of its statutory duties.

- 12.2. The CCG must ensure that all new Primary Medical Services Contracts contain appropriate provisions such that the CCG is able to discharge its obligations in clause 12.1.
- 12.3. The CCG must ensure that any information provided under this Agreement complies with all relevant national data sets issued by NHS England and NHS Digital.

## F. General

## 13. Financial Provisions and Liability

Notification of the Delegated Funds and Adjustments to the Delegated Funds

- 13.1. NHS England will, in respect of each Financial Year, notify the CCG of the proportion of the funds allocated to NHS England by the Secretary of State pursuant to Chapter 6 of the NHS Act and which are to be paid to the CCG for the purpose of meeting expenditure in respect of the Delegated Functions for that Financial Year (the "Delegated Funds").
- 13.2. Except in relation to pooled funds and subject to the terms of this clause 13 (Financial Provisions and Liability) and, in particular, clause 13.4, the CCG must use the Delegated Funds to meet expenditure in respect of the exercise of the Delegated Functions. Without prejudice to the generality of the foregoing, the CCG must make:
  - 13.2.1. all payments in relation to the Primary Medical Services Contracts including payments in relation to QOF and implementing financial adjustments or sanctions (including in relation to breaches of provider obligations); and
  - 13.2.2. all payments under the Premises Costs Directions.
- 13.3. NHS England may, in any Financial Year by sending a notice to the CCG of such increase or decrease, increase or reduce the Delegated Funds:
  - 13.3.1. in order to take into account any monthly adjustments or corrections to the Delegated Funds that NHS England considers appropriate (following discussions with the CCG), including without limitation adjustments following any changes to the Delegation or Delegated Functions (including changes

- pursuant to paragraph 6 or paragraph 16 of the Delegation), changes in allocations, changes in contracts or otherwise;
- 13.3.2. in order to comply with a change in the amount allocated to NHS England by the Secretary of State pursuant to section 223B of the NHS Act:
- 13.3.3. to take into account any Losses arising under clause 13.35;
- 13.3.4. to take into account any Claim Losses;
- 13.3.5. to take into account any adjustments that NHS England considers appropriate (including without limitation in order to make corrections or otherwise to reflect notional budgets) to reflect funds transferred (or that should have been transferred) to the CCG in respect of the Delegated Funds and/or funds transferred (or that should have been transferred) to the CCG and in respect of which the CCG has management or administrative responsibility under clauses 13.13 to 13.23 of this Agreement; or
- 13.3.6. in order to ensure compliance by NHS England of its obligations under the NHS Act (including without limitation, Chapter 6 of the NHS Act) or the HSCA or any action taken or direction made by the Secretary of State under the NHS Act or the HSCA.
- 13.3A NHS England acknowledges that the intention of clause 13.3 is to reflect genuine corrections and adjustments to the Delegated Funds and may not be used to change the allocation of the Delegated Funds unless there are significant or exceptional circumstances that would require such corrections or adjustments (including but not limited to a change in the mandate published by the Department of Health or other external factors).
- 13.4. The CCG acknowledges that it must comply with its statutory financial duties, including those under sections 223H and 223I of the NHS Act to the extent that these sections apply in relation to the receipt of the Delegated Funds.
- 13.5. The CCG acknowledges its duty under section 14S of the NHS Act to assist and support NHS England in discharging its duty under section 13E so far as relating to securing continuous improvement in the quality of primary medical services and agrees that it shall take this duty into

- account in relation to the exercise of the Delegated Functions and the use of the Delegated Funds.
- 13.6. The CCG must ensure that it uses the Delegated Funds in such a way as to ensure that NHS England is able to fulfil its functions, including without limitation the Reserved Functions, effectively and efficiently in accordance with this Agreement.
- 13.7. NHS England may in respect of the Delegated Funds:
  - 13.7.1. notify the CCG of the capital resource limit and revenue resource limit that will apply in any Financial Year;
  - 13.7.2. notify the CCG regarding the payment of sums by the CCG to NHS England in respect of charges referable to the valuation or disposal of assets and such conditions as to records, certificates or otherwise:
  - 13.7.3. by notice, require the CCG to take such action or step in respect of the Delegated Funds, in order to ensure compliance by NHS England of its duties or functions under the NHS Act or the HSCA (including without limitation, Chapter 6 of the NHS Act) or any action taken or direction made by the Secretary of State under the NHS Act or the HSCA (including, without limitation, Chapter 6 of the NHS Act).
- 13.8. Schedule 5 (Financial Provisions and Decision Making Limits) sets out further financial provisions in respect of the exercise of the Delegated Functions and, in particular, Table 1 in Schedule 5 (Financial Provisions and Decision Making Limits) sets out certain financial limits and approvals required in relation to the exercise of the Delegated Functions. NHS England's Standing Financial Instructions shall be updated accordingly.

## Payment and Transfer

13.9. The CCG acknowledges that the Delegated Funds do not form part of and are separate to the funds allocated annually under section 223G of the NHS Act (the "Annual Allocation").

- 13.10. NHS England will pay the Delegated Funds to the CCG using the same revenue transfer process as used for the Annual Allocation or using such other process as notified to the CCG from time to time.
- 13.11. Without prejudice to any other obligation upon the CCG, the CCG agrees that it must deal with the Delegated Funds in accordance with:
  - 13.11.1. the terms and conditions of this Agreement;
  - 13.11.2. the business rules as set out in NHS England's planning guidance or such other documents issued by NHS England from time to time;
  - 13.11.3. any Capital Investment Guidance or Primary Medical Care Infrastructure Guidance;
  - 13.11.4. any Guidance or Contractual Notice issued by NHS England from time to time in relation to the Delegated Funds (including in relation to the form or contents of any accounts in relation to the Delegated Funds); and
  - 13.11.5. the HM Treasury guidance *Managing Public Money* (dated July 2013 and found at <a href="https://www.gov.uk/government/uploads/system/uploads/attachment\_data/file/212123/Managing\_Public\_Money\_AA\_v2\_-chapters\_annex\_web.pdf">https://www.gov.uk/government/uploads/system/uploads/attachment\_data/file/212123/Managing\_Public\_Money\_AA\_v2\_-chapters\_annex\_web.pdf</a>).
- 13.12. Without prejudice to any other obligation upon the CCG, the CCG agrees that it must provide all information, assistance and support to NHS England in relation to the audit and/or investigation (whether internal or external and whether under Law or otherwise) in relation to the use of or payment of the Delegated Funds and the discharge of the Delegated Functions.

Administrative and/or Management Services and Funds in relation to the Capital Expenditure Functions

- 13.13. The Parties acknowledge that the Capital Expenditure Functions are a Reserved Function.
- 13.14. The Parties further acknowledge that:
  - 13.14.1. accordingly, the Delegated Funds do not include any funds in respect of amounts payable in relation to the Capital Expenditure Functions ("Capital Expenditure Funds"); and

- 13.14.2. NHS England remains responsible and accountable for the discharge of the Capital Expenditure Functions and nothing in clauses 13.13 to 13.16 shall be construed as a divestment or delegation of NHS England's Capital Expenditure Functions.
- 13.15. Without prejudice to clause 13.14 above, the CCG will comply with any Guidance issued in relation to the Capital Expenditure Functions and shall (on request from NHS England) provide the following administrative services to NHS England in respect of the Capital Expenditure Funds:
  - 13.15.1. the administration and payment of sums that NHS England has approved as payable in relation to the Capital Expenditure Functions;
  - 13.15.2. if requested by NHS England and taking into account (i) any other support or services provided to NHS England by NHS Property Services Limited or otherwise and (ii) any Guidance issued in respect of the Capital Expenditure Functions, the provision of advice and/or recommendations to NHS England in respect of expenditure to be made under the Capital Expenditure Functions; and
  - 13.15.3. such other support or administrative assistance to NHS England that NHS England may reasonably request in order to facilitate the discharge by NHS England of its responsibilities under or in respect of the Capital Expenditure Functions.
- 13.16. NHS England may, at the same time as it transfers the Delegated Funds to the CCG under clause 13.10, transfer to the CCG such amounts as are necessary to enable the discharge of the CCG's obligations under this clause 13 (Financial Provisions and Liability) in respect of the Capital Expenditure Functions.

Administrative and/or Management Services and Funds in relation to Section 7A Functions

- 13.17. The Parties acknowledge that the Section 7A Functions are part of the Reserved Functions.
- 13.18. The Parties further acknowledge that:

- 13.18.1. accordingly, the Delegated Funds do not include any funds in respect of amounts payable in relation to the Section 7A Functions (whether such arrangements are included in or under Primary Medical Services Contracts or not) ("Section 7A Funds"); and
- 13.18.2. NHS England remains responsible and accountable for the discharge of the Section 7A Functions and nothing in this clause 13 (Financial Provisions and Liability) shall be construed as a divestment or delegation of the Section 7A Functions.
- 13.19. The CCG will provide the following services to NHS England in respect of the Section 7A Funds:
  - 13.19.1. the administration and payment of sums that NHS England has approved as payable under or in respect of arrangements for the Section 7A Functions: and
  - 13.19.2. such other support or administrative assistance to NHS England that NHS England may reasonably request in order to facilitate the discharge by NHS England of its responsibilities under or in respect of the Section 7A Funds.
- 13.20. NHS England shall, at the same time as it transfers the Delegated Funds to the CCG under clause 13.10, transfer to the CCG such amounts as are necessary to enable the discharge of the CCG's obligations under this clause 13 (Financial Provisions and Liability) in respect of the Section 7A Funds.

Administrative and/or Management Services and Funds in relation to other Reserved Functions

- 13.21. NHS England may ask the CCG to provide certain management and/or administrative services to NHS England (from a date to be notified by NHS England to the CCG) in relation to:
  - 13.21.1. the carrying out of any of the Reserved Functions; and/or
  - 13.21.2. without prejudice to the generality of clause 13.21.1, the handling and consideration of complaints.

- 13.22. If NHS England makes such a request to the CCG, then the CCG will, but only if the CCG agrees to provide such services, from the date requested by NHS England, comply with:
  - 13.22.1. provisions equivalent to those set out above in relation to the Capital Expenditure Functions (clauses 13.13 to 13.16) and the Section 7A Functions (clauses 13.17 to 13.20) including in relation to the administration of any funds for such functions but only to the extent that such provisions are relevant to the management or administrative services to be provided; and
  - 13.22.2. such other provisions in respect of the carrying out of such management and administrative services as agreed between NHS England and the CCG.
- 13.23. If NHS England asks the CCG to provide certain management and administrative services in relation to the handling and consideration of complaints and if the CCG agrees to provide such management and administrative services (with such agreement to be recorded as a variation pursuant to clause 22 (Variations)) then:
  - 13.23.1. NHS England may, in any Contractual Notice issued by NHS England in respect of such service (and as referred to in clause 13.22.2), specify procedures and responsibilities of the CCG and NHS England in relation to such complaints under the Complaints Regulations and all other Law; and
  - 13.23.2. such Contractual Notice may specify procedures in relation to the provision of an annual report to the Chief Executive of NHS England, procedures in relation to the approval of decisions in relation to complaints and/or the appointment of a responsible person by NHS England pursuant to the Complaints Regulations;
  - 13.23.3. such services shall be arrangements made under the provisions of Regulation 3 of the Complaints Regulations; and
  - 13.23.4. provided that any Contractual Notice issued pursuant to this clause shall be discussed and agreed with the CCG prior to the issue of the Contractual Notice by NHS England.

### Pooled Funds

- 13.24. The CCG may, for the purposes of exercising the Delegated Functions under this Agreement, establish and maintain a pooled fund in respect of any part of the Delegated Funds with NHS England in accordance with section 13V of the NHS Act except that the CCG may only do so if NHS England (at its absolute discretion) consents in writing to the establishment of the pooled fund (including any terms as to the governance and payments out of such pooled fund).
- 13.25. At the date of this agreement, details of the pooled funds (including any terms as to the governance and payments out of such pooled fund) of NHS England and the CCG are set out in the Local Terms.

Business Plan, Commissioning Plan and Annual Report

- 13.26. Within two (2) months of the date of the Delegation and thereafter three (3) months before the start of each Financial Year, the CCG must prepare a plan setting out how it proposes to exercise the Delegated Functions in that Financial Year and in each of the next two (2) Financial Years (or over such longer period as NHS England may require).
- 13.27. The plan must, in particular, explain how the CCG proposes to ensure NHS England's compliance with its duties in relation to the Delegated Functions under the NHS Act, including without limitation:
  - 13.27.1. sections 223C (expenditure), 223D (controls on total resource use) and 223E (additional controls on resource use) of the NHS Act; and
  - 13.27.2. sections 13E (duty as to improvement in quality of services),13G (duty as to reducing inequalities) and 13Q (public involvement and consultation) of the NHS Act.
- 13.28. The plan must include the following:
  - 13.28.1. details of how the CCG proposes to exercise the Delegated Functions in that Financial Year and in each of the next two (2) Financial Years; and

- 13.28.2. details of how the CCG proposes to ensure NHS England's compliance with its duties to achieve any objectives and requirements relating to the Delegated Functions which are specified in the mandate published by the Department of Health to NHS England for the first Financial Year to which the plan relates; and
- 13.28.3. any other information or detail that NHS England considers necessary to ensure NHS England's compliance with its obligations under section 13T of the NHS Act or any other provision of the NHS Act or other Law.
- 13.29. The CCG must revise the plan at the request of NHS England and submit a revised plan to NHS England before the date specified by NHS England from time to time.
- 13.30. As soon as practicable after the end of each Financial Year (and in any event within two (2) months of the end of each Financial Year or such longer period as NHS England may specify), the CCG must provide to NHS England a report on how the CCG has exercised the Delegated Functions during the previous Financial Year.
- 13.31. The report referred to in clause 13.30 above must include sufficient detail to ensure NHS England's compliance with its statutory obligations under section 13U of the NHS Act.
- 13.32. Following receipt of the report referred to in clause 13.30 above, NHS England may (at its absolute discretion) require such further information from the CCG as NHS England considers necessary to ensure NHS England's compliance with its obligations under section 13U of the NHS Act.
- 13.33. The CCG shall comply with any Contractual Notices issued from time to time by NHS England in relation to the inclusion of information in relation to the Delegated Functions in any plan prepared by the CCG under section 14Z11 of the NHS Act or in any report prepared under section 14Z15 of the NHS Act.

Risk sharing

- 13.34. In accordance with section 13Z(6) of the NHS Act, NHS England retains liability in relation to the exercise of the Delegated Functions and nothing in this Agreement affects the liability of NHS England in relation to the Delegated Functions.
- 13.34A For the avoidance of doubt, NHS England retains liability in respect of any Losses arising in respect of NHS England's negligence, fraud, recklessness or deliberate breach in respect of the Delegated Functions and, if the CCG suffers any Losses in respect of such actions by NHS England, NHS England shall make such adjustments to the Delegated Funds (or other amounts payable to the CCG) in order to reflect any Losses suffered by the CCG (except to the extent that the CCG is liable for such Loss pursuant to clause 13.35).
- 13.35. The CCG is liable (and shall pay) to NHS England for any Losses suffered by NHS England that result from or arise out of the CCG's negligence, fraud, recklessness or deliberate breach of the Delegation (including any actions that are taken that exceed the authority conferred by the Delegation) or this Agreement and, in respect of such Losses, NHS England may, at its discretion and without prejudice to any other rights, either require payment from the CCG or make such adjustments to the Delegated Funds pursuant to clause 13.3. The CCG shall not be liable to the extent that the Losses arose prior to the date of this Agreement.
- 13.36. Nothing in this clause 13 (Financial Provisions and Liability) or this Agreement shall affect or prejudice NHS England's right to exercise its rights (whether arising under administrative law, common law or statute) in relation to actions or steps of the CCG, including any actions or steps that exceed the authority conferred by the Delegation or are a breach of the terms and conditions of this Agreement.

## 14. Claims and Litigation

14.1. Schedule 2 (*Delegated Functions*) sets out further detail in relation to the performance management of the Primary Medical Services Contracts.

- 14.2. Nothing in this clause 14 (Claims and Litigation) shall be interpreted as affecting the reservation to NHS England of the Reserved Functions (including the reservation to NHS England of all functions in relation to the performers list activities).
- 14.3. Except in the circumstances set out in clause 14.7 and subject always to compliance with this clause 14 (*Claims and Litigation*), the CCG shall be responsible for and shall retain the conduct of any Claim.

### 14.4. The CCG must:

- 14.4.1. comply with any policy issued by NHS England from time to time in relation to the conduct of or avoidance of Claims and/or the pro-active management of Claims;
- 14.4.2. without prejudice to clause 14.4.1, in respect of legal advice or assistance in relation to a Claim, comply with any requirements of NHS England from time to time (whether set out in a policy issued pursuant to clause 14.4.1 or otherwise) in relation to the use of solicitors or barristers and, at the date of this Agreement, NHS England's requirement is that a CCG must obtain prior approval from NHS England in respect of the firm of solicitors instructed to provide legal advice or assistance in relation to a Claim;
- 14.4.3. if it receives any correspondence, issue of proceedings, claim document or other document concerning any Claim or potential Claim, immediately notify NHS England and send to NHS England all copies of such correspondence;
- 14.4.4. co-operate fully with NHS England in relation to such Claim and the conduct of such Claim;
- 14.4.5. provide, at its own cost, to NHS England all documentation and other correspondence that NHS England requires for the purposes of considering and/or resisting such Claim; and/or
- 14.4.6. at the request of NHS England, take such action or step or provide such assistance as may in NHS England's discretion be necessary or desirable having regard to the nature of the Claim and the existence of any time limit in relation to avoiding, disputing, defending, resisting, appealing, seeking a review or compromising such Claim or to comply with the

requirements of the NHSLA or any insurer in relation to such Claim.

- 14.5. NHS England shall use its reasonable endeavours to keep the CCG informed in respect of the conduct and/or outcome of the Claim except that NHS England shall have no obligation to do so due to any administrative or regulatory requirement, the requirement of any insurer or the NHSLA or for any other reason that NHS England may consider necessary or appropriate, at its absolute discretion, in relation to the conduct of that Claim or related matter.
- 14.6. Subject to clause 14.4 and Schedule 5 (Financial Provisions and Decision Making Limits) the CCG is entitled to conduct the Claim in the manner it considers appropriate and is also entitled to pay or settle any Claim on such terms as it thinks fit.

NHS England Stepping into Claims

14.7. NHS England may, at any time following discussion with the CCG, send a notice to the CCG stating that NHS England will take over the conduct of the Claim and the CCG must immediately take all steps necessary to transfer the conduct of such Claim to NHS England. In such cases, NHS England shall be entitled to conduct the Claim in the manner it considers appropriate and is also entitled to pay or settle any Claim on such terms as it thinks fit.

NHS England Stepping out of Claims

14.8. NHS England may, at any time after it has exercised its rights set out in clause 14.7 above and following discussion with the CCG, send a notice to the CCG stating that the CCG will be required to take over the conduct of the Claim from NHS England and NHS England must immediately take all steps necessary to transfer the conduct of such Claim to the CCG. In such cases, the CCG shall be entitled to conduct the Claim in the manner it considers appropriate in accordance with its obligations under this clause 14 (Claims and Litigation) and subject to Schedule 4 (Further Information Sharing Provisions) and Schedule 5 (Financial Provisions and Decision Making Limits).

Claim Losses

- 14.9. The CCG and NHS England shall notify each other within a reasonable time period of becoming aware of any Claim Losses.
- 14.10. If the CCG considers that, as a result of a Claim Loss, the Delegated Funds will be insufficient to meet the Claim Loss as well as discharge the Delegated Functions, then the CCG shall immediately notify NHS England and the Parties shall meet to discuss and agree any adjustment that may be needed pursuant to clause 13.3 (and taking into account any funds, provisions or other resources retained by NHS England in respect of such Claim Losses).
- 14.11. The CCG acknowledges that NHS England will pay to the CCG the funds that are attributable to the Delegated Functions. Accordingly, the CCG acknowledges that the Delegated Funds are required to be used to discharge and/or pay any Claim Losses. NHS England may, in respect of any Claim Losses, at its discretion and without prejudice to any other rights, either require payment from the CCG for such Claim Losses or pursuant to clause 13.3 make such adjustments to the Delegated Funds to take into account the amount of any Claim Losses (other than any Claim Losses in respect of which NHS England has retained any funds, provisions or other resources to discharge such Claim Losses). For the avoidance of doubt, in circumstances where NHS England suffers any Claim Losses, then NHS England shall be entitled to recoup such Claim Losses pursuant to clause 13.3. If and to the extent that NHS England has retained any funds, provisions or other resources to discharge such Claim Losses, then NHS England may either use such funds to discharge the Claim Loss or make an upward adjustment to the amounts paid to the CCG pursuant to clause 13.3.

### 15. Breach

- 15.1. If the CCG does not comply with the Delegation or the terms of this Agreement, then NHS England may:
  - 15.1.1. exercise its rights under this Agreement; and/or
  - 15.1.2. take such steps as it considers appropriate under the CCG Assurance Framework.

- 15.2. Without prejudice to clause 15.1, if the CCG does not comply with the Delegation or the terms of this Agreement (including if the CCG exceeds its delegated authority under the Delegation), NHS England may (at its sole discretion):
  - 15.2.1. waive such non-compliance in accordance with clause 15.3 and the Delegation;
  - 15.2.2. ratify any decision in accordance with paragraph 15 of the Delegation;
  - 15.2.3. revoke the Delegation and terminate this Agreement in accordance with clause 17 (*Termination*) below;
  - 15.2.4. exercise the Escalation Rights in accordance with clause 16 (Escalation Rights); and/or
  - 15.2.5. exercise its rights under common law.
- 15.3. NHS England may waive any non-compliance by the CCG with the terms of this Agreement provided that the CCG provides a written report to NHS England pursuant to clause 15.4 and, after considering the CCG's written report, NHS England is satisfied that the waiver is justified.

#### 15.4. If:

- 15.4.1. the CCG does not comply (or the CCG considers that it may not be able to comply) with this Agreement and/or the Delegation; or
- 15.4.2. NHS England notifies the CCG that it considers the CCG has not complied, or may not be able to comply with, this Agreement and/or the Delegation,

then the CCG must provide a written report to NHS England within ten (10) days of the non-compliance (or the date on which the CCG considers that it may not be able to comply with this Agreement) or such notification pursuant to clause 15.4.2 setting out:

- 15.4.3. details of and reasons for the non-compliance (or likely non-compliance) with the Agreement and/or the Delegation; and
- 15.4.4. a plan for how the CCG proposes to remedy the non-compliance.

## 16. Escalation Rights

- 16.1. If the CCG does not comply with this Agreement and/or the Delegation, NHS England may exercise the following Escalation Rights:
  - 16.1.1. NHS England may require a suitably senior representative of the CCG to attend a review meeting within ten (10) days of NHS England becoming aware of the non-compliance; and
  - 16.1.2. NHS England may require the CCG to prepare an action plan and report within twenty (20) days of the review meeting (to include details of the non-compliance and a plan for how the CCG proposes to remedy the non-compliance).
- 16.2. Nothing in clause 16 (Escalation Rights) will affect NHS England's right to revoke the Delegation and/or terminate this Agreement in accordance with clause 17 (Termination) below.

### 17. Termination

- 17.1. The CCG may:
  - 17.1.1. notify NHS England that it requires NHS England to revoke the Delegation; and
  - 17.1.2. terminate this Agreement

with effect from midnight on 31 March in any calendar year, provided that:

- 17.1.3. on or before 30 September of the previous calendar year, the CCG sends written notice to NHS England of its requirement that NHS England revoke the Delegation and intention to terminate this Agreement; and
- 17.1.4. the CCG meets with NHS England within ten (10) Operational Days of NHS England receiving the notice set out at clause 17.1.3 above to discuss arrangements for termination and transition of the Delegated Functions to a successor commissioner.

- in which case NHS England shall revoke the Delegation and this Agreement shall terminate with effect from midnight on 31 March in the next calendar year.
- 17.2. NHS England may revoke the Delegation at midnight on 31 March in any year, provided that it gives notice to the CCG of its intention to terminate the Delegation on or before 30 September in the year prior to the year in which the Delegation will terminate, and in which case clause 17.4 will apply.
- 17.3. The Delegation may be revoked and this Agreement may be terminated by NHS England at any time, including in (but not limited to) the following circumstances:
  - 17.3.1. the CCG acts outside of the scope of its delegated authority;
  - 17.3.2. the CCG fails to perform any material obligation of the CCG owed to NHS England under the Delegation or this Agreement;
  - 17.3.3. the CCG persistently commits non-material breaches of the Delegation or this Agreement;
  - 17.3.4. NHS England is satisfied that its intervention powers under section 14Z21 of the NHS Act apply;
  - 17.3.5. to give effect to legislative changes;
  - 17.3.6. failure to agree to a National Variation in accordance with clause 22 (*Variations*);
  - 17.3.7. NHS England and the CCG agree in writing that the Delegation shall be revoked and this Agreement shall terminate on such date as is agreed; and/or
  - 17.3.8. the CCG merges with another CCG or other body.
- 17.4. This Agreement will terminate immediately upon revocation or termination of the Delegation (including revocation and termination in accordance with this clause 17 (*Termination*)) except that the Survival Clauses will continue in full force and effect. This Agreement shall not terminate immediately if the Delegation is amended by a revocation and re-issue of an amended Delegation.

- 17.5. Upon revocation or termination of the Delegation and this Agreement (including revocation and termination in accordance with this clause 17 (*Termination*)), the Parties must:
  - 17.5.1. agree a plan for the transition of the Delegated Functions from the CCG to the successor commissioner, including details of the transition, the Parties' responsibilities in relation to the transition, the Parties' arrangements in respect of those staff engaged in the Delegated Functions and the date on which the successor commissioner will take responsibility for the Delegated Functions;
  - 17.5.2. implement and comply with their respective obligations under the plan for transition agreed in accordance with clause 17.5.1 above; and
  - 17.5.3. use all reasonable endeavours to minimise any inconvenience or disruption to the commissioning of healthcare in the Area.
- 17.6. Without prejudice to clause 15.3 and for the avoidance of doubt, NHS England may waive any right to terminate this Agreement under this clause 17 (*Termination*).

## 18. Staffing

- 18.1. The Parties acknowledge and agree that the CCG may only engage staff to undertake the Delegated Functions under one of the following three staffing models:
  - 18.1.1. "Model 1 Assignment" under the terms of which the staff of NHS England remain in their current roles and locations and provide services to the CCG under a service level agreement;
  - 18.1.2. "Model 2 Secondment" under the terms of which certain staff of NHS England are seconded to the CCG (and, for the avoidance of doubt, such secondments will terminate on revocation or termination of the Delegation); or
  - 18.1.3. "Model 3 Employment" under the terms of which the CCG may create new posts within the CCG to undertake the Delegated Functions provided that the CCG may only do so if it first offers to existing staff of NHS England an opportunity to

apply for such posts and such staff must be appointed if they are deemed appointable,

together, the "Staffing Models".

- 18.2. The CCG and NHS England, must within six (6) months of the date of this Agreement, agree which of the Staffing Models (set out at clauses 18.1.1 to 18.1.3 above) will be adopted by the CCG and the date on which such Staffing Model shall take effect.
- 18.3. In the absence of any agreement under clause 18.2, and up until such date as the CCG's preferred Staffing Model shall take effect (as referred to in clause 18.2 above), Model 1 described in clause 18.1.1 above will apply. The terms on which Model 1 will apply are set out in Schedule 8 (Assignment of NHS England Staff to the CCG).
- 18.4. The CCG must comply with any Guidance issued by NHS England from time to time in relation to the Staffing Models and such Guidance may make changes to the Staffing Models from time to time.
- 18.5. For the avoidance of doubt, any breach by the CCG of the terms of this clause 18 (Staffing), including any breach of the Guidance issued in accordance with clause 18.4 above, will be a breach of the terms and conditions of this Agreement for the purposes of clauses 13.3 and 13.35.
- 18.6. Without prejudice to clause 18.7, it is the understanding of the Parties that the provisions of the Transfer Regulations will not operate to transfer the employment of any staff of NHS England or any other party to the CCG on the commencement of the Delegation and this Agreement.
- 18.7. The Parties acknowledge that if at any time before or after the revocation or termination of the Delegation and this Agreement the Transfer Regulations do apply, the Parties must co-operate and comply with their obligations under the Transfer Regulations.

### 19. Disputes

19.1. This clause does not affect NHS England's right to take action under the CCG Assurance Framework.

- 19.2. If a dispute arises out of or in connection with this Agreement or the Delegation ("**Dispute**") then the Parties must follow the procedure set out in this clause:
  - 19.2.1. either Party must give to the other written notice of the Dispute, setting out its nature and full particulars ("Dispute Notice"), together with relevant supporting documents. On service of the Dispute Notice, the Agreement Representatives must attempt in good faith to resolve the Dispute;
  - 19.2.2. if the Agreement Representatives are, for any reason, unable to resolve the Dispute within twenty (20) days of service of the Dispute Notice, the Dispute must be referred to the Accountable Officer (or equivalent person) of the CCG and a director of or other person nominated by NHS England (and who has authority from NHS England to settle the Dispute) who must attempt in good faith to resolve it; and
  - 19.2.3. if the people referred to in clause 19.2.2 are for any reason unable to resolve the Dispute within twenty (20) days of it being referred to them, the Parties may attempt to settle it by mediation in accordance with the CEDR model mediation procedure. Unless otherwise agreed between the Parties, the mediator must be nominated by CEDR Solve. To initiate the mediation, a Party must serve notice in writing ("ADR notice") to the other Party to the Dispute, requesting a mediation. A copy of the ADR notice should be sent to CEDR Solve. The mediation will start not later than ten (10) days after the date of the ADR notice.
- 19.3. If the Dispute is not resolved within thirty (30) days after service of the ADR notice, or either Party fails to participate or to continue to participate in the mediation before the expiration of the period of thirty (30) days, or the mediation terminates before the expiration of the period of thirty (30) days, the Dispute must be referred to the Secretary of State, who shall resolve the matter and whose decision shall be binding upon the Parties.

### 20. Freedom of Information

- 20.1. Each Party acknowledges that the other is a public authority for the purposes of the Freedom of Information Act 2000 ("FOIA") and the Environmental Information Regulations 2004 ("EIR").
- 20.2. Each Party may be statutorily required to disclose further information about the Agreement and the Relevant Information in response to a specific request under FOIA or EIR, in which case:
  - 20.2.1. each Party shall provide the other with all reasonable assistance and co-operation to enable them to comply with their obligations under FOIA or EIR;
  - 20.2.2. each Party shall consult the other regarding the possible application of exemptions in relation to the information requested; and
  - 20.2.3. subject only to clause 14 (Claims and Litigation), each Party acknowledges that the final decision as to the form or content of the response to any request is a matter for the Party to whom the request is addressed.
- 20.3. NHS England may, from time to time, issue a FOIA or EIR protocol or update a protocol previously issued relating to the dealing with and responding to of FOIA or EIR requests in relation to the Delegated Functions. The CCG shall comply with such FOIA or EIR protocols.

### 21. Conflicts of Interest

- 21.1. The CCG must comply with its statutory duties set out in:
  - 21.1.1. Chapter A2 of the NHS Act (including those statutory duties relating to the management of conflicts of interest as set out at section 14O of the NHS Act);
  - 21.1.2. the National Health Service (Procurement, Patient Choice and Competition) (No. 2) Regulations 2013/500; and
  - 21.1.3. Regulation 24 of the Public Contracts Regulations 2015/102,

- and must perform its obligations under this Agreement in such a way as to ensure NHS England's compliance with its statutory duties in relation to conflicts of interest.
- 21.2. The CCG must have regard to all relevant guidance published by NHS England in relation to conflicts of interest in the co-commissioning context.

### 22. Variations

- 22.1. The Parties acknowledge that, under paragraph 16 of the Delegation, the Delegation may be reviewed and amended from time to time and that such amendments may be effected by a revocation and re-issue of an amended Delegation.
- 22.2. The Parties acknowledge that, under paragraph 6 of the Delegation, certain additional functions may be delegated from time to time by NHS England to the CCG on a date or dates to be notified to the CCG by NHS England in accordance with clause 8.3. If NHS England amends the Delegation and/or delegates additional functions to the CCG, then NHS England and the CCG shall agree such consequential changes to this Agreement pursuant to this clause 22 (Variations).
- 22.3. Subject to clauses 22.4 to 22.10 below, a variation of this Agreement will only be effective if:
  - 22.3.1. it is materially in the form of the template variation agreement set out at Schedule 6 (*Template Variation Agreement*); and
  - 22.3.2. it is signed by NHS England and the CCG (by their Agreement Representatives or other duly authorised representatives).
- 22.4. The Parties may not vary any provision of this Agreement if the purported variation would contradict or conflict with the Delegation.
- 22.5. NHS England may notify the CCG of any proposed National Variation by issuing a National Variation Proposal by whatever means NHS England may consider appropriate from time to time.

- 22.6. The CCG will be deemed to have received a National Variation Proposal on the date that it is issued by NHS England.
- 22.7. The National Variation Proposal will set out the National Variation proposed and the date on which NHS England requires the National Variation to take effect.
- 22.8. The CCG must respond to a National Variation Proposal within thirty (30) Operational Days following the date that it is issued by serving a written notice on NHS England confirming either:
  - 22.8.1. that it accepts the National Variation Proposal; or
  - 22.8.2. that it refuses to accept the National Variation Proposal, and setting out reasonable grounds for that refusal.
- 22.9. If the CCG accepts the National Variation Proposal in accordance with clause 22.8.1, the CCG agrees (without delay) to take all necessary steps (including executing a variation agreement) in order to give effect to any National Variation by the date on which the proposed National Variation takes effect as set out in the National Variation Proposal.
- 22.10. If the CCG refuses to accept the National Variation Proposal in accordance with clause 22.8.2 or to take such steps as set out in clause 22.9, NHS England may terminate this Agreement and revoke the Delegation in accordance with clause 17.3.6.

## 23. Counterparts

23.1. This Agreement may be executed in counterparts, each of which shall be regarded as an original, but all of which together shall constitute one agreement binding on both of the Parties.

### 24. Notices

- 24.1. Any notices given under this Agreement must be in writing, must be marked for the appropriate department or person and must be served by hand, post or email to the following address:
  - 24.1.1. in the case of NHS England, to NHS England's address for notices set out in the Particulars; or

24.1.2. in the case of the CCG, to the CCG's address for notices set out in the Particulars.

### 24.2. Notices sent:

- 24.2.1. by hand will be effective upon delivery;
- 24.2.2. by post will be effective upon the earlier of actual receipt or five (5) working days after mailing; or
- 24.2.3. by email will be effective when sent (subject to no automated response being received).
- 24.3. NHS England may, at its discretion, issue Contractual Notices from time to time relating to the manner in which the Delegated Functions should be exercised by the CCG.
- 24.4. NHS England may, at its discretion, issue Guidance from time to time, including any protocol, policy, guidance or manual relating to the exercise of the Delegated Functions under this Agreement. NHS England acknowledges that in considering the need and/or content of new Guidance it will engage appropriately with CCGs.

# Schedule 1 Definitions and Interpretation

In this Agreement, the following words and phrases will bear the following meanings:

Agreement	means this agreement between	NHS England and the

CCG comprising the Particulars, the Terms and

Conditions and the Schedules;

Agreement means the CCG Representative and the NHS England

**Representatives** Representative as set out in the Particulars;

APMS Contract means an agreement made in accordance with section

92 of the NHS Act;

**Assigned Staff** means those NHS England staff as agreed between

NHS England and the CCG from time to time;

Caldicott Principles means the patient confidentiality principles set out in the

report of the Caldicott Committee (December 1997 as amended by the 2013 Report, The Information Governance Review – "To Share or Not to Share?") and now included in the NHS Confidentiality Code of

Practice, as may be amended from time to time;

Capital shall have the meaning set out in the Capital Investment

Guidance or such other replacement Guidance as

issued by NHS England from time to time;

**Capital Expenditure** 

**Functions** 

means those functions of NHS England in relation to the use and expenditure of Capital funds (but excluding the

Premises Costs Directions Functions);

Capital Investment

Guidance

means any Guidance issued by NHS England from time to time in relation to the development, assurance and approvals process for proposals in relation to:

- the expenditure of Capital, or investment in property, infrastructure or information and technology; or
- the revenue consequences for commissioners or

## third parties making such investment;

CCG Assurance Framework

means the assurance framework that applies to CCGs

pursuant to the NHS Act;

**Claims** 

means, for or in relation to the Primary Medical Services Contracts (a) any litigation or administrative, mediation, arbitration or other proceedings, or any claims, actions or hearings before any court, tribunal or any governmental, regulatory or similar body, or any department, board or agency or (b) any dispute with, or any investigation, inquiry or enforcement proceedings by, any governmental, regulatory or similar body or agency;

Claim Losses

means all Losses arising in relation to any Claim;

**Complaints Regulations** 

means the Local Authority Social Services and National Health Service Complaints (England) Regulations 2009/309;

**Contractual Notice** 

means a contractual notice issued by NHS England to the CCG or all CCGs (as the case may be) from time to time and relating to the manner in which the Delegated Functions should be exercised by the CCG, in accordance with clause 24.3;

CQC

means the Care Quality Commission;

**Data Controller** 

shall have the same meaning as set out in the DPA;

**Data Subject** 

shall have the same meaning as set out in the DPA;

**Delegated Functions** 

means the functions delegated by NHS England to the CCG under the Delegation and as set out in detail in this Agreement;

**Delegated Funds** 

shall have the meaning in clause 13.1;

**DPA** 

means the Data Protection Act 1998;

#### **Enhanced Services**

means the nationally defined enhanced services, as set out in the Primary Medical Services (Directed Enhanced Services) Directions 2014 or as amended from time to time, and any other enhanced services schemes locally developed by the CCG in the exercise of its Delegated Functions (and excluding, for the avoidance of doubt, any enhanced services arranged or provided pursuant to the Section 7A Functions);

## **Escalation Rights**

means the escalation rights as defined in clause 16 (Escalation Rights);

#### **Financial Year**

shall bear the same meaning as in section 275 of the NHS Act;

### **GMS Contract**

means a general medical services contract made under section 84(1) of the NHS Act;

#### **Good Practice**

means using standards, practices, methods and procedures conforming to the law, reflecting up-to-date published evidence and exercising that degree of skill and care, diligence, prudence and foresight which would reasonably and ordinarily be expected from a skilled, efficient and experienced commissioner;

## Guidance

means any protocol, policy, guidance or manual (issued by NHS England whether under this Agreement or otherwise) and/or any policy or guidance relating to the exercise of the Delegated Functions issued by NHS England from time to time, in accordance with clause 24.4;

#### **HSCA**

means the Health and Social Care Act 2012;

#### **Information Law**

the DPA, the EU Data Protection Directive 95/46/EC; regulations and guidance made under section 13S and section 251 of the NHS Act; guidance made or given under sections 263 and 265 of the HSCA; the Freedom of Information Act 2000; the common law duty of confidentiality; the Human Rights Act 1998 and all other applicable laws and regulations relating to processing of

## Personal Data and privacy;

Law

means any applicable law, statute, bye-law, regulation, direction, order, regulatory policy, guidance or code, rule of court or directives or requirements of any regulatory body, delegated or subordinate legislation or notice of any regulatory body (including, for the avoidance of doubt, the Premises Costs Directions, the Statement of Financial Entitlements Directions and the Primary Medical Services (Directed Enhanced Services) Directions 2014 as amended from time to time);

Local Incentive Schemes

Incentive means an incentive scheme developed by the CCG in the exercise of its Delegated Functions including (without limitation) as an alternative to QOF;

**Local Terms** means the terms set out in Schedule 7 (Local Terms);

means all damages, loss, liabilities, claims, actions, costs, expenses (including the cost of legal and/or professional services) proceedings, demands and charges;

**National Variation** 

Losses

an addition, deletion or amendment to the provisions of this Agreement mandated by NHS England (whether in respect of the CCG or all or some of other Clinical Commissioning Groups) including any addition, deletion or amendment to reflect changes to the Delegation, changes in Law, changes in policy and notified to the CCG in accordance with clause 22 (Variations);

National Variation Proposal

**Variation** a written proposal for a National Variation, which complies with the requirements of clause 22.7;

**Need to Know** has the meaning set out in paragraph 6.2 of Schedule 4 (Further Information Sharing Provisions);

NHS Act means the National Health Service Act 2006 (as amended by the Health and Social Care Act 2012 or

other legislation from time to time);

**NHS England** means the National Health Service Commissioning

Board established by section 1H of the NHS Act, also

known as NHS England;

**Non-Personal Data** means data which is not Personal Data:

**Operational Days** a day other than a Saturday, Sunday or bank holiday in

England;

**Particulars** means the Particulars of this Agreement as set out in

clause 1 (Particulars);

**Party/Parties** means a party or both parties to this Agreement;

**Personal Data** shall have the same meaning as set out in the DPA and

shall include references to Sensitive Personal Data

where appropriate;

**Personal** 

Agreement

**Data** means the agreement governing Information Law issues

completed further to Schedule 4 (Further Information

Sharing Provisions);

**Personnel** means the Parties' employees, officers,

> members, directors, voluntary staff, consultants, and other contractors and sub-contractors acting on behalf of either Party (whether or not the arrangements with such contractors and sub-contractors are subject to legally binding contracts) and such contractors' and their

sub-contractors' personnel;

**PMS Contract** means an arrangement or contract for the provision of

> primary medical services made under section 83(2) of the NHS Act (including any arrangements which are made in reliance on a combination of that section and other powers to arrange for primary medical services);

**Premises Agreements** means tenancies, leases and other arrangements in

relation to the occupation of land for the delivery of

services under the Primary Medical Services Contracts;

**Premises** Costs means the National Health Service (General Medical

#### **Directions**

Services Premises Costs) Directions 2013, as amended;

# Premises Costs Directions Functions

means NHS England's functions in relation to the Premises Costs Directions:

# Primary Medical Care Infrastructure Guidance

means any Guidance issued by NHS England from time to time in relation to the procurement, development and management of primary medical care infrastructure and which may include principles of best practice;

# Primary Medical Services Contracts

#### means:

- PMS Contracts:
- GMS Contracts; and
- APMS Contracts,

in each case as amended or replaced from time to time and including all ancillary or related agreements directly relating to the subject matter of such agreements, contracts or arrangements but excluding any Premises Agreements;

## **GP Access Fund**

Means the former Prime Minister's challenge fund, announced in October 2013 to help improve access to general practice and stimulate innovative ways of providing primary care services;

# Principles of Best Practice

means the Guidance in relation to property and investment which is to be published either before or after the date of this Agreement;

## QOF

means the quality and outcomes framework;

## **Relevant Information**

means the Personal Data and Non-Personal Data processed under the Delegation and this Agreement, and includes, where appropriate, "confidential patient information" (as defined under section 251 of the NHS Act), and "patient confidential information" as defined in the 2013 Report, The Information Governance Review – "To Share or Not to Share?");

**Reserved Functions** means the functions relating to the commissioning of

> primary medical services which are reserved to NHS England (and are therefore not delegated to the CCG under the Delegation) and as set out in detail in clause 8.2 and Schedule 3 (Reserved Functions) of this

Agreement;

**Secretary of State** means the Secretary of State for Health from time to

time;

**Section 7A Functions** means those functions of NHS England exercised

pursuant to section 7A of the NHS Act relating to

primary medical services;

**Section 7A Funds** shall have the meaning in clause 13.18.1;

**Sensitive Personal Data** shall have the same meaning as in the DPA;

Specified Purpose means the purpose for which the Relevant Information is

> shared and processed, being to facilitate the exercise of the CCG's Delegated Functions and NHS England's Reserved Functions as specified in paragraph 2.1 of Schedule 4 (Further Information Sharing Provisions) to

this Agreement;

Statement of Financial means the General Medical Services Statement of

**Entitlements Directions** Financial Entitlements Directions 2013, as amended or

updated from time to time;

**Statutory Guidance** means any applicable health and social care guidance,

> guidelines, direction or determination, framework, standard or requirement to which the CCG and/or NHS England have a duty to have regard, to the extent that the same are published and publicly available or the existence or contents of them have been notified to the

CCG by NHS England from time to time;

**Survival Clauses** means clauses 10 (Information Sharing and Information

> Governance), 13 (Financial Provisions and Liability), 14 (Claims and Litigation) 17 (Termination), 18 (Staffing),

> 19 (Disputes) and 20 (Freedom of Information), together

with such other provisions as are required to interpret these clauses (including the Schedules to this Agreement); and

# **Transfer Regulations**

means the Transfer of Undertakings (Protection of Employment) Regulations 2006, as amended.

# Schedule 2 Delegated Functions

## Part 1: Delegated Functions: Specific Obligations

#### 1. Introduction

1.1. This Part 1 of Schedule 2 (*Delegated Functions*) sets out further provision regarding the carrying out of each of the Delegated Functions.

# 2. Primary Medical Services Contract Management

#### 2.1. The CCG must:

- 2.1.1. manage the Primary Medical Services Contracts on behalf of NHS England and perform all of NHS England's obligations under each of the Primary Medical Services Contracts in accordance with the terms of the Primary Medical Services Contracts as if it were named in the contract in place of NHS England;
- 2.1.2. actively manage the performance of the counter-party to the Primary Medical Services Contracts in order to secure the needs of people who use the services, improve the quality of services and improve efficiency in the provision of the services including by taking timely action to enforce contractual breaches and serve notice;
- 2.1.3. ensure that it obtains value for money under the Primary Medical Services Contracts on behalf of NHS England and avoids making any double payments under any Primary Medical Services Contracts;
- 2.1.4. comply with all current and future relevant national Guidance regarding PMS reviews and the management of practices receiving Minimum Practice Income Guarantee (MPIG) (including without limitation the *Framework for Personal Medical Services (PMS) Contracts Review* guidance published by NHS England in September 2014 (<a href="https://www.england.nhs.uk/2014/02/pms/">https://www.england.nhs.uk/2014/02/pms/</a> And <a href="https://www.england.nhs.uk/commissioning/wp-">https://www.england.nhs.uk/commissioning/wp-</a>

content/uploads/sites/12/2016/05/implement-pms-fund-changes.pdf ));

- 2.1.5. notify NHS England immediately (or in any event within two (2) Operational Days) of any breach by the CCG of its obligations to perform any of NHS England's obligations under the Primary Medical Services Contracts;
- 2.1.6. keep a record of all of the Primary Medical Services Contracts that the CCG manages on behalf of NHS England setting out the following details in relation to each Primary Medical Services Contract:
  - 2.1.6.1. name of counter-party;
  - 2.1.6.2. location of provision of services; and
  - 2.1.6.3. amounts payable under the contract (if a contract sum is payable) or amount payable in respect of each patient (if there is no contract sum).
- 2.2. For the avoidance of doubt, all Primary Medical Services Contracts will be in the name of NHS England.
- 2.3. The CCG must comply with any Guidance in relation to the issuing and signing of Primary Medical Services Contracts in the name of NHS England.
- 2.4. Without prejudice to clause 13 (Financial Provisions and Liability) or paragraph 2.1 above, the CCG must actively manage each of the relevant Primary Medical Services Contracts including by:
  - 2.4.1. managing the relevant Primary Medical Services Contract, including in respect of quality standards, incentives and the QOF, observance of service specifications, and monitoring of activity and finance;
  - 2.4.2. assessing quality and outcomes (including clinical effectiveness, patient experience and patient safety);
  - 2.4.3. managing variations to the relevant Primary Medical Services Contract or services in accordance with national policy, service user needs and clinical developments;
  - 2.4.4. agreeing information and reporting requirements and managing information breaches (which will include use of the NHS Digital IG Toolkit SIRI system);

- 2.4.5. agreeing local prices, managing agreements or proposals for local variations and local modifications:
- 2.4.6. conducting review meetings and undertaking contract management including the issuing of contract queries and agreeing any remedial action plan or related contract management processes; and
- 2.4.7. complying with and implementing any relevant Guidance issued from time to time.

#### **Enhanced Services**

- 2.5. The CCG must manage the design and commissioning of Enhanced Services, including re-commissioning these services annually where appropriate.
- 2.6. The CCG must ensure that it complies with any Guidance in relation to the design and commissioning of Enhanced Services.
- 2.7. When commissioning newly designed Enhanced Services, the CCG must:
  - 2.7.1. consider the needs of the local population in the Area;
  - 2.7.2. support Data Controllers in providing 'fair processing' information as required by the DPA;
  - 2.7.3. develop the necessary specifications and templates for the Enhanced Services, as required to meet the needs of the local population in the Area;
  - 2.7.4. when developing the necessary specifications and templates for the Enhanced Services, ensure that value for money will be obtained;
  - 2.7.5. consult with Local Medical Committees, each relevant Health and Wellbeing Board and other stakeholders in accordance with the duty of public involvement and consultation under section 14Z2 of the NHS Act;
  - 2.7.6. obtain the appropriate read codes, to be maintained by NHS Digital;
  - 2.7.7. liaise with system providers and representative bodies to ensure that the system in relation to the Enhanced Services will be functional and secure; and

2.7.8. support GPs in entering into data processing agreements with data processors in the terms required by the DPA.

## **Design of Local Incentive Schemes**

- 2.8. The CCG may design and offer Local Incentive Schemes for GP practices, sensitive to the needs of their particular communities, in addition to or as an alternative to the national framework (including as an alternative to QOF or directed Enhanced Services), provided that such schemes are voluntary and the CCG continues to offer the national schemes.
- 2.9. There is no formal approvals process that the CCG must follow to develop a Local Incentive Scheme, although any proposed new Local Incentive Scheme:
  - 2.9.1. is subject to consultation with the Local Medical Committee;
  - 2.9.2. must be able to demonstrate improved outcomes, reduced inequalities and value for money; and
  - 2.9.3. must reflect the changes agreed as part of the national PMS reviews.
- 2.10. The ongoing assurance of any new Local Incentive Schemes will form part of the CCG's assurance process under the CCG Assurance Framework.
- 2.11. Any new Local Incentive Scheme must be implemented without prejudice to the right of GP practices operating under a GMS Contract to obtain their entitlements which are negotiated and set nationally.
- 2.12. NHS England will continue to set national standing rules, to be reviewed annually, and the CCG must comply with these rules which shall for the purposes of this Agreement be Guidance.

## **Making Decisions on Discretionary Payments**

2.13. The CCG must manage and make decisions in relation to the discretionary payments to be made to GP practices in a consistent, open and transparent way.

2.14. The CCG must exercise its discretion to determine the level of payment to GP practices of discretionary payments, in accordance with the Statement of Financial Entitlements Directions.

# Making Decisions about Commissioning Urgent Care for Out of Area Registered Patients

- 2.15. The CCG must manage the design and commissioning of urgent care services (including home visits as required) for its patients registered out of area (including re-commissioning these services annually where appropriate).
- 2.16. The CCG must ensure that it complies with any Guidance in relation to the design and commissioning of these services.

## 3. Planning the Provider Landscape

- 3.1. The CCG must plan the primary medical services provider landscape in the Area, including considering and taking decisions in relation to:
  - 3.1.1. establishing new GP practices in the Area;
  - 3.1.2. managing GP practices providing inadequate standards of patient care;
  - 3.1.3. the procurement of new Primary Medical Services Contracts (in accordance with any procurement protocol issued by NHS England from time to time);
  - 3.1.4. closure of practices and branch surgeries;
  - 3.1.5. dispersing the lists of GP practices;
  - 3.1.6. agreeing variations to the boundaries of GP practices; and
  - 3.1.7. coordinating and carrying out the process of list cleansing in relation to GP practices, according to any policy or Guidance issued by NHS England from time to time.
- 3.2. In relation to any new Primary Medical Services Contract to be entered into, the CCG must, without prejudice to any obligation in Schedule 2, Part 2, paragraph 3 (*Procurement and New Contracts*) and Schedule 2, Part 1, paragraph 2.3:
  - 3.2.1. consider and use the form of Primary Medical Services Contract that will ensure compliance with NHS England's

- obligations under Law including the Public Contracts Regulations 2015/102 and the National Health Service (Procurement, Patient Choice and Competition) (No. 2) Regulations 2013/500 taking into account the persons to whom such Primary Medical Services Contracts may be awarded;
- 3.2.2. provide to NHS England confirmation as required from time to time that it has considered and complied with its obligations under this Agreement and the Law; and
- 3.2.3. for the avoidance of doubt, Schedule 5 (Financial Provisions and Decision Making Limits) deals with the sign off requirements for Primary Medical Services Contracts.

## 4. Approving GP Practice Mergers and Closures

- 4.1. The CCG is responsible for approving GP practice mergers and GP practice closures in the Area.
- 4.2. The CCG must undertake all necessary consultation when taking any decision in relation to GP practice mergers or GP practice closures in the Area, including those set out under section 14Z2 of the NHS Act (duty for public involvement and consultation). The consultation undertaken must be appropriate and proportionate in the circumstances and should include consulting with the Local Medical Committee.
- 4.3. Prior to making any decision in accordance with this paragraph 4 (Approving GP Practice Mergers and Closures), the CCG must be able to clearly demonstrate the grounds for such a decision and must have fully considered any impact on the GP practice's registered population and that of surrounding practices. The CCG must be able to clearly demonstrate that it has considered other options and has entered into dialogue with the GP contractor as to how any closure or merger will be managed.
- 4.4. In making any decisions pursuant to paragraph 4 (Approving GP Practice Mergers and Closures), the CCG shall also take account of its obligations as set out in Schedule 2, part 2, paragraph 3 (Procurement and New Contracts), where applicable.

# 5. Information Sharing with NHS England in relation to the Delegated Functions

- 5.1. This paragraph 5 (Information Sharing with NHS England) is without prejudice to clause 9.4 or any other provision in this Agreement. The CCG must provide NHS England with:
  - 5.1.1. such information relating to individual GP practices in the Area as NHS England may reasonably request, to ensure that NHS England is able to continue to gather national data regarding the performances of GP practices;
  - 5.1.2. such data/data sets as required by NHS England to ensure population of the primary medical services dashboard;
  - 5.1.3. any other data/data sets as required by NHS England; and
  - 5.1.4. the CCG shall procure that providers accurately record and report information so as to allow NHS England and other agencies to discharge their functions.
- 5.2. The CCG must use the NHS England approved primary medical services dashboard, as updated from time to time, for the collection and dissemination of information relating to GP practices.
- 5.3. The CCG must (where appropriate) use the NHS England approved GP exception reporting service (as notified to the CCGs by NHS England from time to time).
- 5.4. The CCG must provide any other information, and in any such form, as NHS England considers necessary and relevant.
- 5.5. NHS England reserves the right to set national standing rules (which may be considered Guidance for the purpose of this Agreement), as needed, to be reviewed annually. NHS England will work with CCGs to agree rules for, without limitation, areas such as the collection of data for national data sets and IT intra-operability. Such national standing rules set from time to time shall be deemed to be part of this Agreement.

# 6. Making Decisions in relation to Management of Poorly Performing GP Practices

- 6.1. The CCG must make decisions in relation to the management of poorly performing GP practices and including, without limitation, decisions and liaison with the CQC where the CQC has reported non-compliance with standards (but excluding any decisions in relation to the performers list).
- 6.2. In accordance with paragraph 6.1 above, the CCG must:
  - 6.2.1. ensure regular and effective collaboration with the CQC to ensure that information on general practice is shared and discussed in an appropriate and timely manner;
  - 6.2.2. ensure that any risks identified are managed and escalated where necessary;
  - 6.2.3. respond to CQC assessments of GP practices where improvement is required;
  - 6.2.4. where a GP practice is placed into special measures, lead a quality summit to ensure the development and monitoring of an appropriate improvement plan (including a communications plan and actions to manage primary care resilience in the locality); and
  - 6.2.5. take appropriate contractual action in response to CQC findings.

### 7. Premises Costs Directions Functions

- 7.1. The CCG must comply with the Premises Costs Directions and will be responsible for making decisions in relation to the Premises Costs Directions Functions.
- 7.2. In particular, but without limiting the generality of paragraph 7.1, the CCG shall make decisions concerning:
  - 7.2.1. applications for new payments under the Premises Costs
    Directions (whether such payments are to be made by way of
    grants or in respect of recurring premises costs); and
  - 7.2.2. revisions to existing payments being made under the Premises Costs Directions.

- 7.3. The CCG must comply with any decision-making limits set out in Schedule 5 (*Financial Provisions and Decision Making Limits*) when taking decisions in relation to the Premises Costs Directions Functions.
- 7.4. The CCG will comply with any guidance issued by the Secretary of State or NHS England in relation to the Premises Costs Directions, including the Principles of Best Practice, and any other Guidance in relation to the Premises Costs Directions.
- 7.5. The CCG must work cooperatively with other CCGs to manage premises and strategic estates planning.
- 7.6. The CCG must liaise where appropriate with NHS Property Services Limited and Community Health Partnerships Limited in relation to the Premises Costs Directions Functions.

### Schedule 2

### Part 2 – Delegated Functions: General Obligations

#### 1. Introduction

1.1. This Part 2 of Schedule 2 (*Delegated Functions*) sets out general provisions regarding the carrying out of the Delegated Functions.

## 2. Planning and reviews

2.1. The CCG is responsible for planning the commissioning of primary medical services.

### 2.2. The role of the CCG includes:

- 2.2.1. carrying out primary medical health needs assessments (to be developed by the CCG) to help determine the needs of the local population in the Area;
- 2.2.2. recommending and implementing changes to meet any unmet primary medical services needs; and
- 2.2.3. undertaking regular reviews of the primary medical health needs of the local population in the Area.

#### 3. Procurement and New Contracts

- 3.1. The CCG will make procurement decisions relevant to the exercise of the Delegated Functions and in accordance with the detailed arrangements regarding procurement set out in the procurement protocol issued and updated by NHS England from time to time.
- 3.2. In discharging its responsibilities set out in clause 6 (Performance of the Delegated Functions) of this Agreement and paragraph 1 of this Schedule 2 (Delegated Functions), the CCG must comply at all times with Law including its obligations set out in the National Health Service (Procurement, Patient Choice and Competition) (No. 2) Regulations 2013/500 and any other relevant statutory provisions. The CCG must have regard to any relevant guidance, particularly Monitor's guidance Substantive guidance on the Procurement, Patient Choice and Competition Regulations

(<a href="https://www.gov.uk/government/uploads/system/uploads/attachment\_dat-a/file/283505/SubstantiveGuidanceDec2013\_0.pdf">https://www.gov.uk/government/uploads/system/uploads/attachment\_dat-a/file/283505/SubstantiveGuidanceDec2013\_0.pdf</a>).

- 3.3. Where the CCG wishes to develop and offer a locally designed contract, it must ensure that it has consulted with its Local Medical Committee in relation to the proposal and that it can demonstrate that the scheme will:
  - 3.3.1. improve outcomes;
  - 3.3.2. reduce inequalities; and
  - 3.3.3. provide value for money.

## 4. Integrated working

- 4.1. The CCG must take an integrated approach to working and co-ordinating with stakeholders including NHS England, Local Professional Networks, local authorities, Healthwatch, acute and community providers, the Local Medical Committee, Public Health England and other stakeholders.
- 4.2. The CCG must work with NHS England and other CCGs to co-ordinate a common approach to the commissioning of primary medical services generally.
- 4.3. The CCG and NHS England will work together to coordinate the exercise of their respective performance management functions.

## 5. Resourcing

5.1. NHS England may, at its discretion provide support or staff to the CCG. NHS England may, when exercising such discretion, take into account, any relevant factors (including without limitation the size of the CCG, the number of Primary Medical Services Contracts held and the need for the Local NHS England Team to continue to deliver the Reserved Functions).

# Schedule 3 Reserved Functions

### 1. Introduction

- 1.1. This Schedule 3 (Reserved Functions) sets out further provision regarding the carrying out of the Reserved Functions.
- 1.2. The CCG will work collaboratively with NHS England and will support and assist NHS England to carry out the Reserved Functions.

## 2. Management of the national performers list

- NHS England will continue to perform its primary medical care functions under the National Health Service (Performers Lists) (England) Regulations 2013.
- 2.2. NHS England's functions in relation to the management of the national performers list include:
  - 2.2.1. considering applications and decision-making in relation to inclusion on the national performers list, inclusion with conditions and refusals;
  - 2.2.2. identifying, managing and supporting primary care performers where concerns arise; and
  - 2.2.3. managing suspension, imposition of conditions and removal from the national performers list.
- 2.3. NHS England may hold local Performance Advisory Group ("PAG") meetings to consider all complaints or concerns that are reported to NHS England in relation to a named performer and NHS England will determine whether an initial investigation is to be carried out.
- 2.4. NHS England may notify the CCG of all relevant PAG meetings at least seven (7) days in advance of such meetings. NHS England may require a representative of the CCG to attend such meetings to discuss any performer concerns and/or quality issues that may impact on individual performer cases.

2.5. The CCG must develop a mechanism to ensure that all complaints regarding any named performer are escalated to the Local NHS England Team for review. The CCG will comply with any Guidance issued by NHS England in relation to the escalation of complaints about a named performer.

## 3. Management of the revalidation and appraisal process

- 3.1. NHS England will continue to perform its functions under the Medical Profession (Responsible Officers) Regulations 2010 (as amended by the Medical Profession (Responsible Officers) (Amendment) Regulations 2013).
- 3.2. All functions in relation to GP appraisal and revalidation will remain the responsibility of NHS England, including:
  - 3.2.1. the funding of GP appraisers;
  - 3.2.2. quality assurance of the GP appraisal process; and
  - 3.2.3. the responsible officer network.
- 3.3. Funding to support the GP appraisal is incorporated within the global sum payment to GP practices.
- 3.4. The CCG must not remove or restrict the payments made to GP practices in respect of GP appraisal.

# 4. Administration of payments and related performers list management activities

- 4.1. NHS England reserves its functions in relation to the administration of payments to individual performers and related performers list management activities under the National Health Service (Performers Lists) (England) Regulations 2013 and other relevant legislation.
- 4.2. NHS England may continue to pay GPs who are suspended from the national performers list under the Secretary of State's Determination: Payments to Medical Practitioners Suspended from the Medical Performers List (1 April 2013).

4.3. For the avoidance of doubt, the CCG is responsible for any ad hoc or discretionary payments to GP practices (including those under section 96 of the NHS Act) in accordance with clause 6.2.1.4 and Schedule 2 (Delegated Functions) Part 1 paragraphs 2.13 and 2.14 of this Agreement, including where such payments may be considered a consequence of actions taken under the National Health Service (Performers Lists) (England) Regulations 2013.

### 5. Section 7A Functions

- 5.1. In accordance with clauses 13.17 to 13.20, NHS England retains the Section 7A Functions and will be responsible for taking decisions in relation to the Section 7A Functions.
- 5.2. In accordance with clauses 13.17 to 13.20, the CCG will provide certain management and/or administrative services to NHS England in relation to the Section 7A Functions.

## 6. Capital Expenditure Functions

6.1. In accordance with clauses 13.13 to 13.16, NHS England retains the Capital Expenditure Functions and will be responsible for taking decisions in relation to the Capital Expenditure Functions.

## 7. Functions in relation to complaints management

- 7.1. NHS England retains its functions in relation to complaints management and will be responsible for taking decisions in relation to the management of complaints. Such complaints include (but are not limited to):
  - 7.1.1. complaints about GP practices and individual named performers;
  - 7.1.2. controlled drugs; and
  - 7.1.3. whistleblowing in relation to a GP practice or individual performer.
- 7.2. The CCG must immediately notify the Local NHS England Team of all complaints received by or notified to the CCG and must send to the Local NHS England Team copies of any relevant correspondence.

- 7.3. The CCG must co-operate fully with NHS England in relation to any complaint and any response to such complaint.
- 7.4. In accordance with clauses 13.21 to 13.23, NHS England may ask the CCG to provide certain management and/or administrative services to NHS England (from a date to be notified by NHS England to the CCG) in relation to the handling and consideration of complaints.

# 8. Such other ancillary activities that are necessary in order to exercise the Reserved Functions

- 8.1. NHS England will carry out such other ancillary activities that are necessary in order for NHS England to exercise the Reserved Functions.
- 8.2. NHS England will continue to comply with its obligations under the Controlled Drugs (Supervision of Management and Use) Regulations 2013.
- 8.3. The CCG must assist NHS England's controlled drug accountable officer ("CDAO") to carry out its functions under the Controlled Drugs (Supervision of Management and Use) Regulations 2013.
- 8.4. The CCG must nominate a relevant senior individual within the CCG (the "CCG CD Lead") to liaise with and assist NHS England to carry out its functions under the Controlled Drugs (Supervision of Management and Use) Regulations 2013.
- 8.5. The CCG CD Lead must, in relation to the Delegated Functions:
  - 8.5.1. on request provide NHS England's CDAO with all reasonable assistance in any investigation involving primary medical care services;
  - 8.5.2. report all complaints involving controlled drugs to NHS England's CDAO;
  - 8.5.3. report all incidents or other concerns involving the safe use and management of controlled drugs to NHS England's CDAO;
  - 8.5.4. analyse the controlled drug prescribing data available; and

8.5.5. on request supply (or ensure organisations from whom the CCG commissions services involving the regular use of controlled drugs supply) periodic self-declaration and/or self-assessments to NHS England's CDAO.

# Schedule 4 Further Information Sharing Provisions

#### 1. Introduction

- 1.1. The purpose of this Schedule 4 (Further Information Sharing Provisions) and the associated Personal Data Agreement is to set out the scope for the secure and confidential sharing of information between the Parties on a **Need To Know** basis between individual Personnel in order to enable the Parties to exercise their primary medical care commissioning functions in accordance with the law. This Schedule and the associated Personal Data Agreement is designed to:
  - 1.1.1. inform about the reasons why Relevant Information may need to be shared and how this will be managed and controlled by the organisations involved;
  - 1.1.2. describe the purposes for which the Parties have agreed to share Relevant Information;
  - 1.1.3. set out the lawful basis for the sharing of information between the Parties, and the principles that underpin the exchange of Relevant Information;
  - 1.1.4. describe roles and structures to support the exchange of Relevant Information between the Parties;
  - apply to the sharing of Relevant Information relating to GPs where necessary;
  - 1.1.6. apply to the sharing of Relevant Information whatever the medium in which it is held and however it is transmitted;
  - 1.1.7. ensure that Data Subjects are, where appropriate, informed of the reasons why Personal Data about them may need to be shared and how this sharing will be managed;
  - 1.1.8. apply to the activities of the Parties' Personnel; and
  - 1.1.9. describe how complaints relating to Personal Data sharing between the Parties will be investigated and resolved, and how the information sharing will be monitored and reviewed.

# 2. Purpose

- 2.1. The Specified Purpose(s) of the data sharing initiative is to facilitate the exercise of the CCG's Delegated Functions and NHS England's Reserved Functions:
  - 2.1.1. the management of the primary medical service performers' list in accordance with section 91 of the NHS Act;
  - 2.1.2. management of GP revalidation and appraisal;
  - 2.1.3. administration of payments and related performers list management activities;
  - 2.1.4. planning and delivering the provision of appropriate care services:
  - 2.1.5. improving the health of the local population;
  - 2.1.6. performance management of GP providers;
  - 2.1.7. investigating and responding to incidents and complaints; and
  - 2.1.8. reducing risk to individuals, service providers and the public as a whole.
- 2.2. Specific and detailed purposes are set out in the Personal Data Agreement appended to this Schedule.

## 3. Benefits of information sharing

3.1. The benefits of sharing information are the achievement of the Specified Purposes set out above, with benefits for service users and other stakeholders in terms of the improved local delivery of primary healthcare services.

## 4. Legal basis for Sharing

- 4.1. Each Party shall comply with all relevant Information Law requirements and good practice in relation to the processing of Relevant Information shared further to this Agreement.
- 4.2. The Parties shall identify the lawful basis for sharing Relevant Information for each purpose and data flow, and document these in the attached Personal Data Agreement.

### 5. Relevant Information to be shared

5.1. The Relevant Information to be shared is set out in the attached Personal Data Agreement.

### 6. Restrictions on use of the Shared Information

- 6.1. Each Party shall only process the Relevant Information as is necessary to achieve the Specified Purpose, and, in particular, shall not use or process Relevant Information for any other purpose unless agreed in writing by the Data Controller that released the information to the other. There shall be no other use or onward transmission of the Relevant Information to any third party without a lawful basis first being determined, and the originating Data Controller being notified.
- 6.2. Access to, and processing of, the Relevant Information provided by a Party must be the minimum necessary to achieve the Specified Purpose. Information and Sensitive Personal Data will be handled at all times on a restricted basis, in compliance with Information Law requirements, and Personnel should only have access to Personal Data on a justifiable **Need to Know** basis for the purpose of performing their duties in connection with the services they are there to deliver. The **Need to Know** requirement means that the Data Controllers' Personnel will only have access to Personal Data or Sensitive Personal Data if it is lawful for such Personnel to have access to such data for the Specified Purpose and the function they are required to fulfil at that particular time, in relation to the Specified Purpose, cannot be achieved without access to the Personal Data or Sensitive Personal Data specified.
- 6.3. Having this Agreement in place does not give licence for unrestricted access to data that the other Data Controller may hold. It lays the parameters for the safe and secure sharing and processing of information for a justifiable **Need to Know** purpose.
- 6.4. Neither Party shall subcontract any processing of the Relevant Information without the prior written consent of the other Party. Where a Party subcontracts its obligations, it shall do so only by way of a written agreement with the sub-contractor which imposes the same

- obligations as are imposed on the Data Controllers under this Agreement.
- 6.5. Neither Party shall cause or allow Data to be transferred to any territory outside the European Economic Area without the prior written permission of the responsible Data Controller.
- 6.6. Any particular restrictions on use of certain Relevant Information are included in the attached Personal Data Agreement.

## 7. Ensuring fairness to the Data Subject

- 7.1. In addition to having a lawful basis for sharing information, the DPA generally requires that the sharing must be fair. In order to achieve fairness to the Data Subjects, the Parties will put in place the following arrangements:
  - 7.1.1. amendment of internal guidance to improve awareness and understanding among Personnel;
  - 7.1.2. amendment of privacy notices and policies; and
  - 7.1.3. consideration given to further activities to promote public understanding where appropriate.
- 7.2. Each Party shall procure that its notification to the Information Commissioner's Office reflects the flows of information under this Agreement.
- 7.3. Further provision in relation to specific data flows is included in the attached Personal Data Agreement.

#### 8. Governance: Personnel

8.1. Each Party must take reasonable steps to ensure the suitability, reliability, training and competence, of any Personnel who have access to the Personal Data (and Sensitive Personal Data) including reasonable background checks and evidence of completeness should be available on request by each Party.

- 8.2. The Parties agree to treat all Relevant Information as confidential and imparted in confidence and must safeguard it accordingly. Where the Personnel are not healthcare professionals (for the purposes of the DPA) the employing Parties must procure that its Personnel operate under a duty of confidentiality which is equivalent to that which would arise if that person were a health professional.
- 8.3. Each Party shall ensure that all Personnel required to access the Personal Data (including Sensitive Personal Data) are informed of the confidential nature of the Personal Data and each Party shall include appropriate confidentiality clauses in employment/service contracts of all Personnel that have any access whatsoever to the Relevant Information, including details of sanctions against any employee acting in a deliberate or reckless manner that may breach the confidentiality or the non-disclosure provisions of Information Law requirements, or causes damage to or loss of the Relevant Information.
- 8.4. Each Party shall provide evidence (further to any reasonable request) that all Personnel that have any access to the Relevant Information whatsoever are adequately and appropriately trained to comply with their responsibilities under Information Law and this Agreement.
- 8.5. Each Party shall ensure that:
  - 8.5.1. only those employees involved in delivery of the Agreement use or have access to the Relevant Information; and
  - 8.5.2. that such access is granted on a strict **Need to Know** basis and shall implement appropriate access controls to ensure this requirement is satisfied and audited. Evidence of audit should be made freely available on request by the originating Data Controller. These access controls are set out in the attached Personal Data Agreement; and
  - 8.5.3. specific limitations on the Personnel who may have access to the Information are set out in the attached Personal Data Agreement.
- 9. Governance: Protection of Personal Data

- 9.1. At all times, the Parties shall have regard to the requirements of Information Law and the rights of Data Subjects.
- 9.2. Wherever possible (in descending order of preference), only anonymised information, or strongly or weakly pseudonymised information will be shared and processed by Parties, without the need to share easily identifiable Personal Data. The Parties shall cooperate in exploring alternative strategies to avoid the use of Personal Data in order to achieve the Specified Purpose. However, it is accepted that some Relevant Information shared further to this Agreement may be Personal Data/Sensitive Personal Data.
- 9.3. Processing of any Personal Data or Sensitive Personal Data shall be to the minimum extent necessary to achieve the Specified Purpose, and on a **Need to Know** basis. If either Party:
  - 9.3.1. becomes aware of any unauthorised or unlawful processing of any Relevant Information or that any Relevant Information is lost or destroyed or has become damaged, corrupted or unusable; or
  - 9.3.2. becomes aware of any security breach,

in respect of the Relevant Information it shall promptly notify the other Party. The Parties shall fully cooperate with one another to remedy the issue as soon as reasonably practicable.

- 9.4. In processing any Relevant Information further to this Agreement, each Party shall:
  - 9.4.1. process the Personal Data (including Sensitive Personal Data) only in accordance with the terms of this Agreement and otherwise only in accordance with written instructions from the originating Data Controller in respect of its Relevant Information:
  - 9.4.2. process the Personal Data (including Sensitive Personal Data) only to the extent as is necessary for the provision of the Specified Purpose or as is required by law or any regulatory body;
  - 9.4.3. process the Personal Data (including Sensitive Personal Data) only in accordance with Information Law requirements and shall not perform its obligations under this Agreement in

- such a way as to cause any other Data Controller to breach any of their applicable obligations under Information Law; and
- 9.4.4. process the Personal Data in accordance with the eight data protection principles (the "Data Protection Principles") in Schedule 1 to the DPA.
- 9.5. Each Party shall act generally in accordance with the Seventh Data Protection Principle, and in particular shall implement and maintain appropriate technical and organisational measures to protect the Personal Data (and Sensitive Personal Data) against unauthorised or unlawful processing and against accidental loss, destruction, damage, alteration or disclosure. These measures shall be appropriate to the harm which might result from any unauthorised or unlawful processing, accidental loss, destruction or damage to the Personal Data (and Sensitive Personal Data) and having regard to the nature of the Personal Data (and Sensitive Personal Data) which is to be protected. In particular, each Data Controller shall:
  - 9.5.1. ensure that only Personnel authorised under this Agreement have access to the Personal Data (and Sensitive Personal Data);
  - 9.5.2. ensure that the Relevant Information is kept secure and in an encrypted form, and shall use all reasonable security practices and systems applicable to the use of the Relevant Information to prevent and to take prompt and proper remedial action against, unauthorised access, copying, modification, storage, reproduction, display or distribution, of the Relevant Information;
  - 9.5.3. obtain prior written consent from the originating Data Controller in order to transfer the Relevant Information to any third party;
  - 9.5.4. permit the other Data Controllers or the Data Controllers' representatives (subject to reasonable and appropriate confidentiality undertakings), to inspect and audit the data processing activities carried out further to this Agreement (and/or those of its agents, successors or assigns) and comply with all reasonable requests or directions to enable the Data Controllers to verify and/or procure that the other

- Data Controller is in full compliance with its obligations under this Agreement; and
- 9.5.5. if requested, provide a written description of the technical and organisational methods and security measures employed in processing Personal Data.
- 9.5.6. Specific requirements as to information security are set out in the Schedule.
- 9.5.7. Each Party shall use best endeavours to achieve and adhere to the requirements of the NHS Information Governance Toolkit, particularly in relation to Confidentiality and Data Protection Assurance, Information Security Assurance and Clinical Information Assurance.
- 9.5.8. The Parties' Single Points of Contact ("SPoC") set out in paragraph 14 (Governance: Single Points of Contact) below will be the persons who, in the first instance, will have oversight of third party security measures.

### 10. Governance: Transmission of Information between the Parties

- 10.1. This paragraph supplements paragraph 9 (Governance: Protection of Personal Data) of this Schedule.
- 10.2. Transfer of Personal Data between the Parties shall be done through secure mechanisms including use of the N3 network, encryption, and approved secure (NHS.net / gcsx) email.
- 10.3. Faxes shall only be used to transmit Personal Data in an emergency.
- 10.4. Wherever possible, Personal Data should be transmitted in pseudonymised form, with only reference to the NHS number in 'clear' transmissions. Where there are significant consequences for the care of the patient, then additional data items, such as the postcode, date of birth and/or other identifiers should also be transmitted, in accordance with good information governance and clinical safety practice, so as to ensure that the correct patient record / data is identified.
- 10.5. Any other special measures relating to security of transfer are specified in the attached Personal Data Agreement.

- 10.6. Each Party shall keep an audit log of Relevant Information transmitted and received in the course of this Agreement.
- 10.7. The Parties' Single Point of Contact notified pursuant to paragraph 14 (Governance: Single Points of Contact) will be the persons who, in the first instance, will have oversight of the transmission of information between the Parties.

## 11. Governance: Quality of Information

- 11.1. The Parties will take steps to ensure the quality of the Relevant Information and to comply with the fourth Data Protection Principle.
- 11.2. Special measures relating to ensuring quality are set out in the attached Personal Data Agreement.

## 12. Governance: Retention and Disposal of Shared Information

- 12.1. The non-originating Party shall securely destroy or return the Relevant Information once the need to use it has passed or, if later, upon the termination of this Agreement, howsoever determined. Where Relevant Information is held electronically the Relevant Information will be deleted and formal notice of the deletion sent to the Party that shared the Relevant Information. Once paper information is no longer required, paper records will be securely destroyed or securely returned to the Party they came from.
- 12.2. Each Party shall provide an explanation of the processes used to securely destroy or return the information, or verify such destruction or return, if requested by the other Party and shall comply with any request of the Data Controllers to dispose of data in accordance with specified standards or criteria.
- 12.3. If either Party is required by any law, regulation, or government or regulatory body to retain any documents or materials that it would otherwise be required to return or destroy under this paragraph 12 (Governance: Retention and Disposal of Shared Information), it shall

- notify the other Party in writing of that retention, giving details of the documents or materials that it must retain.
- 12.4. Retention of any data shall comply with the Fifth Data Protection Principle and with all good practice including the Records Management NHS Code of Practice, as updated or amended from time to time.
- 12.5. Any special retention periods are set out in attached Personal Data Agreement.
- 12.6. Each Party shall ensure that Relevant Information held in paper form is held in secure files, and, when it is no-longer needed, destroyed using a cross cut shredder or subcontracted to a confidential waste company that complies with European Standard EN15713.
- 12.7. Each Party shall ensure that, when no longer required, electronic storage media used to hold or process Personal Data are destroyed or overwritten to current policy requirements.
- 12.8. Electronic records will be considered for deletion once the relevant retention period has ended.
- 12.9. In the event of any bad or unusable sectors of electronic storage media that cannot be overwritten, the Party shall ensure complete and irretrievable destruction of the media itself in accordance with policy requirements.

### 13. Governance: Complaints and Access to Personal Data

- 13.1. Each Party shall assist the other in responding to any request made under Information Law made by persons who wish to access copies of information held about them ("Subject Access Requests").
- 13.2. Complaints about information sharing shall be routed through each Party's own complaints procedure but reported to the Single Points of Contact set out in paragraph 14 (Governance: Single Points of Contact) below.

- 13.3. The Parties shall use all reasonable endeavours to work together to resolve any dispute or complaint arising under this Agreement or any data processing carried out further to it.
- 13.4. Basic details of the Agreement shall be included in the appropriate log under each Party's Publication Scheme.

## 14. Governance: Single Points of Contact

14.1. The Parties each shall appoint a single point of contact to whom all queries relating to the particular information sharing should be directed in the first instance. Details of the single points of contact shall be set out in the attached Personal Data Agreement.

## 15. Monitoring and review

15.1. The Parties shall monitor and review on an ongoing basis the sharing of Relevant Information to ensure compliance with Information Law and best practice. Specific monitoring requirements are set out in the attached Personal Data Agreement.

### **Template Personal Data Agreement**

Data flow: [Description]

# Description of information flow and single points of contact for parties involved

Originating Data Controller	[Insert:]			
Contact details for single point of contact for Originating Data Controller	Name of point of contact	Title	Contact (email)	Contact (phone)
Recipient Data Controller	[Insert:]			1
Contact details for single point of contact of Recipient Data Controller	Name of point of contact	Title	Contact (email)	Contact (phone)

### Description of information to be shared

Comprehensive	[Insert:]
description of	
Relevant	
Information to be	
shared	
Anonymised / not	Yes / No
information about	
individual	
persons	
Strongly	Yes / No
pseudonyimsed	
Weakly	Yes / No
pseudonymised	
Person -	Yes / No
identifiable data	
Justification for	[Insert or N/A:]

the level of
identifiability
required

### Legal basis for disclosure and use

DPA Schedule 2	[Insert or N/A:]	
condition/s		
DPA Schedule 3	[Insert or N/A:]	
condition/s		
Confidentiality	Explicit consent	Yes / No
		[If yes, how documented?:]
	Implied Consent	Yes / No
		[If yes, how have you implied
		consent?:]
	Statutory required/permitted	[Insert statutory basis:]
	disclosure	
	Public interest disclosure	[Insert how the public interest
		favours use/disclosure of the
		information:]
	Other legal basis	[Insert:]
s. 13Z3 / 14Z23	S. 13Z3 condition(s) to	[Insert:]
NHS Act 2006	permit disclosure	
justification	S. 14Z23 condition(s) to	[Insert:]
	permit disclosure	
Other specific		
legal		

considerations	
Danielia di anno anno anno	

### Restrictions on use of information

[Insert:]			

### Governance arrangements

Specific measures to	[Insert:]
ensure fairness to the	
Data Subject, including	
privacy impact	
assessments undertaken	
Access controls on use	[Insert:]
of information	
Specific limitations on	[Insert:]
Personnel who may	
access information	
Other specific security	[Insert:]
requirements	
(transmission)	
Other specific security	[Insert:]
requirements (general)	
Specific requirements as	[Insert:]
to ensuring quality of	
information	
Specific requirements for	[Insert:]
retention and destruction	
of information	
Specific monitoring and	[Insert:]
review arrangements	

## Schedule 5 Financial Provisions and Decision Making Limits

### Financial Limits and Approvals

- 1. The CCG shall ensure that any decisions in respect of the Delegated Functions and which exceed the financial limits set out below are only taken:
  - 1.1. by the following persons and/or individuals set out in column 2 of Table1 below; and
  - 1.2. following the approval of NHS England (if any) as set out in column 3 of the Table 1 below.
- 2. NHS England may, from time to time, update Table 1 by sending a notice to the CCG of amendments to Table 1.

Table 1 – Financial Limits					
Decision	Person/Individual	NHS England Approval			
General					
Taking any step or action in relation to the settlement of a Claim, where the value of the settlement exceeds £100,000	CCG Accountable Officer or Chief Finance Officer or Chair	NHS England Head of Legal Services and Local NHS England Team Director or Director of Finance			
Any matter in relation to the Delegated Functions which is novel, contentious or repercussive	CCG Accountable Officer or Chief Finance Officer or Chair	Local NHS England Team Director or Director of Finance or NHS England Region Director or Director of Finance or NHS England Chief Executive or Chief Financial Officer			
Revenue Contracts					
The entering into of any Primary Medical Services Contract which has or is capable of having a term which exceeds five (5) years	CCG Accountable Officer or Chief Finance Officer or Chair	Local NHS England Team Director or Director of Finance			

### Capital

Note: As at the date of this Agreement, the CCG will not have delegated or directed responsibility for decisions in relation to Capital expenditure (and these decisions are retained by NHS England) but the CCG may be required to carry out certain administrative services in relation to Capital expenditure under clause 13 (Financial Provisions and Liability).

### Schedule 6

### **Template Variation Agreement**

Variation Reference:		[insert reference]
		[insert party] [Note – only NHS England may propose National Variations]
Date	of Proposal:	[insert date]
Date	of Variation Agreeme	ent: [insert date]
-	alised words and phrasem in the Agreement re	ses in this Variation Agreement have the meanings given ferred to above.
1.	The Parties have agre	ed the [National] Variation summarised below:
2.		on is reflected in the attached Schedule and the Parties lent is varied accordingly.
3.	The Variation takes ef	fect on [insert date].
	ITNESS OF WHICH t late(s) shown below	he Parties have signed this Variation Agreement on
Sign	ed by	NHS England [Insert name of Authorised Signatory] [for and on behalf of] [ ]
Sign	ed by	[Insert name] Clinical Commissioning Group [Insert name of Authorised Signatory][for and on behalf of] [ ]

### Schedule to Variation Agreement

[Insert details of variation]

### Schedule 7 Local Terms

"There are no Local Terms" in this Schedule 7

# Schedule 8 Assignment of NHS England Staff to the CCG

### 1. Introduction

- 1.1. The purpose of this Schedule 8 (Assignment of NHS England Staff to the CCG) is to give clarity to the CCG and NHS England, in circumstances where NHS England staff are assigned to the CCG under Model 1 of the Staffing Models.
- 1.2. In accordance with clause 18 of this Agreement, the Parties have agreed that the CCG may only engage staff to undertake the Delegated Functions under one of the three Staffing Models referred to in that clause.
- 1.3. The Parties agree and acknowledge that until such time as the CCG's preferred Staffing Model takes effect, the engagement of staff to undertake the Delegated Functions shall be in accordance with the terms of this Schedule 8 (Assignment of NHS England Staff to the CCG) (the "Arrangements").

#### 2. Duration

2.1. The Arrangements shall commence on the date of this Agreement and shall continue until the date on which the Parties agree which of the Staffing Models (set out at clauses 18.1.1 to 18.1.3) will be adopted by the CCG and the date on which such Staffing Model shall take effect.

### 3. Services

- 3.1. NHS England agrees to make available the Assigned Staff to the CCG to perform administrative and management support services together with such other services specified in Schedule 7 (Local Terms) (the "Services") so as to facilitate the CCG in undertaking the Delegated Functions pursuant to the terms of this Agreement.
- 3.2. NHS England shall take all reasonable steps to ensure that the Assigned Staff shall:

- 3.2.1. faithfully and diligently perform duties and exercise such powers as may from time to time be reasonably assigned to or vested in them; and
- 3.2.2. perform all duties assigned to them pursuant to this Schedule 8 (Assignment of NHS England Staff to the CCG).
- 3.3. The CCG shall notify NHS England if the CCG becomes aware of any act or omission by any Assigned Staff which may have a material adverse impact on the provision of the Services or constitute a material breach of the terms and conditions of employment of the Assigned Staff.
- 3.4. NHS England shall be released from its obligations to make the Assigned Staff available for the purposes of this Schedule 8 (Assignment of NHS England Staff to the CCG) whilst the Assigned Staff are absent:
  - 3.4.1. by reason of industrial action taken in contemplation of a trade dispute;
  - 3.4.2. as a result of the suspension or exclusion of employment or secondment of any Assigned Staff by NHS England;
  - 3.4.3. in accordance with the Assigned Staff's respective terms and conditions of employment and policies, including, but not limited to, by reason of training, holidays, sickness, injury, trade union duties, paternity leave or maternity or where absence is permitted by Law;
  - 3.4.4. if making the Assigned Staff available would breach or contravene any Law;
  - 3.4.5. as a result of the cessation of employment of any individual Assigned Staff; and/or
  - 3.4.6. at such other times as may be agreed between NHS England and the CCG.

### 4. Employment of the Assigned Staff

4.1. NHS England shall employ the Assigned Staff and shall be responsible for the employment of the Assigned Staff at all times on whatever terms and conditions as NHS England and the Assigned Staff may agree from time to time.

- 4.2. NHS England shall pay the Assigned Staff their salaries and benefits and make any deductions for income tax liability and national insurance or similar contributions it is required to make from the Assigned Staff's salaries and other payments.
- 4.3. The Assigned Staff shall carry out the Services from NHS England's places of work and may be required to attend the offices of the CCG from time to time in the course of carrying out the Services. Nothing in this Schedule 8 (Assignment of NHS England Staff to the CCG) shall be construed or have effect as constituting any relationship of employer and employee between the CCG and the Assigned Staff.
- 4.4. NHS England shall not, and shall procure that the Assigned Staff shall not, hold themselves out as employees of the CCG.

### 5. Management

- 5.1. NHS England shall have day-to-day control of the activities of the Assigned Staff and deal with any management issues concerning the Assigned Staff including, without limitation, performance appraisal, discipline and leave requests.
- 5.2. The CCG agrees to provide all such assistance and co-operation that NHS England may reasonably request from time to time to resolve grievances raised by Assigned Staff and to deal with any disciplinary allegations made against Assigned Staff arising out of or in connection with the provision of the Services which shall include, without limitation, supplying NHS England with all information and the provision of access to all documentation and personnel as NHS England requires for the purposes of considering and dealing with such issues and participating promptly in any action which may be necessary.

#### 6. Conduct of Claims

6.1. If the CCG becomes aware of any matter that may give rise to a claim by or against a member of Assigned Staff, notice of that fact shall be given as soon as possible to NHS England. NHS England and the CCG shall co-operate in relation to the investigation and resolution of any such claims or potential claims. 6.2. No admission of liability shall be made by or on behalf of the CCG and any such claim shall not be compromised, disposed of or settled without the consent of NHS England.

### 7. Confidential Information and Property

- 7.1. For the avoidance of doubt, this paragraph 8 (Confidential Information and Property) is without prejudice to any other provision of this Agreement in relation to confidential information.
- 7.2. It is acknowledged that to enable the Assigned Staff to provide the Services, the Parties may share information of a highly confidential nature being information or material which is the property of NHS England or the CCG or which NHS England or the CCG are obliged to hold confidential including, without limitation, all official secrets, information relating to the working of any project carried on or used by the relevant Party, research projects, strategy documents, tenders, financial information, reports, ideas and know-how, employee confidential information and patient confidential information and any proprietary party information (any and all of the foregoing being "Confidential Information").
- 7.3. The Parties agree to adopt all such procedures as the other party may reasonably require and to keep confidential all Confidential Information and that the Parties shall not (save as required by law) disclose the Confidential Information in whole or in part to anyone and agree not to disclose the Confidential Information other than in connection with the provision of the Services.
- 7.4. The obligations under this Agreement apply to all and any Confidential Information whether the Confidential Information was in or comes into the possession of the relevant person prior to or following this Agreement and such obligations shall continue at all times following the termination of the Arrangements but shall cease to apply to information which may come into the public domain otherwise than through unauthorised disclosure by NHS England or the CCG, as the case may be.

### 8. Intellectual Property

8.1. All Intellectual Property (meaning any invention, idea, improvement, discovery, development, innovation, patent, writing, concept design made, process information discovered, copyright work, trademark, trade name and/or domain name) made, written, designed, discovered or originated by the Assigned Staff shall be the property of NHS England to the fullest extent permitted by law and NHS England shall be the absolute beneficial owner of the copyright in any such Intellectual Property.



# Primary Care Commissioning Committee April 2019

Agenda Item: 19.06	Author of the Paper:
Report date: 4 <sup>th</sup> April 2019	Angela Price Primary Care Programme Lead angela.price@southseftonccg.nhs.uk Tel: 01513178379
Title: Transition Plan to Delegation	
Summary/Key Issues: This plan highlights the training and work under with subject matter expertise to work with the Coprocedures were shared to enable a smooth tracks a result of this process we have produced a responsibilities of individual organisations.	ansition.
Recommendation The Primary Care Commissioning Committee i the transition plan.	s asked to note the detail in  Note  Approve  Ratify

### Links to Corporate Objectives (x those that apply) To focus on the identification of QIPP (Quality, Improvement, Productivity & Prevention) schemes and the implementation and delivery of these to achieve the CCG QIPP target Х and to support delivery of financial recovery. To progress Shaping Sefton as the strategic plan for the CCG, in line with the NHSE planning requirements set out in the "Five Year Forward View", underpinned by Χ transformation through the agreed strategic blueprints and programmes and as part of the NHS Cheshire and Merseyside Healthcare Partnership. To ensure that the CCG maintains and manages performance & quality across the Х mandated constitutional measures. To support Primary Care Development through the development of an enhanced model of care and supporting estates strategy, underpinned by a complementary primary care Х quality contract. To advance integration of in-hospital and community services in support of the CCG Х locality model of care. To advance the integration of Health and Social Care through collaborative working with Sefton Metropolitan Borough Council, supported by the Health and Wellbeing Board.

Process	Yes	No	N/A	Comments/Detail (x those that apply)
Patient and Public			Х	
Engagement				
Clinical Engagement			Х	
Equality Impact			Х	
Assessment				
Legal Advice Sought			Х	
Resource Implications	Х			
Considered				
Locality Engagement		Х		
Presented to other		Χ		
Committees				

Lin	iks to National Outcomes Framework (x those that apply)
Х	Preventing people from dying prematurely
Х	Enhancing quality of life for people with long-term conditions
Х	Helping people to recover from episodes of ill health or following injury
Х	Ensuring that people have a positive experience of care
Х	Treating and caring for people in a safe environment and protecting them from avoidable
	harm



### **Primary Care Commissioning Committee April 2019** Agenda Item: 19.06 **Author of the Paper:** Angela Price Primary Care Programme Lead Report date: 4th April 2019 angela.price@southseftonccg.nhs.uk Tel: 01513178379 **Title:** Transition Plan to Delegation **Summary/Key Issues:** This plan highlights the training and work undertaken prior to delegation. NHSE identified staff with subject matter expertise to work with the CCG to ensure that processes, documents and procedures were shared to enable a smooth transition. As a result of this process we have produced a guide for practices which clarifies the responsibilities of individual organisations. Recommendation Note The Primary Care Commissioning Committee is asked to note the detail in **Approve** the transition plan. Ratify

Link	Links to Corporate Objectives (x those that apply)							
x	To focus on the identification of QIPP (Quality, Improvement, Productivity & Prevention) schemes and the implementation and delivery of these to achieve the CCG QIPP target and to support delivery of financial recovery.							
×	To progress Shaping Sefton as the strategic plan for the CCG, in line with the NHSE planning requirements set out in the "Five Year Forward View", underpinned by transformation through the agreed strategic blueprints and programmes and as part of the NHS Cheshire and Merseyside Healthcare Partnership.							
х	To ensure that the CCG maintains and manages performance & quality across the mandated constitutional measures.							
x	To support Primary Care Development through the development of an enhanced model of care and supporting estates strategy, underpinned by a complementary primary care quality contract.							
x	To advance integration of in-hospital and community services in support of the CCG locality model of care.							
Х	To advance the integration of Health and Social Care through collaborative working with							

Sefton Metropolitan Borough Council, supported by the Health and Wellbeing Board.

Process	Yes	No	N/A	Comments/Detail (x those that apply)
Patient and Public Engagement			x	
Clinical Engagement			Х	
Equality Impact Assessment			х	
Legal Advice Sought			Х	
Resource Implications Considered	Х			
Locality Engagement		х		
Presented to other Committees		Х		

Link	Links to National Outcomes Framework (x those that apply)							
Х	Preventing people from dying prematurely							
Х	Enhancing quality of life for people with long-term conditions							
Х	Helping people to recover from episodes of ill health or following injury							
Х	Ensuring that people have a positive experience of care							
х	Treating and caring for people in a safe environment and protecting them from avoidable harm							

### NHS South Sefton and Southport and Formby CCG Transitional Plan

	Task		Elements & Milestones	Frequency	Training Delivered	Trainer	Comments	
1	Contract Management	1.1	Issuing/amending/variations of contracts for GMS/PMS/APMS	Monthly	Angela Price, Jane Elliott, Clare Touhey, Stephen Bennett, Lesley McKinnell	Diane Lewis, NHSE	C371BF76.pub  X:\Primary Care\ Delegation\ Contract Variations\	
		2.1	Invite practices to participate in DESs	Annual	Clare Touhey, Jane Elliott	Diane Lewis, NHSE		
		2.2	Collate participation responses	Annual	Clare Touhey, Jane Elliott	Diane Lewis, NHSE	]	
	Commission all medical enhanced services as directed Directed Enhanced Services (DES)  Learning Disabilities Health Checks Scheme	2.3	Inform finance team of participation to inform budget setting	Annual	Rob Smith/ Leah Robinson CQRS training (NHSE function to pay)	Diane Lewis, NHSE	W	
	Minor Surgery • Extended Hours scheme	2.4	Offer services on CQRS	Adhoc	Clare Touhey, Jane Elliott	Diane Lewis, NHSE	Delegated GMS   (Part 12) 2018-19	
	Primary Care Network  Out of Area Registration	2.5	Advise practices of claims process	Annual	Practice Claims Guide developed (NHSE/CCG)	Diane Lewis, NHSE	FS V2 docx	
	Special Allocation Scheme  • Hepatitis B (new-born babies) - PHE	2.6	Review and approve achievement on CQRS	Monthly	Clare Touhey, Jane Elliott	Diane Lewis, NHSE		
2	Meningococcal C (Men C) fresher's - PHE     MMR (aged 16 and over) - PHE     Pertussis - PHE	2.7	Process quarterly/annual claims	Quarterly	Primary Care Team, spreadsheet of payments to NHSE)	Diane Lewis, NHSE	https://www.england.nhs.uk/gp/gpfv/investment/gp-contract/	
	Rotavirus for infants - PHE Childhood seasonal influenza - PHE Pneumococcal Adult & At Risk - PHE Seasonal Influenza - PHE Shingles (routine aged 70) - PHE	2.8	Manage practice queries	Adhoc	Primary Care Team/ Rob Smith Tamara Ramirez- Roman - Practice queries to CCG NHS net account	Diane Lewis, NHSE		
	Shingles (catch up) - PHE     Pneumococcal Child (PCV) - PHE	2.9	Provide CQRS reports as required	Adhoc	Clare Touhey, Jane Elliott	Diane Lewis, NHSE		
3	CQRS	3.1	Setting up of new users and amendment to access to practices as requested by users.	Adhoc	Clare Touhey, Jane Elliott	Diane Lewis, NHSE		
			Amendment of data entered by practices as requested	Adhoc	Clare Touhey, Jane Elliott	Diane Lewis, NHSE	CQRS_User_Guide. pdf	
4	Complete Year End Processes for QOF	4.2	Amendment of data entered by practices as requested	Adhoc	Clare Touhey, Jane Elliott	Diane Lewis, NHSE		
		5.1	To follow application process	Adhoc	Jane Elliott	Diane Lewis, NHSE	W W	
5	GP List Closures Applications	5.2	Approve/reject applications at medical meetings	Adhoc	Primary Care Operational Group	Diane Lewis, NHSE	18 03 12 List Application for Closure Practice List Closure Application To Extend Closure Action Plan.docx	
		5.3	Inform Practice, CCG and PCS of decision	Adhoc	Jane Elliott	Diane Lewis, NHSE		
6	Practice Boundary changes	6.1	Application form, process, report and decision	Adhoc	Clare Touhey, Jane Elliott	Diane Lewis, NHSE	Application to change the Landrice area docx	
7	Easter/Christmas opening hours	7.1	Opening hours of GP practices during Christmas to be recorded by Primary Care Team	Bi Annual	Primary Care Team	Diane Lewis, NHSE	Copy of document	
8	Proactively performance manage contracts	8.1	GP Investigations / Practice issues	Adhoc	Primary Care Team	Use Policy Guidance Manual	https://www.england.nhs.uk/publication/primary-medical-care-policy-	
		9.1	Dealing with Practice Issues/Partners Disputes	Adhoc		(working document)	and-guidance-manual-pgm/	
9	Management of a Crisis	9.2						
		9.3	Working with relevant stakeholders	Adhoc	Primary Care Team			
10	CQC reports	10.1	Action any issues arising	Adhoc	Angela Price	Carla Sutton, NHSE		
11	Maximise the opportunities afforded as the result of any new tendering / recommissioned of services	11.1	Ensure all new tenders deliver high quality value for money services	Adhoc	Supported by CSU			
	l	12.1	Mobilisation of the new contracts	Annual	Supported by CSU	CCG		

Matther contents in the contents in terevision processing of the contents of	
13.1   Continuation with QFP Practices   15.1   Continuation of this bitter   15.1   Continue to practice   15.1   Continue	
13   Communication of the Principles   13.2   Small bit operations   Marriary   Marria	
1.1   Appointed preferred developed in the processing of the pro	
Secondary   Company   Co	
Description Elabore learning national directive of the Parallel Acades memorities approved sorted processing and provided in Acades and acades acades and acades acades and acades acades and acades acades and acades acades and acades acades and acades acades and acades and acades and acades and acades acades and acades and acades aca	
The search of ship is processing of Premise Improvement (and a force of the comparison of the comparis	
Accessored and proceeding of Premises Improvement 15 Approval for designed recommendation of section of the Section of Section Section of Section Sect	
15   Gassesment and processing of Prinnises Improvement Gards (Sanda Marganet and processing of Prinnises Improvement Gards (Sanda Marganet and processing of Prinnises Improvement Gards)   Author Care National Care Sequence. Level of approve COCks Variable (Sanda Marganet and Sanda Marganet and	
15 Graff Reseases. Level of Jacpor to CGS yearing consistent and Approval of Principal Institution Uniform State Consistent of the Institution Conference of	
15 contracted Monitoring 16 contracted Monitoring 17 contractable Monitoring 18 contractable Monitoring 19 contractable Monitorin	
16   Section and organing minions of CP apparent or Proceed First Section and support in Control of Proceeding Process (subsiting process (subsiting nutrional directive of the transfer date)   16.4   Perpent of Process (subsiting nutrional directive of the transfer date)   16.4   Perpent of Process (subsiting nutrional directive of the transfer date)   16.4   Perpent of Process (subsiting nutrional directive of the transfer date)   17.2   Capital biodistic process (subsiting nutrional directive of the transfer date)   17.2   Sectional Debuty Purioses   Annual Process (subsiting nutrional directive of the transfer date)   17.3   Moderator Excesses with NRS England North   Annual Processes (subsiting nutrional directive of the transfer date)   16.1   Care control perpentitis   Annual Processes (subsiting nutrional directive of the transfer date)   16.1   Care control perpentitis   Annual Processes (subsiting nutrional directive of the transfer date)   16.1   Care control perpentitis   Annual Processes (subsiting nutrional directive of the transfer date)   16.1   Care control perpentitis   Annual Processes (subsiting nutrional directive)   16.2   Section Perpentitional directive of the transfer date (subsitional directive)   16.2   Section Perpentitional directive of the transfer date (subsitional directive)   16.4   Section Perpentitional directive of the transfer date (subsitional directive)   16.4   Section Perpentitional directive of the transfer date (subsitional directive)   16.4   Section Perpentitional directive of the transfer date (subsitional directive)   16.4   Section Perpentitional directive of the transfer date (subsitional directive)   16.4   Section Perpentitional directive of the transfer date (subsitional directive)   16.4   Section Perpentitional directive of the transfer date (subsitional directive)   16.4   Section Perpentitional directive of the transfer date (subsitional directive)   16.4   Section Perpentitional directive of the transfer date (subsitional directive)   16.4   Section Perp	
Estates and regular privates of CP assessed mark (miles)  16.2   Set moves begate resolution establishor of support of Pacifical trained visitor rates  17.3   Set Report of Pacifical trained visitor rates  17.1   Copyrish bird issaed of COGs and Delivery Pharmas  17.2   Sid assessmettive lev  17.3   Moderation Exercise with HNE England North  17.4   Moderation Exercise with HNE England North  17.5   Moderation	
10   Season patients directive of the transfer date)   10   10   10   10   10   10   10   1	
16.4 Payment of Practice rates with stratements and Adhoc decided in the transfer date)  17.1 Capital bidding process (eventing reformal decide) of the transfer date)  18.1 Staffing Model 18.1 To approve the process for staffing model Annual plant (annual plant of the transfer date) of the transfer date)  19.1 Coron control payments Annual plant (annual plant of the transfer date) of the transfer date)  19.2 Core control payments Annual plant (annual plant of the transfer date)  19.3 Core control payments Annual plant (annual plant of the transfer date)  20.1 Interpretation and Translation BAU Core control payments Annual plant of the transfer date (annual plant of the transfer date)  21.1 NHS England function BAU Core control payments Annual plant of the pretation and Translation BAU Core control payments Annual plant of the payments Annual pla	
17   Geptial bids basked of CCGs and Delivery Partners   Annual decicies of the transfer date)   17   Bid as assessment/verwise   Annual 17   Bid as assessment/verwise   Annual 17   Annual 17   Bid as assessment/verwise   Annual 17	
17   Girl Caylala bloking process (awating national discrete of the transfer date)   17.2   Bid assessment/try-few   Annual	
decircle of the transfer date)  17.2 decircle very the transfer date)  17.3 Moderation Exercises with NHS England North Annual  18.1 To approve the process for staffing model  19.1 Core contract payments 19.2 Despoyments 19.3 Core contract experiences 19.3 Core contract calculations 19.3 Core contract calculations 19.3 Core contract calculations 19.3 Core contract calculations 19.4 Moderation 19.5 Complaints 20.1 Interpretation 20.1 Interpretation 20.1 Interpretation 20.1 Interpretation 20.1 Interpretation 20.1 Interpretation 20.1 NHS England function 20.1 NHS England function 20.1 NHS England function 20.1 NHS England function 20.2 Within SFE 20.2 Description of the process of the proce	
18 Staffing Model 19 Finance 19 F	
19.1 Core contract payments 19.2 DES payments 19.2 DES payments 19.3 Core contract calculations 19.3 Core contract calculations 20.1 Interpretation 20.1 Interpretation 20.1 Interpretation 21.1 NHS England function 21.1 NHS England function 22.1 Complaints 21.1 NHS England function 22.2 Within SFE Adhoc 23. Patient Assignments/Allocations 23.1 Following Practice closures in conjunction with local Practices 24. Clare Touthey, Jane Elliott 24. Clinical Waste 25. 1 Check submissions, take appropriate action 26. 1 Check submissions, take appropriate action 27. Contractual Monitoring 28. A with NHSE Finance 28. A with NHSE Finance 29. A with NHSE Finance 20. CCG to continue to use NHSE on intermine and intermine measure NHSE on intermine and intermine measure NHSE until turther notice NHSE on intermine with NHSE until turther notice NHSE on int	
Finance   19.2 DES payments	
19	
Interpretation  20.1 Interpretation and Translation  BAU CCG to continue to use NHSE contracts as an interim measure  Adhoc Responsibility remains with NHSE until further notice  21.1 NHS England function  Adhoc Responsibility remains with NHSE until further notice  22.2 Within SFE  Adhoc Angela Price, Clare Toutey, Jane Elliott, Rob Smith NHSE until further notice  23.1 Following Practice closures in conjunction with local Practices and PCSE  Adhoc Clare Toutey, Jane Elliott, Rob Smith NHSE  24.1 To set up Purchase Orders  Adhoc Responsibility remains with NHSE until further notice  Adhoc Clare Toutey, Jane Elliott, Rob Smith NHSE  Adhoc Responsibility remains with NHSE until further notice  Adhoc Clare Toutey, Jane Elliott, Rob Smith NHSE  25.1 Check submissions, take appropriate action  Annually  Webtool has recently changed, updated process to be agreed  Check submissions, take appropriate action  Annually  Angela Price, Clare Toutey, Jane Elliott, Polane Lewis, NHSE  Annually  Webtool has recently changed, updated process to be agreed  Contractual Monitoring  27.1 Friends and Family Test  Monthly Clare Toutey  Michelle Lyons, NHSE	
20.1 Interpretation 20.1 I	
21 Complaints  22 Locum payments  22 Locum payments  23 Patient Assignments/Allocations  23.1 Following Practice closures in conjunction with local Practices and PCSE  24 Clinical Waste  25 Edec  26 Check submissions, take appropriate action  27.1 Check submissions, take appropriate action  27.1 Contractual Monitoring  27.1 Friends and Family Test  Adhoc  Responsibility remains with NHSE until further notice  Adhoc  Clare Touhey, Jane Elliott  Clare Touhey, Jane Elliott  Diane Lewis, NHSE	
Aggla Price, Clare Touhey, Jane Elliott, Rob Smith Michelle Lyons, NHSE  23 Patient Assignments/Allocations  23.1 Following Practice closures in conjunction with local Practices and PCSE  Adhoc Clare Touhey, Jane Elliott  Adhoc Responsibility remains with NHSE until further notice  TBD - Diane Lewis, NHSE  TBD - Diane Lewis, NHSE  Diane Lewis, NHSE  Diane Lewis, NHSE  Webtool has recently changed, updated process to be agreed  Check submissions, take appropriate action  Annually Changed, updated process to be agreed  Contractual Monitoring  27.1  Angela Price, Clare Touhey, Jane Elliott, Rob Smith Michelle Lyons, NHSE  Diane Lewis, NHSE  Webtool has recently changed, updated process to be agreed  Diane Lewis, NHSE  Diane Lewis, NHSE  Webtool has recently changed, updated process to be agreed  Annually Clare Touhey, Jane Elliott, Rob Smith Michelle Lyons, NHSE  Adhoc Clare Touhey, Jane Elliott, Rob Smith NHSE  Diane Lewis, NHSE  Annually Clare Touhey, Jane Elliott, Rob Smith NHSE  Diane Lewis, NHSE  Angela Price, Jane Elliott, Rob Smith Michelle Lyons, NHSE	
24 Clinical Waste  24.1 To set up Purchase Orders  Adhoc Responsibility remains with NHSE until further notice  Edec  25.1 Check submissions, take appropriate action  Annually be agreed  26.1 Check submissions, take appropriate action  Annually be agreed  Annual	
24.1 To set up Purchase Orders  Adhoc Responsibility remains with NHSE until further notice  Webtool has recently changed, updated process to be agreed  Check submissions, take appropriate action  Annually  Check submissions, take appropriate action  Check submissions, take appropriate action  Annually  Check submissions, take appropriate action  Annually  Annually  Annually  Angela Price, Jane Elliott, Friends and Family Test  Michelle Lyons, NHSE	
25.1 Check submissions, take appropriate action Annually be agreed Diane Lewis, NHSE  26.1 Check submissions, take appropriate action Annually be agreed Diane Lewis, NHSE  27.1 Contractual Monitoring  27.1 Angela Price, Jane Elliott, Michelle Lyons, NHSE  28.1 Check submissions, take appropriate action Annually Diane Lewis, NHSE  29.1 Angela Price, Jane Elliott, Michelle Lyons, NHSE  29.1 Contractual Monitoring Michelle Lyons, NHSE	
26.1 Check submissions, take appropriate action Annually Changed, updated process to be agreed Diane Lewis, NHSE  27.1 Contractual Monitoring  27.1 Angela Price, Jane Elliott, Friends and Family Test Monthly Clare Touhey Michelle Lyons, NHSE  Timetable doc.  Webtool has recently changed, updated process to be agreed  Diane Lewis, NHSE  X:\Primary Care\ Designation\(\frac{Fill efter}{Friends}\) X:\Primary Care\ Designation\(\frac{Fill efter}{Fill efter}\) X:\Primary Care\ Designation\(F	
27.1 Contractual Monitoring  27.1 Angela Price, Jane Elliott, Michelle Lyons, NHSE  27.1 Friends and Family Test  Monthly Clare Touhey Michelle Lyons, NHSE	
27.2 Publication of GP Earnings on individual GP practice websites Annual N/A CCG	imary Care\ tion\FFT\3rd

20		20.4			
20	Indemnity Payments	28.1		Jane Malkin	

Training Required

### **SF/SSCCG Primary Care Programme Report**

Last Updated	11/04/2019 15:17
Total Projects	24
Green Status	20
Amber Status	3
Red Status	1
Closed	5



	Dusiness As								
	Business As Usual/Transformation/National								
Project	Mandate etc	Programme	Sub- section of Programme	Start Date	Planned End Date	Status	NHSE Contact	Resource Name	Comments/Updates
Resilience Funding	Transformation	GP Five Year Forward View		01.04.2017	31.03.2021	G	Gemma Cullen	Angela Price/Jane Elliott/ Clare Touhey	Application process/dates for submission to be agreed for 2019/2020
Clerical and Admin Training (Active Signposting)	Transformation	GP Five Year Forward View	Releasing Time for Care	01.04.2017	31.03.2021	G	Gemma Cullen	Jane Elliott/ Claire Touhey	A training post is now operational via iMerseyside to support signposting. There was a presentation at both wider groups re document management, EOIs are being sort from general practice.
International Recruitment	Transformation	GP Five Year Forward View	GP Five Year Forward View	01.04.2017	31.03.2021	G	Sharon Howard	Craig Gillespie/Kati Scholtz	Meeting held on Tuesday 26th June - 5 interested practices attended. Further weekend event being held for potential recruits currently being planed for early 2019. Potentially there are two recruits identified within C&M. Practice readiness forms will need to be completed by GP practices
GPFV - Additional 5000 Doctors	Transformation	GP Five Year Forward View	GP Five Year Forward View	01.04.17	31.03.2021	R	Sharon Howard	Angela Price	5000 is the number of doctors required nationally, this drilled down to C&M is 194 additional doctors.
ETTF- Estates Bids	Transformation	GP Five Year Forward View	ETTF	01.04.2017	ongoing	А	David Scannell	Jan Leonard/Sam McCumiskey	3 bids for co-location have been submitted. The bids are at PID stage, there is currently no further progress to report with the bids.
E-Consultations/Online Consulting	Transformation	GP Five Year Forward View	Online Consulting	01.04.2017	31.03.2020	G	Gemma Cullen (Sophie Bartsch for C&M – GC can act as SS/SF link)	Jane Elliot/Paul Shillcock	A roll out plan has been agreed across both CCGs. Discussions are in place regarding how this could work in the 7 day access service.
7 Day Access	Transformation	GP Five Year Forward View	Primary Care HUB	01.08.2017	ongoing	G	Jan Hughes	Angela Price /Clinical Leads	Both services went live on Monday 1st October 2018, positive feedback is being recieved. Monthly reporting data is being received. Merseysdie are working with NHS Digital regarding ERS and ICE, testing took place in March 2019. First contact physiotherpy has started in S&F. Contract meetings with both providers taking place quarterly. Both services were commissioned via an APMS contract for a 2 year period (September 2020) with the option to extend for a year if needed.
Clinical Pharmacy Pilot	Transformation	GP Five Year Forward View	Workforce	01.04.2017	ongoing	G	Gemma Cullen	Susanne Lynch/Clinical Leads	Pilots will commence in Crosby and North localities based on hub models, initially dealing with medication queries, discharge summaries etc.
Apex/Insight	Transformation	GP Five Year Forward View	Workload		ongoing	G	Gemma Cullen (Cathy Leech for C&M – GC can act as SS/SF link)	Leads Angela Price	Apex/Insight has been demonstrated at both wider group meetings. Practices have been asked to express an interest in using the tool as part of an NHSE pilot.
ETTF- IT	Transformation	GP Five Year Forward View	ETTF	01.04.2017	ongoing	А	David Scannell	Jane Elliot/Paul Shillcock	Bids were submitted May/June 2018. It was announced that the ETTF bids had been declined. A telecon was undertaken between NHSE and iMerseyside to discuss in more detail and it was suggested there may be potential to resubmit bids next year.

	Business As								
Project	Usual/Transformation/National Mandate etc		Sub- section of Programme	Start Date	Planned End Date	Status	NHSE Contact	Resource Name	Comments/Updates
Project  Practice Nurse 10 Point Plan	Transformation	GP Five Year Forward View	Practice Nurse 10 point plan	23.07.2017	Ongoing	G	Pippa Rose	Colette Page	ETP has employed a p/t mentor to increase numbers of practices accepting student nurses and number of mentors within practices. 8 PNs to become clinical supervisors allowing them to offer clinical supervision to PNsacross both CCGs. Progress being made with a preceptorship programme for new PNs. Promoting 'All our Health' learning platform - to embed prevention, health protection and promotion of wellbeing. MECC training delivered previously and training in Anti-Microbial Resistance ongoing. 4 more PNs and 3 pharmacists on the NMP course. 7 PNs on the GPN Leadership for Quality programme. Promoting the 2 year Advanced Care Practitioner course across CCGs. In dicsussion with NHSE lead to promote Nursing Associate trainees within practices. Recruitment and retention continues to be challenging for many practices nationally. Promoting GPN as a viable career pathway is ongoing.
Workforce Steering Group	Transformation	GP Five Year Forward View	Workforce	01/05/2018	ongoing	A	Glen Coleman/Sharon Howard	Angela Price	A C&M workforce steering group has been developed. This is an extension of the Task and Finish Group for International Recruitment. This group will feed into the LWAB (Local Workforce Action Board).
2019/2020 GP Contract Plan	GP Contract	Delegation		01/04/2019	31/03/2020	G		Angela Price	Transition plan to full delegation of primary medical care achieved.
Primary Care Networks	GP Contract	Delegation	DES	01/04/2019	31/03/2020	G	Sharon Howard / Gemma Cullen	Angela Price	8 Networks / DES specification published - PCN agreements / data sharing agreements / Clinical Director posts / PCN delivery of extended access to 100% of population by 1st July 2019. 7 specifications being developed nationally in 2019/20 for delivery by PCNs in 2021.
Learning Disabilities	GP Contract	Delegation	DES	01.04.2019	31/03/2020	G		Angie Price	Practice/Federation Delivery - Improve uptake / registers. South Sefton 17/18 85 health checks undertaken, 18/19 register 659, health checks undertaken 204 to date, a further update is due. Southport and Formby, 17/18 173 health checks undertaken, 18/19 register 759, health checks undertaken 351 to date, further update is due.
GP Contract	Business As Usual	Delegation	NHS Digital Finance Data	01.04.2019	31.03.2020	G	N/A	Angela Price	Practice finance visits are being planned using NHS digital data on NHS funding earned in 2017/18
Local Quality Contract Phase 4	Business As Usual	Primary Care Operational	Business As Usual	01.04.2018	01.09.2019	G	N/A	Angela Price/ Craig Gillespie/ Kati Scholtz	Phase 4 LQC has been operational 01.04.18 - 31.03.19. Validation panel is being conveened July 2019 to assess practice end of year reports.
Local Quality Contract Phase 5	Business As Usual	Primary Care Operational	Business As Usual	01.04.2019	01.09.2020	G	N/A	Angela Price/Craig Gillespie/Kati Scholtz	Phase S LQC agreed and circulated to GP practices - sign up to schemes by 30th April 2019.
Localities	Business As Usual	ссс	ccg	01.04.2019	ongoing	G	N/A	Jane Elliott/ Claire Touhey	Business as usual, PCNs have formed some discussions at localities. Locality mangers - Clare Touhey for Southport and Formby, and Jane Elliot for South Sefton
Influenza Coordination 2019/2020	National Mandate	ссс	ccg	01.04.2019	ongoing	G	Jennie Piet	Colette Page	Member of Flu task & Finish group with NHSE and sub group at Sefton Local authority. Begin developing coordination plans for 2019/20 season
Local Authority Health Protection Forum	National Mandate	ссс	ссс	01.04.19	ongoing	G	Charlotte Smith	Colette Page	Attend bi monthly Health Protection Forum (HPF) chaired by the LA. Flu subgroup reports into the HPF, which then feeds into NHSE.

	Business As								
	Usual/Transformation/National								
Project	Mandate etc	Programme	Sub- section of Programme	Start Date	Planned End Date	Status	NHSE Contact	Resource Name	Comments/Updates
NHSE/ PHE S&I Programme Boards	National Mandate	cce	ссс	01.04.2019	Ongoing	G	Hayley Mercer	Colette Page	Attend quarterly Screening & Immunisation Programme Boards. Review targets achieved by CCGs and address concerns. Contact practices not attaining national targets to gain assurance or assist/provide support where necessary. Liaise with PHE S&I coordinator/ managers.
Practice Nurse/ HCA meetings					ongoing	G	N/A		Arrange and coordinate PN/ HCA meetings and training events. Source speakers/ training providers. Utilise flexible cash funding available from Health Education England to support Non-medical prescribing courses for PNs and clinical pharmacists, Fully fund a clinical supervision course for 8 PNs to offer supervision across the CCG in line with GPFV PN 10 PP. Review priorities across CCG in LTCs/ disease areas and focus training for PNs/ HCAs around these.
Protected Learning Time (PLT)	Business As Usual	cce	cce		ongoing	G	N/A	Colette Page	Arrange 4-6 PLT events per year. Source consultants and expert speakers from acute/ community/ voluntary providers. Arrange for pharmacuetical sponsors to support the events. Facilitate the afternoon with 150-200 delegates in attendance.
Project Activity Key					Project RAG Key				
7		Ahead of Schedule			G				
Л		Behind Schedule	1		A				
$\rightarrow$		On target			R				
X		No Activity	_						



### **Investment and Evolution**

Five year framework for GP contract reform to implement
The NHS Long Term Plan



Staying local & together together with you



### Overview

- The new contract focuses on:
  - Workload/ Workforce issues
  - Indemnity
  - QOF
  - Primary Care Network Contract
  - Joining up urgent care services
  - Digital technologies
  - New services





### **Workforce and Workload Issues**

Through the new Additional Roles Reimbursement Scheme, PCNs will be guaranteed funding for up to 20,000 additional staff by 2023/24.

Scheme will meet 70% of costs for clinical pharmacists, physicians' associates, first contact physiotherapists, first contact community paramedics, and 100% of costs for social prescribing link workers.

Practices will fund other staff groups GP and nurses through the core contract

There will also be a new Primary Care Fellowship Scheme and additional Training Hubs.

Introductory arrangements: between 1 July 19 and 31 March 20, every PCN of at least 30k population will be able to claim for 70% funding for 1 wte clinical pharmacist and 100% for 1 wte social prescribing link worker.





### **Indemnity Costs and Coverage**

New, permanent centrally-funded Clinical negligence Scheme for General Practice will start from April 2019.

NHS GP Service Providers (Including OOHs) are eligible to become members and will not have to pay a subscription now or in the future. Costs met through a national allocation.

### Will cover:

- general practice
- Out of hours
- And all staff groups
- The wider PC Network

Membership is free. Funded through a one-off permanent adjustment to global sum.

However, practice contract funding will rise in 19/20 by 1.4%





### Improvement to QOF

QOF Review findings are being implemented.

28 indicators (175 points) being retired from April 2019.

74 points will be used to create a new **Quality Improvement** domain. The first two modules in 19/20 are:

- Prescribing safety
- End of Life Care

101 points used for more clinically appropriate indicators: diabetes, blood pressure control, cervical screening.





### **Primary Care Network Contract**

General practice takes leading role in every PCN.

Network Contract is a DES (extension of GP core contract and not separate to it).

DES supports practices of all sizes, working together within neighbourhoods.

CCGs able to add local investment through Supplementary Network Services.

100% coverage by 'go live' date July 2019.

Each Network to have a named accountable Clinical Director.





### **Joined Up Urgent Care Services**

PCNs join up the delivery of urgent care in the community.

Funding and responsibility for providing CCG-commissioned 7 day access services transfers to Network DES by April 2021.

From July 2019, the Extended Hours DES requirements are introduced across every Network until March 2021.

Following an Access Review in 2019, a more coherent set of access arrangements will start to be implemented in 2020.

111 direct booking into practices will be introduced nationally in 2019.

Waiting time data published monthly from 2021 alongside hospital data.



### **Practices and Patients to Benefit from Digital Technologies**

GPIT Futures Programme will replace GPSoC

National funding to give PCNs access to digital first support from April 2021 from an agreed list of suppliers on a new framework.

By April 2021: all patients have the right to digital first primary care, including web and video consultations (25% by July 2019).

From 2020, all patients to have digital access to their full records.

From 2019, no longer possible for GP providers to advertise or host paid-for GP services that fall within the scope of NHS funded primary medical services.

NHS England will ensure that predictive analytical tools are available to PCNs





### **Delivering New Services**

Scale of investment phased and secured for delivery of the NHS Long Term Plan commitments.

PCNs will need to deliver seven national Network Service Specifications.

### Five of the seven start by April 2020:

- Structured medication reviews
- Enhanced health in care homes
- Anticipatory care (with community services)
- Personalised care
- Supporting early cancer diagnosis

### Remaining two start by 2021:

- Cardio-vascular disease case finding
- Locally agreed action to tackle health inequalities

2019: Review of vaccination and immunisations arrangements and outcomes under GP contracts to be undertaken and will also cover screening.

2020: Network Dashboard available and also Network Investment and Impact Fund to accelerate progress against commitments and NHS Long Term Plan.





### 2019/20

#### Key changes and development work

#### Network

- Network Contract DES goes live
- Network participation practice payment starts
- Design of new national network service specifications starts

#### Workforce

- Additional Roles Reimbursement Scheme launched, starting with clinical pharmacists and social prescribing link workers
- · Design of primary care fellowship programme

#### Indemnity

New centrally, funded Clinical Negligence Scheme for General Practice starts

#### QOF reform

- 28 indicators retired; 15 new indicators; introduction of personalised care adjustment and Quality Improvement domain
- · Review of heart failure, asthma and chronic obstructive pulmonary disease QOF domains

#### Testbeds

New primary care testbed programme launched

#### Digital

- · New digital improvement requirements introduced including access by patients online to full record
- Revisions to rurality index payment and London adjustment
  - Review of premium for registering new patients
- Review of out of area registration and choice of digital-first registration
- Requirement for Electronic ordering of repeat prescriptions and using electronic repeat dispensing for all patients for whom it is clinically appropriate as a default from April 2019
  - All practices will ensure at least 25% of appointments are available for online booking by July 2019

### Advertising

• Ban on GP providers advertising or hosting paid-for GP services that fall within the scope of NHS funded primary medical services

#### **Access**

- Extended Hours Access DES requirements introduced across all practices in every network (until March 2021)
- NHS111 direct appointment booking into practices introduced nationally
- Review of wider access arrangements





### 2020/21

#### 2020/21 Key changes and development work

#### Networks

- New Dashboard to monitor progress on network metrics
- National Network Investment and Impact Fund launched
- Anticipatory care requirements (with community services) start
  - Enhanced health in care home requirements start
- Structured Medication review requirements start for priority groups
  - Personalised care requirements start
- Early cancer diagnosis support requirements start

#### Workforce

- Additional Roles Reimbursement Scheme extended to include physician associates and first contact physiotherapists
  - Primary care training hubs established

#### QOF reform

- Further changes introduced post review of heart failure, asthma, and COPD domains
- Review of mental health domain
- New QI modules

### Digital

• Requirement for online presence, to give patients access online to correspondence and to no longer be using facsimile machines for either NHS or patient communications;

### Out of area registration

• Potential out of area registration reform - post review

#### Vacs and Imms

• Changes to vaccination and immunisation arrangements – post review

#### Access

• Start of transition to new access arrangements – post review

### 2021/22

- Key changes and development work
- Networks
- Cardio-vascular disease case finding requirements start
- Prevention and inequalities requirements start

Workforce

- Additional Roles Reimbursement Scheme extended to include community paramedics
- QOF
- Further changes introduced, including new QI modules
- Digital
- New digital-first support offer



### **Key Issues Report to Governing Body**



Chair: Gill Brown

Key Issue	Risk Identified	Mitigating Actions
Patients potentially being charged excessive fees when being held in telephone queuing systems when ringing their GP practice.	Some patients may be unable to contact practices due to affordability.	Clarify position with NHSE / National position on telephone lines for patient use and share with practices.

### Information Points for Southport and Formby CCG Governing Body (for noting)

The committee noted the changes to the GP contract and are awaiting further guidance due to be published at the end of March 2019. The committee will review this at the next meeting.

### **Key Issues Report to Governing Body**



	South Sefton & NHSE Joint Commissioning	Meeting Part 1	, Thursday	y 21 <sup>st</sup> March 2019
--	---	----------------	------------	-------------------------------

Chair: Graham Bayliss

Key Issue	Risk Identified	Mitigating Actions

### Information Points for South Sefton CCG Governing Body (for noting)

The committee noted the changes to the GP contract and are awaiting further guidance due to be published at the end of March 2019. The committee will review this at the next meeting.